UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 10-Q

(Mark One)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2022

or

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to Commission file number: 001-35666

Summit Midstream Partners, LP

(Exact name of registrant as specified in its charter)

Delaware 45-5200503

(State or other jurisdiction of I.R.S. Employer incorporation or organization)

Identification No.)

910 Louisiana Street, Suite 4200

Houston, TX 77002

(Address of principal executive offices) (Zip Code)

(832) 413-4770

(Registrant's telephone number, including area code)

Not applicable

(Former name, former address and former fiscal year, if changed since last report)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered							
Common Units	SMLP	New York Stock Exchange							
Indicate by check mark whether the registrant (1) has filed all reports required such shorter period that the registrant was required to file such reports), and (2 Indicate by check mark whether the registrant has submitted electronically eveduring the preceding 12 months (or for such shorter period that the registrant v) has been subject to such filing requir rry Interactive Data File required to be	ements for the past 90 days. x Yes 0 No							
x Yes o No									
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.									
Large accelerated filer		Accelerated filer x							
Non-accelerated filer \Box		Smaller reporting company X							
Emerging growth company \Box									
If an emerging growth company, indicate by check mark if the registrant has elstandards provided pursuant to Section 13(a) of the Exchange Act. 0	lected not to use the extended transition	n period for complying with any new or revised financial accounting							
Indicate by check mark whether the registrant is a shell company (as defined in	n Rule 12b-2 of the Exchange Act). \Box	Yes x No							
Indicate the number of shares outstanding of each of the issuer's classes of cor	mmon stock, as of the latest practicable	date.							
Class		As of July 29, 2022							
Common Units		10,166,175 units							

TABLE OF CONTENTS

COMMONLY U	<u>JSED OR DEFINED TERMS</u>	<u>2</u>
<u>PART I</u>	FINANCIAL INFORMATION	
Item 1.	<u>Financial Statements.</u>	
	<u>Unaudited Condensed Consolidated Balance Sheets.</u>	<u>5</u>
	<u>Unaudited Condensed Consolidated Statements of Operations.</u>	<u>6</u>
	<u>Unaudited Condensed Consolidated Statements of Partners' Capital.</u>	<u>7</u>
	<u>Unaudited Condensed Consolidated Statements of Cash Flows.</u>	<u>8</u>
	Notes to Unaudited Condensed Consolidated Financial Statements.	<u>9</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations.	<u>22</u>
Item 3.	Quantitative and Qualitative Disclosures about Market Risk.	<u>41</u>
Item 4.	Controls and Procedures.	<u>41</u>
<u>PART II</u>	OTHER INFORMATION	<u>42</u>
Item 1.	<u>Legal Proceedings.</u>	<u>42</u>
Item 1A.	Risk Factors.	<u>42</u>
Item 5.	Other Information.	<u>42</u>
Item 6.	Exhibits.	<u>42</u>
<u>SIGNATURES</u>		<u>44</u>

COMMONLY USED OR DEFINED TERMS

2015 Blacktail Release	a 2015 rupture of our four-inch produced water gathering pipeline near Williston, North Dakota
2022 Senior Notes	Summit Holdings' and Finance Corp.'s 5.5% senior unsecured notes due August 2022
2025 Senior Notes	Summit Holdings' and Finance Corp.'s 5.75% senior unsecured notes due April 2025
2026 Secured Notes	Summit Holdings' and Finance Corp.'s 8.500% senior secured second lien notes due 2026
ABL Facility	the asset-based lending credit facility governed by the ABL Agreement
ABL Agreement	Loan and Security Agreement, dated as of November 2, 2021, among Summit Holdings, as borrower, SMLP and certain subsidiaries from time to time party thereto, as guarantors, Bank of America, N.A., as agent, ING Capital LLC, Royal Bank of Canada and Regions Bank, as co-syndication agents, and Bank of America, N.A ING Capital LLC, RBC Capital Markets and Regions Capital Markets, as joint lead arrangers and joint bookrunners
ASU	Accounting Standards Update
Bison Midstream	Bison Midstream, LLC
Board of Directors	the board of directors of our General Partner
condensate	a natural gas liquid with a low vapor pressure, mainly composed of propane, butane, pentane and heavier hydrocarbon fractions
Co-Issuers	Summit Holdings and Finance Corp.
DFW Midstream	DFW Midstream Services LLC
DJ Basin	Denver-Julesburg Basin
Double E	Double E Pipeline, LLC
Double E Pipeline	a 135 mile, 1.35 Bcf/d, FERC-regulated interstate natural gas transmission pipeline that commenced operation in November 2021 and provides transportation service from receipt points in the Delaware Basin to various delivery points in and around the Waha hub in Texas
Double E Project	the development and construction of the Double E Pipeline
ECP	Energy Capital Partners II, LLC and its parallel and co-investment funds
EPA	Environmental Protection Agency
Epping	Epping Transmission Company, LLC
EPU	earnings or loss per unit
FASB	Financial Accounting Standards Board
Finance Corp.	Summit Midstream Finance Corp.
GAAP	accounting principles generally accepted in the United States of America
General Partner	Summit Midstream GP, LLC
GP	general partner
GP Buy-In Transaction	the transactions contemplated by the Purchase Agreement dated May 3, 2020, between the Partnership and the affiliates of its then private equity sponsor, ECP, to acquire Summit Investments
Grand River	Grand River Gathering, LLC
Guarantor Subsidiaries	Bison Midstream and its subsidiaries, Grand River and its subsidiaries, DFW Midstream, Summit Marketing Summit Permian, Permian Finance, OpCo, Summit Utica, Meadowlark Midstream, Summit Permian II and Mountaineer Midstream
Hub	geographic location of a storage facility and multiple pipeline interconnections
LIBOR	London Interbank Offered Rate
Mbbl/d	one thousand barrels per day
MD&A	Management's Discussion and Analysis of Financial Condition and Results of Operations
Meadowlark Midstream	Meadowlark Midstream Company, LLC
MMBTU	metric million British thermal units
MMcf/d	one million cubic feet per day
Mountaineer Midstream	Mountaineer Midstream Company, LLC

MVC	minimum volume commitment
NGLs	natural gas liquids; the combination of ethane, propane, normal butane, iso-butane and natural gasolines that when removed from unprocessed natural gas streams become liquid under various levels of higher pressure and lower temperature
NYSE	New York Stock Exchange
Ohio Gathering	Ohio Gathering Company, L.L.C. and Ohio Condensate Company, L.L.C.
OpCo	Summit Midstream OpCo, LP
play	a proven geological formation that contains commercial amounts of hydrocarbons
Permian Finance	Summit Midstream Permian Finance, LLC
Permian Holdco	Summit Permian Transmission Holdco, LLC
Permian Term Loan Facility	the term loan governed by the Credit Agreement, dated as of March 8, 2021, among Summit Permian Transmission, LLC, as borrower, MUFG Bank Ltd., as administrative agent, Mizuho Bank (USA), as collateral agent, ING Capital LLC, Mizuho Bank, Ltd. and MUFG Union Bank, N.A., as L/C issuers, coordinating lead arrangers and joint bookrunners, and the lenders from time to time party thereto
Permian Transmission Credit Facilities	the credit facilities governed by the Credit Agreement, dated as of March 8, 2021, among Summit Permian Transmission, LLC, as borrower, MUFG Bank Ltd., as administrative agent, Mizuho Bank (USA), as collateral agent, ING Capital LLC, Mizuho Bank, Ltd. and MUFG Union Bank, N.A., as L/C issuers, coordinating lead arrangers and joint bookrunners, and the lenders from time to time party thereto
produced water	water from underground geologic formations that is a by-product of natural gas and crude oil production
Revolving Credit Facility	the Third Amended and Restated Credit Agreement dated as of May 26, 2017, as amended by the First Amendment to Third Amended and Restated Credit Agreement dated as of September 22, 2017, the Second Amendment to Third Amended and Restated Credit Agreement dated as of June 26, 2019, the Third Amendment to Third Amended and Restated Credit Agreement dated as of December 24, 2019 and the Fourth Amendment to Third Amended and Restated Credit Agreement dated as of December 18, 2020
SEC	Securities and Exchange Commission
Securities Act	Securities Act of 1933, as amended
segment adjusted EBITDA	total revenues less total costs and expenses; plus (i) other income excluding interest income, (ii) our proportional adjusted EBITDA for equity method investees, (iii) depreciation and amortization, (iv) adjustments related to MVC shortfall payments, (v) adjustments related to capital reimbursement activity, (vi) unit-based and noncash compensation, (vii) impairments and (viii) other noncash expenses or losses, less other noncash income or gains
Senior Notes	The 2025 Senior Notes and the 2026 Secured Notes, collectively
Series A Preferred Units	Series A Fixed-to-Floating Rate Cumulative Redeemable Perpetual Preferred Units
shortfall payment	the payment received from a counterparty when its volume throughput does not meet its MVC for the applicable period
SMLP	Summit Midstream Partners, LP
SMLP LTIP	SMLP Long-Term Incentive Plan
SMP Holdings	Summit Midstream Partners Holdings, LLC, also known as SMPH
SMPH Term Loan	the Term Loan Agreement, dated as of March 21, 2017, among SMP Holdings, as borrower, the lenders party thereto and Credit Suisse AG, Cayman Islands Branch, as Administrative Agent and Collateral Agent
Subsidiary Series A Preferred Units	Series A Fixed Rate Cumulative Redeemable Preferred Units issued by Permian Holdco
Summit Holdings	Summit Midstream Holdings, LLC
	Summit Midstream Holdings, LLC Summit Midstream Partners, LLC

Table of Contents

Summit Permian	Summit Midstream Permian, LLC
Summit Permian II	Summit Midstream Permian II, LLC
Summit Permian Transmission	Summit Permian Transmission, LLC
Summit Utica	Summit Midstream Utica, LLC
the Partnership	Summit Midstream Partners, LP and its subsidiaries
the Partnership Agreement	the Fourth Amended and Restated Agreement of Limited Partnership of the Partnership dated May 28, 2020
throughput volume	the volume of natural gas, crude oil or produced water gathered, transported or passing through a pipeline, plant or other facility during a particular period; also referred to as volume throughput
unconventional resource basin	a basin where natural gas or crude oil production is developed from unconventional sources that require hydraulic fracturing as part of the completion process, for instance, natural gas produced from shale formations and coalbeds; also referred to as an unconventional resource play
wellhead	the equipment at the surface of a well, used to control the well's pressure; also, the point at which the hydrocarbons and water exit the ground

PART I - FINANCIAL INFORMATION

Item 1. Financial Statements.

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES UNAUDITED CONDENSED CONSOLIDATED BALANCE SHEETS

		June 30, 2022		December 31, 2021
ASSETS		(In thousands, ex	cept u	nit amounts)
Cash and cash equivalents	\$	12,367	¢	7,349
Restricted cash	Ф	3,333	Ф	12,223
Accounts receivable		59,392		62,121
Other current assets		3,716		5,676
Total current assets		78,808		87,369
Property, plant and equipment, net		1,531,450		1,726,082
Intangible assets, net		156,877		172,927
Investment in equity method investees		519,553		523,196
Other noncurrent assets		23,645		12,888
	\$	2,310,333	\$	2,522,462
TOTAL ASSETS	D.	2,310,333	Ф	2,322,402
LIABILITIES AND CAPITAL				
Trade accounts payable	\$	8,378	\$	10,498
Accrued expenses	•	21,323	•	14,462
Deferred revenue		9,720		10,374
Ad valorem taxes payable		4,738		8,570
Accrued compensation and employee benefits		6,760		11,019
Accrued interest		15,992		12,737
Accrued environmental remediation		2,108		3,068
Current portion of long-term debt		7,526		_
Other current liabilities		10,476		8,509
Total current liabilities		87,021	_	79,237
Long-term debt, net of issuance costs		1,232,512		1,355,072
Noncurrent deferred revenue		40,362		42,570
Noncurrent accrued environmental remediation		2,272		2,538
Other noncurrent liabilities		32,184		32,357
Total liabilities		1,394,351	1	1,511,774
Commitments and contingencies (Note 14)				
Mezzanine Capital				
Subsidiary Series A Preferred Units (93,039 and 91,439 units issued and outstanding at June 30, 2022 and December 31, 2021, respectively)		113,574		106,325
Partners' Capital				
Series A Preferred Units (65,508 and 143,447 units issued and outstanding at June 30, 2022 and December 31, 2021, respectively)		81,290		169,769
Common limited partner capital (10,166,175 and 7,169,834 units issued and outstanding at June 30, 2022 and December 31, 2021, respectively)		721,118		734,594
Total partners' capital		802,408		904,363
TOTAL LIABILITIES AND CAPITAL	\$	2,310,333	\$	2,522,462

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

	Three Months Ended June 30,				Six Months E	nded	ded June 30,	
		2022		2021		2022	2021	
			(In	thousands, exce	pt pe	r-unit amounts)		
Revenues:	_	_,	_					
Gathering services and related fees	\$	61,631	\$	74,233	\$	125,651	\$	144,580
Natural gas, NGLs and condensate sales		28,278		16,416		50,736		37,180
Other revenues		9,154		9,392		18,802		17,599
Total revenues		99,063		100,041		195,189		199,359
Costs and expenses:								
Cost of natural gas and NGLs		26,831		16,626		49,082		37,102
Operation and maintenance		22,277		17,507		39,339		34,100
General and administrative		10,473		29,360		23,433		39,938
Depreciation and amortization		30,111		28,364		60,556		56,875
Transaction costs		(13)		450		233		217
Gain on asset sales, net		(313)		(4)		(310)		(140)
Long-lived asset impairments		84,614		33		84,628		1,525
Total costs and expenses		173,980		92,336		256,961		169,617
Other income (expense), net		(4)		352		(4)		408
Gain (loss) on interest rate swaps		3,936		(2,686)		10,964		(2,692)
Loss on ECP Warrants		_		(12,159)		_		(13,634)
Interest expense		(24,887)		(15,502)		(49,050)		(29,455)
Loss before income taxes and equity method investment income		(95,872)		(22,290)		(99,862)		(15,631)
Income tax (expense) benefit		(325)		248		(375)		262
Income from equity method investees		4,393		2,304		8,428		4,619
Net loss	\$	(91,804)	\$	(19,738)	\$	(91,809)	\$	(10,750)
Net income attributable to Subsidiary Series A Preferred Units		(3,164)		(4,089)		(8,877)		(8,021)
Net loss attributable to Summit Midstream Partners, LP	\$	(94,968)	\$	(23,827)	\$	(100,686)	\$	(18,771)
Less: net income attributable to Series A Preferred Units		(1,888)		(3,849)		(4,108)		(8,136)
Add: deemed contribution from 2022 Preferred Exchange Offer		_		_		20,974		_
Add: deemed contribution from 2021 Preferred Exchange Offer				8,326				8,326
Net loss attributable to common limited partners	\$	(96,856)	\$	(19,350)	\$	(83,820)	\$	(18,581)
Net loss per limited partner unit:								
Common unit – basic	\$	(9.53)	\$	(2.91)	\$	(8.45)	\$	(2.91)
Common unit – diluted	\$	(9.53)	\$	(2.91)	\$	(8.45)	\$	(2.91)
Weighted-average limited partner units outstanding:								
Common units – basic		10,166		6,656		9,919		6,392
Common units – diluted		10,166		6,656		9,919		6,392

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF PARTNERS' CAPITAL

	Partners' Capital					
	Units P			ommon Limited Partners Capital		Total
				(In thousands)		
Partners' capital, January 1, 2022	\$	169,769	\$	734,594	\$	904,363
Net income (loss)		2,220		(7,938)		(5,718)
Unit-based compensation		_		1,690		1,690
Tax withholdings and associated payments on vested SMLP LTIP awards		_		(562)		(562)
Tax withholdings on 2022 Preferred Exchange Offer		_		(2,652)		(2,652)
Effect of 2022 Preferred Exchange Offer, inclusive of a \$20.9 million deemed contribution to common unit holders (Note 10)		(92,587)		92,587		_
Partners' capital, March 31, 2022	\$	79,402	\$	817,719	\$	897,121
Net income (loss)		1,888		(96,856)		(94,968)
Unit-based compensation		_		582		582
Tax withholdings and associated payments on vested SMLP LTIP awards		_		(327)		(327)
Partners' capital, June 30, 2022	\$	81,290	\$	721,118	\$	802,408

	Partners' Capital					
	Series A Preferred Units			Common Limited Partners Capital		Total
				(In thousands)		
Partners' capital, January 1, 2021	\$	174,425	\$	748,466	\$	922,891
Net income		4,287		769		5,056
Unit-based compensation		_		1,967		1,967
Tax withholdings and associated payments on vested SMLP LTIP awards		_		(1,274)		(1,274)
Partners' capital, March 31, 2021	\$	178,712	\$	749,928	\$	928,640
Net income (loss)		3,849		(27,676)		(23,827)
Unit-based compensation		_		1,048		1,048
Tax withholdings and associated payments on vested SMLP LTIP awards		_		(98)		(98)
Tax withholdings on 2021 Preferred Exchange Offer		_		(465)		(465)
Effect of 2021 Preferred Exchange Offer, inclusive of an \$8.3 million deemed contribution						
to common unit holders (Note 10)		(20,654)		20,654		_
Partners' capital, June 30, 2021	\$	161,907	\$	743,391	\$	905,298

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

	Six Mont	hs Ended June 30,
	2022	2021
	(In	thousands)
Cash flows from operating activities:		
Net loss	\$ (91,8	09) \$ (10,750)
Adjustments to reconcile net loss to net cash provided by operating activities:		
Depreciation and amortization	61,0	25 57,344
Noncash lease expense	4	68 519
Amortization of debt issuance costs	4,4	41 3,447
Unit-based and noncash compensation	2,2	3,015
Income from equity method investees	(8,4	28) (4,619)
Distributions from equity method investees	20,4	51 13,116
Gain on asset sales, net	(3	10) (140)
Loss on ECP Warrants		— 13,634
(Gain) loss on interest rate swaps, unsettled	(11,6	17) 2,692
Long-lived asset impairment	84,6	528 1,525
Changes in operating assets and liabilities:		
Accounts receivable	(3,0	10) (1,993)
Trade accounts payable	(1,0	90) 2,989
Accrued expenses	11,5	599 (3,127)
Deferred revenue, net	(2,7	(2,787)
Ad valorem taxes payable	(3,3	84) (3,691)
Accrued interest	3,2	.54 —
Accrued environmental remediation, net	(1,2	26) (512)
Other, net	(4,3	57) 15,555
Net cash provided by operating activities	60,1	.59 86,217
Cash flows from investing activities:		
Capital expenditures	(14,7	(94) (5,962)
Proceeds from sale of Lane G&P System, net of cash sold in transaction	75,5	20 —
Proceeds from asset sale	4,6	8,000
Investment in Double E equity method investee	(8,4	44) (48,943)
Net cash provided from (used in) investing activities	56,8	882 (46,905)
Cash flows from financing activities:		
Repayments on Revolving Credit Facility		— (95,000°
Borrowings under Permian Transmission Credit Facility		— 53,500
Repayments on Permian Transmission Term Loan	(2,2	42) —
Repayments on ABL Facility	(139,0	·
Borrowings on ABL Facility	23,0	
Debt issuance costs		— (5,179)
Proceeds from asset sale		
Other, net	(2,6	
Net cash used in financing activities	(120,9	
Net change in cash, cash equivalents and restricted cash	(3,8	
Cash, cash equivalents and restricted cash, beginning of period	19,5	
	\$ 15,7	
Cash, cash equivalents and restricted cash, end of period	\$ 15,/	υυ φ /,525

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES

NOTES TO UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

1. ORGANIZATION, BUSINESS OPERATIONS AND PRESENTATION AND CONSOLIDATION

Organization. Summit Midstream Partners, LP (including its subsidiaries, collectively "SMLP" or the "Partnership") is a Delaware limited partnership that was formed in May 2012 and began operations in October 2012. SMLP is a value-oriented limited partnership focused on developing, owning and operating midstream energy infrastructure assets that are strategically located in unconventional resource basins, primarily shale formations, in the continental United States. The Partnership's business activities are primarily conducted through various operating subsidiaries, each of which is owned or controlled by its wholly owned subsidiary holding company, Summit Holdings, a Delaware limited liability company.

Business Operations. The Partnership provides natural gas gathering, compression, treating and processing services as well as crude oil and produced water gathering services pursuant to primarily long-term, fee-based agreements with its customers. The Partnership's results are primarily driven by the volumes of natural gas that it gathers, compresses, treats and/or processes as well as by the volumes of crude oil and produced water that it gathers. Other than the Partnership's investments in Double E and Ohio Gathering, all of its business activities are conducted through wholly owned operating subcidiaries.

Presentation and Consolidation. The Partnership prepares its condensed consolidated financial statements in accordance with GAAP as established by the FASB and pursuant to the rules and regulations of the SEC pertaining to interim financial information. The unaudited condensed consolidated financial statements contained in this report include all normal and recurring material adjustments that, in the opinion of management, are necessary for a fair statement of the financial position, results of operations and cash flows for the interim periods presented herein. These unaudited condensed consolidated financial statements and notes thereto should be read in conjunction with the consolidated financial statements and related notes that are included in the Partnership's Annual Report on Form 10-K for the year ended December 31, 2021.

The Partnership makes estimates and assumptions that affect the reported amounts of assets and liabilities at the balance sheet dates, including fair value measurements, the reported amounts of revenues and expenses and the disclosure of commitments and contingencies. Although management believes these estimates are reasonable, actual results could differ from its estimates.

The unaudited condensed consolidated financial statements contained in this report include the assets, liabilities and results of operations of SMLP and its subsidiaries. All intercompany transactions among the consolidated entities have been eliminated in consolidation. Comprehensive income or loss is the same as net income or loss for all periods presented.

Risks and Uncertainties. The Partnership continues to closely monitor the impact of the COVID-19 pandemic on all aspects of its business, including how it has impacted and will impact its customers, employees, supply chain and distribution network. The Partnership is unable to predict the ultimate impact that COVID-19 may have on its business, future results of operations, financial position or cash flows.

Given the dynamic nature of the COVID-19 pandemic and related market conditions, the Partnership cannot reasonably estimate the period of time that these events will persist or the full extent of the impact they will have on its business. The full extent to which the Partnership's operations continue to be impacted by the COVID-19 pandemic will depend largely on future developments, which remain highly uncertain and cannot be accurately predicted. While many of the restrictions and measures initially implemented during 2020 have since been softened or lifted in varying degrees in different locations around the world, and the manufacture and distribution of COVID-19 vaccines during 2021 and throughout the first half of 2022 helped to initiate a recovery from the pandemic, the uncertainty regarding new potential variants of COVID-19 and the success of any vaccines in respect thereof, may in the future cause a reduction in global economic activity or prompt the re-imposition of certain restrictions and measures. In addition, even if not required by governmental authorities, increases in COVID-19 cases, such as if a new variant emerges, may result in significantly reduced economic activity, particularly in affected areas, which could result in a sharp reduction in the demand for oil and a decline in oil prices as occurred during 2020.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES AND RECENTLY ISSUED ACCOUNTING STANDARDS APPLICABLE TO THE PARTNERSHIP

There have been no changes to the Partnership's significant accounting policies since December 31, 2021.

New accounting standards not yet implemented.

ASU No. 2020-6 Debt – Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging – Contracts in Entity's Own Equity (Subtopic 815–40) ("ASU 2020-6"). ASU 2020-6 simplifies the accounting for certain financial instruments with characteristics of liabilities and equity, including convertible instruments and contracts on an entity's own equity. The ASU is part of the FASB's simplification initiative, which aims to reduce unnecessary complexity in GAAP. The ASU's amendments are effective for fiscal years beginning after December 15, 2023, and interim periods within those

Table of Contents

fiscal years. The Partnership is currently evaluating the provisions of ASU 2020-6 to determine its impact on the Partnership's consolidated financial statements and disclosures.

ASU No. 2020-4 Reference Rate Reform ("ASU 2020-4"). ASU 2020-4 provides optional expedients and exceptions for applying GAAP to contracts, hedging relationships and other transactions affected by reference rate reform on financial reporting. The amendments in ASU 2020-4 are effective as of March 12, 2020 through December 31, 2022. The Partnership does not expect the provisions of ASU 2020-4 will have a material impact on its consolidated financial statements and disclosures.

3. DISPOSITION

Lane G&P System. On June 30, 2022, the Partnership, completed the disposition of all the equity interests in Summit Permian, which owns the Lane Gathering and Processing System ("Lane G&P System"), to Longwood Gathering and Disposal Systems, LP ("Longwood"), a wholly owned subsidiary of Matador Resources Company ("Matador"), for a cash sale price of \$75.0 million. In connection with the transaction, the Partnership released to a subsidiary of Matador, and Matador agreed to assume, take or-pay firm capacity on the Double E Pipeline. The Partnership is the operator of the Double E joint venture and owns a 70% interest.

During the quarterly period ended June 30, 2022, the Partnership recognized an impairment of \$84.5 million related to the disposition of the Lane G&P System based on total cash proceeds received of \$77.5 million, including \$2.0 million of cash sold in the transaction, net assets of \$160.8 million, and other costs to sell of \$1.2 million. The cash proceeds were used to reduce amounts outstanding under the ABL Facility and are subject to a final working capital adjustment.

4. REVENUE

The following table presents estimated revenue expected to be recognized during the remainder of 2022 and over the remaining contract period related to performance obligations that are unsatisfied and are comprised of estimated minimum volume commitments.

(In thousands)	 2022	 2023	 2024	 2025	2026	Thereafter
Gathering services and related fees	\$ 41.041	\$ 63,214	\$ 52,412	\$ 35,805	\$ 21.024	\$ _

Revenue by Category. In the following tables, revenue is disaggregated by geographic area and major products and services. For more detailed information about reportable segments, see Note 15 – Segment Information.

	Three Months Ended June 30, 2022										
	Gathering services and related fees	Natural gas, NGLs and condensate sales	Other revenues	Total							
		(In the	ousands)								
Reportable Segments:											
Northeast	\$ 12,940	\$ —	\$ —	\$ 12,940							
Rockies	16,510	14,927	4,728	36,165							
Permian	1,822	10,514	1,298	13,634							
Piceance	19,981	2,067	1,276	23,324							
Barnett	10,378	786	1,671	12,835							
Total reportable segments	61,631	28,294	8,973	98,898							
Corporate and other	_	(16)	181	165							
Total	\$ 61,631	\$ 28,278	\$ 9,154	\$ 99,063							

Civ	Month	c Endo	l Impa	20	2022

	_	Gathering services and related fees	Natural gas, NGLs and condensate sales	Other revenues		Total
	_		(In tho	usands)		
Reportable Segments:						
Northeast	9	27,576	\$ —	\$ —	\$	27,576
Rockies		34,299	28,586	9,885		72,770
Permian		3,669	17,381	2,317		23,367
Piceance		40,052	3,962	2,551		46,565
Barnett		20,055	786	3,734		24,575
Total reportable segments	_	125,651	50,715	18,487		194,853
Corporate and other		_	21	315		336
Total	\$	125,651	\$ 50,736	\$ 18,802	\$	195,189

Three	Monthe	Ended	June 30	2021

	Gathe and	ring services related fees	Natural gas, NGLs and condensate sales	Other rever	nues	Total
			(In tl	nousands)		<u> </u>
Reportable Segments:						
Northeast	\$	17,961	\$ —	\$	_	\$ 17,961
Rockies		18,407	8,506		6,098	33,011
Permian		2,262	6,875		1,049	10,186
Piceance		25,527	1,025		1,233	27,785
Barnett		10,076	10		1,012	11,098
Total reportable segments		74,233	16,416		9,392	100,041
Corporate and other		_	_		_	_
Total	\$	74,233	\$ 16,416	\$	9,392	\$ 100,041

Six Months Ended June 30, 2021

		Six World's Ended Julie 30, 2021				
	_	Gathering services and related fees	Natural gas, NGLs and condensate sales	Other revenues	Total	
			(In the	ousands)		
Reportable Segments:						
Northeast	9	32,733	\$ —	\$ —	\$ 32,733	
Rockies		37,303	20,843	11,309	69,455	
Permian		4,461	13,393	1,809	19,663	
Piceance		50,311	2,878	2,409	55,598	
Barnett		19,772	66	2,072	21,910	
Total reportable segments	_	144,580	37,180	17,599	199,359	
Corporate and other		_	_	_	_	
Total	9	5 144,580	\$ 37,180	\$ 17,599	\$ 199,359	

Contract balances. Contract assets relate to the Partnership's rights to consideration for work completed but not billed at the reporting date and consist of the estimated MVC shortfall payments expected from its customers and unbilled activity

associated with contributions in aid of construction. Contract assets are transferred to trade receivables when the rights become unconditional. The following table provides information about contract assets from contracts with customers:

		Total
	(In	thousands)
Contract assets, beginning balance December 31, 2021	\$	10,327
Additions		4,802
Transfers out		(8,867)
Contract assets, ending balance June 30, 2022	\$	6,262

As of June 30, 2022, receivables with customers totaled \$49.9 million and contract assets totaled \$6.3 million and are included in the accounts receivable caption on the unaudited condensed consolidated balance sheets.

As of December 31, 2021, receivables with customers totaled \$50.5 million and contract assets totaled \$10.3 million and are included in the accounts receivable caption on the unaudited condensed consolidated balance sheets.

Contract liabilities (deferred revenue) relate to the advance consideration received from customers primarily for contributions in aid of construction. The Partnership recognizes contract liabilities under these arrangements in revenue over the contract period. See Note 7 – Deferred Revenue for additional details.

5. PROPERTY, PLANT AND EQUIPMENT

Details on the Partnership's property, plant and equipment follow.

	June 30, 2022		Dec	ember 31, 2021
	(In thousands)			
Gathering and processing systems and related equipment	\$	2,077,369	\$	2,225,267
Construction in progress		31,874		49,082
Land and line fill		10,212		10,644
Other		52,472		51,863
Total		2,171,927		2,336,856
Less: accumulated depreciation		(640,477)		(610,774)
Property, plant and equipment, net	\$	1,531,450	\$	1,726,082

Depreciation expense and capitalized interest for the Partnership follow.

	Three Months Ended June 30,		Six Months Ended June 30,			June 30,	
	 2022		2021		2022		2021
			(In tho	usands)	1		
Depreciation expense	\$ 23,714	\$	21,318	\$	47,762	\$	42,784
Capitalized interest	251		149		583		324

6. EQUITY METHOD INVESTMENTS

The Partnership has equity method investments in Double E and Ohio Gathering, the balances of which are included in the Investment in equity method investees caption on the unaudited condensed consolidated balance sheets. Details of the Partnership's equity method investments follow.

	June 30, 2022 December 31,			
	(In thousands)			
Double E	\$ 284,790	\$ 280,952		
Ohio Gathering	234,763	242,244		
Total	\$ 519,553	\$ 523,196		

7. DEFERRED REVENUE

Certain of the Partnership's gathering and/or processing agreements provide for monthly or annual MVCs. The amount of the shortfall payment is based on the difference between the actual throughput volume shipped and/or processed for the applicable period and the MVC for the applicable period, multiplied by the applicable gathering or processing fee.

Many of the Partnership's gas gathering agreements contain provisions that can reduce or delay the cash flows that it expects to receive from MVCs to the extent that a customer's actual throughput volumes are above or below its MVC for the applicable contracted measurement period.

The balances in deferred revenue as of June 30, 2022 and December 31, 2021 are primarily related to contributions in aid of construction which will be recognized as revenue over the life of the contract.

A rollforward of current deferred revenue follows.

		Total
	(In	thousands)
Current deferred revenue, January 1, 2022	\$	10,374
Add: additions		2,509
Less: revenue recognized		(3,163)
Current deferred revenue, June 30, 2022	\$	9,720

A rollforward of noncurrent deferred revenue follows.

	Total	
	(In thousands)	
Noncurrent deferred revenue, January 1, 2022	\$	42,570
Add: additions		269
Less: reclassification to current deferred revenue		(2,477)
Noncurrent deferred revenue, June 30, 2022	\$	40,362

8. DEBT

Debt for the Partnership at June 30, 2022 and December 31, 2021, follows:

	 June 30, 2022	Dece	mber 31, 2021
	(In thou	sands)	
ABL Facility: Summit Holdings' asset based credit facility due May 1, 2026	\$ 151,000	\$	267,000
Permian Transmission Credit Facility : Permian Transmission's variable rate senior secured credit facility due March 8, 2028	_		160,000
Permian Transmission Term Loan: Permian Transmission's variable rate senior			
secured term loan due March 8, 2028	157,758		_
2025 Senior Notes: Summit Holdings' 5.75% senior unsecured notes due April 15, 2025	259,463		259,463
2026 Secured Notes: Summit Holdings' and Finance Corp's 8.50% senior unsecured notes due October 15,			
2026	700,000		700,000
Less: unamortized debt discount and debt issuance costs	(28,183)		(31,391)
Total debt	 1,240,038		1,355,072
Less: current portion	(7,526)		_
Total long-term debt	\$ 1,232,512	\$	1,355,072

ABL Facility. On November 2, 2021, the Partnership, the Partnership's subsidiary, Summit Holdings, and the subsidiaries of Summit Holdings party thereto entered into a first-lien, senior secured credit facility, consisting of a \$400.0 million asset-based revolving credit facility (the "ABL Facility"), subject to a borrowing base comprised of a percentage of eligible accounts receivable of Summit Holdings and its subsidiaries that guarantee the ABL Facility (collectively, the "ABL Facility Subsidiary Guarantors") and a percentage of eligible above-ground fixed assets including eligible compression units, processing plants, compression stations and related equipment of Summit Holdings and the ABL Facility Subsidiary Guarantors. As of June 30,

2022, the most recent borrowing base determination of eligible assets totaled \$584.5 million, an amount greater than the \$400.0 million of aggregate commitments.

The ABL Facility will mature on May 1, 2026; provided that, (a) if the outstanding amount of the 2025 Senior Notes (or any permitted refinancing indebtedness in respect thereof that has a final maturity, scheduled amortization or any other scheduled repayment, mandatory prepayment, mandatory redemption or sinking fund obligation prior to the date that is 120 days after the Termination Date (as defined in the ABL Agreement)) on such date equals or exceeds \$50,000,000, then the ABL Facility will mature on December 13, 2024 and (b) if both (i) any amount of the 2025 Senior Notes (or any permitted refinancing indebtedness in respect thereof that has a final maturity, scheduled amortization or any other scheduled repayment, mandatory prepayment, mandatory redemption or sinking fund obligation prior to the date that is 120 days after the Termination Date) is outstanding on such date and (ii) Liquidity (as defined in the ABL Agreement) is less than an amount equal to the sum of the then aggregate outstanding principal amount of the 2025 Senior Notes (or any permitted refinancing indebtedness in respect thereof that has a final maturity, scheduled amortization or any other scheduled repayment, mandatory prepayment, mandatory redemption or sinking fund obligation prior to the date that is 120 days after the Termination Date) plus the Threshold Amount (as defined in the ABL Agreement) on such date, then the ABL Facility will mature on January 14, 2025.

As of June 30, 2022, the applicable margin under the adjusted LIBOR borrowings was 3.25%, the interest rate was 4.375% and the available borrowing capacity of the ABL Facility totaled \$243.1 million after giving effect to the issuance of \$5.9 million in outstanding but undrawn irrevocable standby letters of credit.

The ABL Facility requires that Summit Holdings not permit (i) the First Lien Net Leverage Ratio (as defined in the ABL Agreement) as of the last day of any fiscal quarter to be greater than 2.50:1.00, or (ii) the Interest Coverage Ratio (as defined in the ABL Agreement) as of the last day of any fiscal quarter to be less than 2.00:1.00. As of June 30, 2022, the First Lien Net Leverage Ratio was 0.64:1.00 and the Interest Coverage Ratio was 2.73:1.00 As of June 30, 2022, the Partnership was in compliance with the financial covenants of the ABL Facility.

Permian Transmission Credit Facility. On March 8, 2021 (the "Permian Closing Date"), the Partnership's unrestricted subsidiary, Permian Transmission, entered into a Credit Agreement which provided for \$175.0 million of senior secured credit facilities (the "Permian Transmission Credit Facilities"), including a \$160.0 million Term Loan Facility and a \$15.0 million Working Capital Facility. Borrowings from the Permian Transmission Credit Facilities were used to finance Permian Transmission's capital calls associated with its investment in Double E. In January 2022, in accordance with the terms of Permian Transmission Credit Facilities, the Partnership converted the \$160.0 million portion of the Permian Transmission Credit Facilities into a term loan facility (the "Permian Term Loan Facility").

As of June 30, 2022, the applicable margin under adjusted LIBOR borrowings was 2.375%, the interest rate was 3.4% and the available borrowing capacity of the Permian Transmission Credit Facility totaled \$7.5 million, subject to a commitment fee of 0.7% after giving effect to the issuance of \$7.5 million in outstanding but undrawn irrevocable standby letters of credit and surety bonds. As of June 30, 2022, the Partnership was in compliance with the financial covenants of the Permian Transmission Credit Facility.

Permian Transmission Term Loan. As described above, in January 2022, the Permian Transmission Term Loan Facility was converted into a Term Loan (the "Permian Transmission Term Loan"). The Permian Transmission Term Loan is due on May 8, 2028. As of June 30, 2022, the applicable margin under adjusted LIBOR borrowings was 2.375% and the interest rate was 3.4%. As of June 30, 2022, the Partnership was in compliance with the financial covenants governing the Permian Transmission Term Loan.

In accordance with the terms of the Permian Transmission Term Loan, Permian Transmission is required to make mandatory principal repayments. Below is a summary of the remaining mandatory principal repayments as of June 30, 2022:

(In thousands)	Total	2022	2023	2024	2025	2026	Thereafter
Amortizing principal repayments	\$ 157,758	\$ 2,405	\$ 10,507	\$ 15,524	\$ 16,580	\$ 16,967	\$ 95,775

2026 Secured Notes. In November 2021, the Co-Issuers issued the 2026 Secured Notes. The 2026 Secured Notes are senior secured second lien notes due 2026. The 2026 Secured Notes will pay interest semi-annually in cash in arrears on April 15 and October 15 of each year, commencing on April 15, 2022, and will be jointly and severally guaranteed, on a senior second-priority secured basis (subject to permitted liens), by the Partnership and each restricted subsidiary of the Partnership (other than the Co-Issuers) that is an obligor under the ABL Agreement, or under the Co-Issuers' 2025 Senior Notes on the issue date of the 2026 Secured Notes.

The 2026 Secured Notes are effectively subordinated to any of our or the guarantors' current and future secured first lien indebtedness, including indebtedness incurred under the ABL Facility, to the extent of the value of the collateral securing such indebtedness, and our and the guarantors' current and future debt that is secured by liens on assets other than the collateral, to the extent of the value of such assets. The 2026 Secured Notes are structurally subordinated to all indebtedness and other

liabilities of our subsidiaries that do not guarantee the 2026 Secured Notes. The 2026 Secured Notes are effectively equal to our and the guarantors' obligations under any future second lien indebtedness and effectively senior to all of our future junior lien indebtedness and existing and future unsecured indebtedness, including our outstanding senior unsecured notes, to the extent of the value of the collateral, and senior to any of our future subordinated indebtedness. The 2026 Secured Notes will mature on October 15, 2026.

Before October 15, 2023, the Co-Issuers may redeem the 2026 Secured Notes, in whole or in part, at a price equal to 100% of their principal amount, plus a make-whole premium, together with accrued and unpaid interest to, but not including the redemption date. On and after October 15, 2023, the Co-Issuers may redeem all or part of the 2026 Secured Notes at redemption prices (expressed as percentages of principal amount) equal to: (a) 104.250% for the twelve-month period beginning October 15, 2023; (b) 102.125% for the twelve-month period beginning October 15, 2024; and (c) 100.000% for the twelve-month period beginning on October 15, 2025 and at any time thereafter, in each case plus accrued and unpaid interest, if any, to, but not including, the redemption date. As of June 30, 2022, the Partnership was in compliance with the financial covenants governing its 2026 Secured Notes.

2025 Senior Notes. In February 2017, the Co-Issuers co-issued the 2025 Senior Notes. The 2025 Senior Notes will pay interest semi-annually in cash in arrears on April 15 and October 15 of each year. The 2025 Senior Notes are senior, unsecured obligations and rank equally in right of payment with all of the Partnership's existing and future senior obligations. The 2025 Senior Notes are effectively subordinated in right of payment to all of the Partnership's secured indebtedness, to the extent of the collateral securing such indebtedness.

The Co-Issuers have the right to redeem all or part of the 2025 Senior Notes at a redemption price of 101.438% (with the redemption price declining to 100.00% on April 15, 2023 and thereafter), plus accrued and unpaid interest, if any, to, but not including the redemption date.

As of June 30, 2022, the Partnership was in compliance with the financial covenants governing its 2025 Senior Notes.

9. FINANCIAL INSTRUMENTS

Fair Value. A summary of the estimated fair value of our financial instruments follows.

	June 30, 2022					Decembe	r 31,	1, 2021	
		arrying Value ⁽¹⁾		Estimated fair value (Level 2)		Carrying Value ⁽¹⁾		Estimated fair value (Level 2)	
				(In the	usand	ls)		_	
2025 Senior Notes	\$	259,463	\$	203,354	\$	259,463	\$	234,814	
2026 Secured Notes		700,000		628,250		700,000		718,083	

⁽¹⁾ Excludes applicable unamortized debt issuance costs and debt discounts.

The carrying values on the balance sheets of the ABL Facility and Permian Transmission Term Loan represents their fair value due to its floating interest rate. The fair values of the 2026 Secured Notes and 2025 Senior Notes are based on an average of nonbinding broker quotes as of June 30, 2022 and December 31, 2021. The use of different market assumptions or valuation methodologies may have a material effect on their estimated fair value.

Interest Rate Swaps. In connection with the Permian Transmission Term Loan, formerly the Permian Transmission Credit Facility, the Partnership entered into amortizing interest rate swap agreements. As of June 30, 2022 and December 31, 2021, the outstanding notional amounts of interest rate swaps were \$142.0 million and \$144.0 million, respectively. These interest rate swaps manage exposure to variability in expected cash flows attributable to interest rate risk. Interest rate swaps convert a portion of the Partnership's variable rate debt to fixed rate debt. The Partnership chooses counterparties for its derivative instruments that it believes are creditworthy at the time the transactions are entered into, and the Partnership actively monitors the creditworthiness where applicable. However, there can be no assurance that a counterparty will be able to meet its obligations to the Partnership. The Partnership presents its derivative positions on a gross basis and does not net the asset and liability positions.

As of June 30, 2022, the Partnership's interest rate swap agreements had a fair value of \$10.8 million and are recorded within other noncurrent assets, other noncurrent liabilities and other current liabilities within the unaudited condensed consolidated balance sheets. Changes in the derivative instruments' fair value are determined using level 2 inputs from the fair value hierarchy. For the three and six months ended June 30, 2022, the Company recorded a gain on interest rate swaps of \$3.9 million and \$11.0 million, respectively.

10. PARTNERS' CAPITAL AND MEZZANINE CAPITAL

Common Units. A rollforward of the number of issued and outstanding common limited partner units follows for the period from December 31, 2021 to June 30, 2022.

	Common Units
Units, December 31, 2021	7,169,834
2022 Preferred Exchange Offer, net of units withheld for taxes	2,853,875
Common units issued for SMLP LTIP, net	142,466
Units, June 30, 2022	10,166,175

Series A Preferred Units. As of June 30, 2022, the Partnership had 65,508 Series A Preferred Units outstanding and \$17.4 million of accrued and unpaid distributions on its Series A Preferred Units.

Series A Preferred Unit Exchange Offers. In January 2022, the Partnership completed an offer to exchange its Series A Preferred Units for newly issued common units (the "2022 Preferred Exchange Offer"), whereby it issued 2,853,875 SMLP common units, net of units withheld for withholding taxes, in exchange for 77,939 Series A Preferred Units. As a result of the transaction, the Partnership recognized a deemed contribution of \$20.9 million from the Series A Preferred Unit holders to the common unit holders.

In April 2021, the Partnership completed an offer to exchange its Series A Preferred Units for newly issued common units (the "2021 Preferred Exchange Offer"), whereby it issued 538,715 SMLP common units, net of units withheld for withholding taxes, in exchange for 18,662 Series A Preferred Units. As a result of the transaction, the Partnership recognized a deemed contribution of \$8.3 million from the Series A Preferred Unit holders to the common unit holders.

Subsidiary Series A Preferred Units. The Partnership records its Subsidiary Series A Preferred Units at fair value upon issuance, net of issuance costs, and subsequently records an effective interest method accretion amount each reporting period to accrete the carrying value to a most probable redemption value that is based on a predetermined internal rate of return measure. If the Partnership elects to make payment-in-kind ("PIK") distributions to holders of its Subsidiary Series A Preferred Units, these PIK distributions increase the liquidation preference on each Subsidiary Series A Preferred Unit. Net Income (Loss) attributable to our common units includes adjustments for PIK distributions and redemption accretion.

During the six months ended June 30, 2022, the Partnership accrued a \$1.6 million of cash distribution, which was paid in July 2022, and elected to make PIK distributions, issuing 1,600 Subsidiary Series A Preferred Units, to the holders of its Subsidiary Series A Preferred Units. As of June 30, 2022, the Partnership had 93,039 Subsidiary Series A Preferred Units issued and outstanding.

If the Subsidiary Series A Preferred Units were redeemed on June 30, 2022, the redemption amount would be \$117.6 million when considering the applicable multiple of invested capital metric and make-whole amount provisions contained in the Subsidiary Series A Preferred Unit agreement.

The following table shows the change in our Subsidiary Series A Preferred Unit balance from January 1, 2022 to June 30, 2022, net of \$2.5 million and \$3.9 million of unamortized issuance costs at June 30, 2022 and December 31, 2021, respectively:

	(In	thousands)
Balance at December 31, 2021	\$	106,325
PIK distributions		1,600
Redemption accretion, net of issuance cost amortization		7,277
Cash distribution		(1,628)
Balance at June 30, 2022	\$	113,574

Warrants. On May 28, 2020, and in connection with the GP Buy-In Transaction, the Partnership issued (i) a warrant to purchase up to 537,307 SMLP common units (8,059,609 SMLP common units prior to the Reverse Unit Split) to SMP TopCo, LLC, a Delaware limited liability company and affiliate of ECP ("ECP NewCo") (the "ECP NewCo Warrant"), and (ii) a warrant to purchase up to 129,360 SMLP common units (1,940,391 SMLP common units prior to the Reverse Unit Split) to SMLP Holdings, LLC, a Delaware limited liability company and affiliate of ECP ("ECP Holdings" and together with ECP NewCo, the "ECP Entities") (the "ECP Holdings Warrant" and together with the ECP NewCo Warrant, the "ECP Warrants").

On August 5, 2021, the ECP Entities cashlessly exercised all of the ECP Warrants for an aggregate of 414,447 SMLP common units, net of the exercise price, as calculated pursuant to Section 3(c) of the ECP Warrants (the "ECP Warrant Exercise").

Cash Distribution Policy. In connection with the GP Buy-In Transaction, the Partnership suspended its cash distributions to holders of its common units, commencing with respect to the quarter ended March 31, 2020. Upon the resumption of distributions, the Partnership Agreement requires that it distribute all available cash, subject to reserves established by its General Partner, within 45 days after the end of each quarter to unitholders of record on the applicable record date. The amount of distributions paid under this policy is subject to fluctuations based on the amount of cash the Partnership generates from its business and the decision to make any distribution is determined by the General Partner, taking into consideration the terms of the Partnership Agreement. There were no distributions paid during the three and six months ended June 30, 2022 or during the twelve months ended December 31, 2021.

11. EARNINGS PER UNIT

The following table details the components of EPU.

Three Months Ended June 30, Six Months Ended							June 30,	
	2022		2021		2022		2021	
		(In	thousands, excep	pt pe	r-unit amounts)			
\$	(91,804)	\$	(19,738)	\$	(91,809)	\$	(10,750)	
	(3,164)		(4,089)		(8,877)		(8,021)	
\$	(94,968)	\$	(23,827)	\$	(100,686)		(18,771)	
\$	(1,888)	\$	(3,849)	\$	(4,108)		(8,136)	
	_		_		20,974		_	
\$	_	\$	8,326	\$	_	\$	8,326	
\$	(96,856)	\$	(19,350)	\$	(83,820)	\$	(18,581)	
	10,166		6,656		9,919		6,392	
	_		_		_		_	
	10,166		6,656		9,919		6,392	
\$	(9.53)	\$	(2.91)	\$	(8.45)	\$	(2.91)	
\$	(9.53)	\$	(2.91)	\$	(8.45)	\$	(2.91)	
	_		169		_		173	
	\$ \$ \$ \$	\$ (91,804) (3,164) \$ (94,968) \$ (1,888) — \$ — \$ (96,856) 10,166 ———————————————————————————————————	\$ (91,804) \$ (3,164) \$ (94,968) \$ \$ (1,888) \$ — \$ \$ — \$ \$ (96,856) \$ \$ 10,166 — 10,1	\$ (91,804) \$ (19,738) (3,164) (4,089) \$ (94,968) \$ (23,827) \$ (1,888) \$ (3,849) — — \$ 8,326 \$ (96,856) \$ (19,350) 10,166 6,656 — — — — — — — — — — — — — — — — — — —	\$ (91,804) \$ (19,738) \$ (3,164) (4,089) \$ (94,968) \$ (23,827) \$ \$ (1,888) \$ (3,849) \$ \$ (96,856) \$ (19,350) \$ \$ (10,166) \$ (6,656) \$ (10,166) \$ (6,656) \$ (10,166) \$	2022 2021 2022	2022 2021 2022	

12. SUPPLEMENTAL CASH FLOW INFORMATION

	Six Months E	Six Months Ended Ju 2022 (In thousands) 42,039 \$ 149 \$ 3,738 \$ 7,277 \$		
	 2022		2021	
	 (In tho	usands)	
Supplemental cash flow information:				
Cash interest paid	\$ 42,039	\$	27,869	
Cash paid for taxes	\$ 149	\$	15	
Noncash investing and financing activities:				
Capital expenditures in trade accounts payable (period-end accruals)	\$ 3,738	\$	6,059	
Accretion of Subsidiary Series A Preferred Units, net of issuance cost amortization	\$ 7,277	\$	5,008	
2022 Preferred Exchange Offer	\$ 92,587	\$	_	
2021 Preferred Exchange Offer	\$ _	\$	20,654	

13. UNIT-BASED AND NONCASH COMPENSATION

SMLP Long-Term Incentive Plan. The Partnership's Long-Term Incentive Plan ("SMLP LTIP") provides for equity awards to eligible officers, employees, consultants and directors of the Partnership, thereby linking the recipients' compensation directly to SMLP's performance. Significant items to note:

- For the six-month period ended June 30, 2022, the Partnership granted 146,529 phantom units and associated distribution equivalent rights to employees in connection with the Partnership's annual incentive compensation award cycle. These awards had a grant date fair value of \$14.21 per common unit and vest ratably over a 3-year period.
- For the six-month period ended June 30, 2022, the Partnership issued 38,664 common units to the Partnership's six independent directors in connection with their annual compensation plan. These awards had a grant date fair value of \$14.21 per common unit and vested immediately.
- On June 24, 2022, unitholders of the Partnership approved the Summit Midstream Partners, LP 2022 Long-Term Incentive Plan, which increased the number of common units available for issuance to the Partnership's employees, consultants and directors. As of June 30, 2022, approximately 1.2 million common units remained available for future issuance under the SMLP LTIP, which includes the impact of 0.3 million granted but unvested phantom units.

14. COMMITMENTS AND CONTINGENCIES

Environmental Matters. Although the Partnership believes that it is in material compliance with applicable environmental regulations, the risk of environmental remediation costs and liabilities are inherent in pipeline ownership and operation. Furthermore, the Partnership can provide no assurances that significant environmental remediation costs and liabilities will not be incurred in the future. The Partnership is currently not aware of any material contingent liabilities that exist with respect to environmental matters, except as noted below.

As of June 30, 2022, the Partnership has recognized (i) a current liability for remediation effort expenditures expected to be incurred within the next 12 months and (ii) a noncurrent liability for estimated remediation expenditures expected to be incurred subsequent to June 30, 2023. Each of these amounts represent the Partnership's best estimate for costs expected to be incurred. Neither of these amounts have been discounted to their present value.

A rollforward of the Partnership's undiscounted accrued environmental remediation follows and is primarily related to the 2015 Blacktail Release and other environmental remediation activities, as detailed below.

	(In t	thousands)
Accrued environmental remediation, December 31, 2021	\$	5,606
Payments made		(1,883)
Changes in estimates		657
Accrued environmental remediation, June 30, 2022	\$	4,380

Table of Contents

In 2015, the Partnership learned of the rupture of a four-inch produced water gathering pipeline on the Meadowlark Midstream system near Williston, North Dakota ("2015 Blacktail Release"). On August 4, 2021, subsidiaries of the Partnership entered into the following agreements to resolve the U.S. federal and North Dakota state governments' environmental claims with respect to the 2015 Blacktail Release: (i) a Consent Decree with the U.S. Department of Justice ("DOJ"), the U.S. Environmental Protection Agency ("EPA"), and the State of North Dakota ("Consent Decree"); (ii) a Plea Agreement with the United States ("Plea Agreement"); and (iii) a Consent Agreement with the North Dakota Industrial Commission ("Consent Agreement" together with the Consent Decree and Plea Agreement, the "Global Settlement"). As of June 30, 2022 and December 31, 2021, the accrued loss liability for the 2015 Blacktail Release was \$33.2 million.

Key terms of the Global Settlement included (i) payment of penalties and fines totaling \$36.3 million, consisting of \$1.25 million in natural resource damages payable to federal and state governments, a \$25.0 million payable to the federal government over 5 years, and a \$10.0 million payable to state governments over six years, with interest applied to unpaid amounts accruing at a fixed rate of 3.25%, and of which \$5.6 million is expected to be paid within the next twelve months; (ii) continuation of remediation efforts at the site of the 2015 Blacktail Release; (iii) other injunctive relief including but not limited to control room management, environmental management system audit, training, and reporting; (iv) guilty pleas by Defendant subsidiary for (a) one charge of negligent discharge of a harmful quantity of oil and (b) one charge of knowing failure to immediately report a discharge of oil; and (v) organizational probation for a minimum period of three years from sentencing, including payment in full of certain components of the fines and penalty amounts. The agreements comprising the Global Settlement were subject to the approval of the U.S. District Court for the District of North Dakota (the "U.S. District Court"). The U.S. District Court entered an order making the civil components of the Global Settlement effective on September 28, 2021 and accepted the sentencing in the Plea Agreement on December 6, 2021, completing approval of the Global Settlement.

In conjunction with the criminal proceedings under the Plea Agreement, the U.S. District Court received two claims for restitution, including claims for diminution in property value and the cost of additional environmental testing. The prosecution notified the U.S District Court that the government declined to support the claims, citing "insufficient evidence to support the claims." Defendant subsidiary of the Partnership has responded that restitution is not applicable in this matter because claimants did not provide any supporting evidence for their claims that showed any harm, and because the plea agreement in this matter does not permit restitution sought by the claimants. On August 4, 2022, the U.S. District Court denied the claims.

Subsidiaries of the Partnership are also participating in two proceedings before the EPA as a result of the Plea Agreement becoming effective. Following the U.S. District Court's entering judgment on Defendant subsidiary's guilty plea to one count of negligent discharge of produced water in violation of the Clean Water Act, Defendant subsidiary was statutorily debarred by operation of law pursuant to 33 U.S.C. § 1368(a) to participate in federal awards performed at the "violating facility," which EPA determined to be the Marmon subsystem of the produced water gathering system in North Dakota. Defendant has submitted a petition for reinstatement, which was denied by the EPA's suspension and debarment office ("SDO") on July 11, 2022. The SDO determined that the term of probation in the Plea Agreement was the appropriate period of time to demonstrate Defendant subsidiary's change of corporate attitude, policies, practices, and procedures. The Partnership and certain subsidiaries have also received a show cause notice from the EPA requesting us to "show cause" why SDO should not issue a Notice of Proposed Debarment to the Defendant subsidiary and certain affiliates under 2 C.F.R. § 180.800(d), to which we have responded.

Legal Proceedings. The Partnership is involved in various litigation and administrative proceedings arising in the ordinary course of business. In the opinion of management, any liabilities that may result from these claims or those arising in the ordinary course of business would not individually or in the aggregate have a material adverse effect on the Partnership's financial position or results of operations.

15. SEGMENT INFORMATION

In accordance with ASC No. 280 - Segment Reporting, the Partnership routinely evaluates whether its reportable segments have changed. In the fourth quarter of 2021, the Partnership changed its segment reporting to align with how the General Partner's Chief Executive Officer, its chief operating decision maker, reviews financial information in order to allocate resources and assess performance. The new segment reporting as of December 31, 2021, resulted from changes enacted to optimize commercial efforts and geographic workforce in order to better align its commercial, engineering, and operational capabilities. As a result of the revised reportable segment structure, the Partnership has recast the corresponding segment information for all periods presented.

As of June 30, 2022, the Partnership's reportable segments are:

• Rockies – Includes the Partnership's wholly owned midstream assets located in the Williston Basin and the DJ Basin.

- **Permian** Includes the Partnership's equity method investment in Double E and its wholly owned Lane G&P System, which was sold on June 30, 2022 to Matador. For additional information regarding the sale of the Lane G&P System, see Note 3 Disposition.
- **Northeast** Includes the Partnership's wholly owned midstream assets located in the Utica and Marcellus shale plays and the equity method investment in Ohio Gathering that is focused on the Utica Shale.
- Piceance Includes the Partnership's wholly owned midstream assets located in the Piceance Basin.
- Barnett Includes the Partnership's wholly owned midstream assets located in the Barnett Shale.

Corporate and Other represents those results that: (i) are not specifically attributable to a reportable segment; (ii) are not individually reportable; or (iii) have not been allocated to a reportable segment for the purpose of evaluating their performance, including certain general and administrative expense items, certain natural gas and crude oil marketing services and transaction costs.

Assets by reportable segment follow.

	 (In thousands) 610,167 \$ 623, 576,087 592, 300,994 458, 501,137 524, 305,855 315, 2,294,240 2,513, 16,093 8,		December 31, 2021	
	(In thousands) 610,167 \$ 623,2 576,087 592,1 300,994 458,9 501,137 524,2 305,855 315,0 2,294,240 2,513,6			
Assets:				
Northeast	\$ 610,167	\$	623,224	
Rockies	576,087		592,148	
Permian	300,994		458,988	
Piceance	501,137		524,218	
Barnett	305,855		315,055	
Total reportable segment assets	2,294,240		2,513,633	
Corporate and Other	16,093		8,829	
Total assets	\$ 2,310,333	\$	2,522,462	

Segment adjusted EBITDA by reportable segment follows.

	Three Months Ended June 30,				Six Months E	nded	June 30,
	 2022		2021		2022		2021
			(In tho	usands)		
Reportable segment adjusted EBITDA							
Northeast	\$ 18,568	\$	23,361	\$	38,636	\$	43,554
Rockies	13,899		14,732		29,729		30,884
Permian	4,817		1,341		8,966		2,592
Piceance	15,350		20,324		31,118		41,358
Barnett	7,247		8,889		16,533		16,905
Total of reportable segments' measures of profit	\$ 59,881	\$	68,647	\$	124,982	\$	135,293

A reconciliation of income or loss before income taxes and income from equity method investees to total of reportable segments' measures of profit follows.

	Three Months	2021 2022 (In thousands) (95,872) \$ (22,290) \$ (99,862) \$ 5,809 41,144 9,627 24,887 15,502 49,050 30,346 28,598 61,025 11,406 6,841 21,858 (1,578) (2,225) (3,306)			Ionths Ended June 30,		
	2022	2021		2022	2021		
_		(In th	ousands))			
Reconciliation of income (loss) before income taxes and							
income from equity method investees to total of							
reportable segments' measures of profit:							
Income (loss) before income taxes and income from equity method investees	(95,872)	\$ (22,290)	\$	(99,862) \$	(15,631)		
Add:							
Corporate and Other expense	5,809	41,144		9,627	49,482		
Interest expense	24,887	15,502		49,050	29,455		
Depreciation and amortization (1)	30,346	28,598		61,025	57,344		
Proportional adjusted EBITDA for equity method investees	11,406	6,841		21,858	13,713		
Adjustments related to capital reimbursement activity	(1,578)	(2,225))	(3,306)	(3,470)		
Unit-based and noncash compensation	582	1,048		2,272	3,015		
Gain on asset sales, net	(313)	(4))	(310)	(140)		
Long-lived asset impairment	84,614	33		84,628	1,525		
Total of reportable segments' measures of profit	59,881	\$ 68,647	\$	124,982 \$	135,293		

⁽¹⁾ Includes the amortization expense associated with our favorable gas gathering contracts as reported in other revenues.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

This Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") is intended to inform the reader about matters affecting the financial condition and results of operations of the Partnership and its subsidiaries for the periods since December 31, 2021. As a result, the following discussion should be read in conjunction with the unaudited condensed consolidated financial statements and notes thereto included in this report and the MD&A and the audited consolidated financial statements and related notes that are included in the Partnership's Annual Report on Form 10-K for the year ended December 31, 2021 (the "2021 Annual Report"). Among other things, those financial statements and the related notes include more detailed information regarding the basis of presentation for the following information. This discussion contains forward-looking statements that constitute our plans, estimates and beliefs. These forward-looking statements involve numerous risks and uncertainties, including, but not limited to, those discussed in Forward-Looking Statements. Actual results may differ materially from those contained in any forward-looking statements.

Overview

We are a value-driven limited partnership focused on developing, owning and operating midstream energy infrastructure assets that are strategically located in unconventional resource basins, primarily shale formations, in the continental United States.

Our financial results are driven primarily by volume throughput across our gathering systems and by expense management. We generate the majority of our revenues from the gathering, compression, treating and processing services that we provide to our customers. A majority of the volumes that we gather, compress, treat and/or process have a fixed-fee rate structure which enhances the stability of our cash flows by providing a revenue stream that is not subject to direct commodity price risk. We also earn a portion of our revenues from the following activities that directly expose us to fluctuations in commodity prices: (i) the sale of physical natural gas and/or NGLs purchased under percentage-of-proceeds or other processing arrangements with certain of our customers in the Rockies, and Piceance segments, (ii) the sale of natural gas we retain from certain Barnett segment customers, (iii) the sale of condensate we retain from our gathering services in the Piceance segment and (iv) additional gathering fees that are tied to the performance of certain commodity price indexes which are then added to the fixed gathering rates.

We also have indirect exposure to changes in commodity prices such that persistently low commodity prices may cause our customers to delay and/or cancel drilling and/or completion activities or temporarily shut-in production, which would reduce the volumes of natural gas and crude oil (and associated volumes of produced water) that we gather. If certain of our customers cancel or delay drilling and/or completion activities or temporarily shut-in production, the associated MVCs, if any, ensure that we will earn a minimum amount of revenue. Commodity prices have increased and remain at higher levels primarily due to Russia's invasion of Ukraine beginning in February 2022, which mitigates the risk of cancelled or delayed drilling and/or completion activities.

The following table presents certain consolidated and reportable segment financial data. For additional information on our reportable segments, see the "Segment Overview for the Three and Six Months Ended June 30, 2022 and 2021" section included herein.

	Three Months	Ended J	une 30,	Six Months Ended June 30,			
	2022		2021		2022		2021
	(In tho	usands)			(In tho	usands	s)
Net loss	\$ (91,804)	\$	(19,738)	\$	(91,809)	\$	(10,750)
Reportable segment adjusted EBITDA							
Northeast	\$ 18,568	\$	23,361	\$	38,636	\$	43,554
Rockies	13,899		14,732		29,729		30,884
Permian	4,817		1,341		8,966		2,592
Piceance	15,350		20,324		31,118		41,358
Barnett	7,247		8,889		16,533		16,905
Net cash provided by operating activities	\$ 14,113	\$	34,787	\$	60,159	\$	86,217
Capital expenditures (1)	6,091		3,352		14,794		5,962
Proceeds from the disposition of Lane G&P System, net of cash sold in							
transaction	75,520		_		75,520		_
Investment in Double E equity method investee (2)	_		43,324		8,444		48,943
Repayments on ABL Facility	(105,000)		_		(139,000)		
Repayments on Revolving Credit Facility	_		(40,000)		_		(95,000)
Borrowings on ABL Facility	23,000		_		23,000		_
Repayments on Permian Transmission Term Loan	(1,147)		_		(2,242)		_
Borrowings under Permian Transmission Credit Facility	_		36,000		_		53,500

⁽¹⁾ See "Liquidity and Capital Resources" herein to the unaudited condensed consolidated financial statements for additional information on capital expenditures.

Trends and Outlook

Our business has been, and we expect our future business to continue to be, affected by the following key trends:

- Ongoing impact of the COVID-19 pandemic and fluctuations in demand for oil and natural gas;
- Ongoing impact of the current Russia-Ukraine conflict and the international sanctions against Russia on commodity prices;
- · Natural gas, NGL and crude oil supply and demand dynamics;
- Production from U.S. shale plays;
- · Capital markets availability and cost of capital; and
- Inflation and shifts in operating costs.

Our expectations are based on assumptions made by us and information currently available to us. To the extent our underlying assumptions about, or interpretations of, available information prove to be incorrect, our actual results may vary materially from our expected results. For additional information, see the "Trends and Outlook" section of MD&A included in the 2021 Annual Report.

⁽²⁾ Inclusive of \$0.6 and \$1.6 million, respectively, of capitalized interest for the three and six months ended June 30, 2021. There was no capitalized interest recorded for the three and six months ended June 30, 2022.

Ongoing impact of the COVID-19 pandemic and fluctuations in demand for oil and natural gas. We continue to closely monitor the impact of the COVID-19 pandemic, including its variants, on all aspects of our business, including how it has impacted and will impact our customers, employees, supply chain and distribution network. In response to the COVID-19 pandemic, we modified our business practices, including restricting employee travel and utilizing COVID-19 pandemic tax relief in 2021 (as allowed by the Consolidated Appropriations Act, 2021, the "ERC Tax Credit"). We are unable to predict the broader implications of and the ultimate impact that COVID-19, its variants and related factors may have on our business, future results of operations, financial position or cash flows. Given the dynamic nature of the COVID-19 pandemic and related market conditions, we cannot reasonably estimate the period of time that these events will persist or the full extent of the impact they will have on our business. The extent to which our operations continue to be impacted by the COVID-19 pandemic will depend largely on future developments, which remain highly uncertain and cannot be accurately predicted, including changes in the severity of the pandemic, countermeasures taken by governments, businesses and individuals to slow the spread of the pandemic, and the development and availability of treatments and the extent to which these treatments and vaccines may remain effective as potential new strains of COVID-19 emerge.

We have collaborated extensively with our customer base regarding impacts to their drilling and completion activities in light of the COVID-19 pandemic. Based on recently updated production forecasts and 2022 development plans from our customers, we currently expect that 2022 activity will be lower than historical periods prior to COVID-19, notwithstanding the recent increase in commodity prices primarily driven by Russia's invasion of Ukraine in February 2022.

Ongoing impact of the current Russia-Ukraine conflict and the international sanctions against Russia on commodity prices. In February of 2022, Russia invaded Ukraine and is still engaged in active armed conflict against the country. As a result, governments in the European Union, the United States and other countries have enacted additional sanctions against Russia, certain of its citizens and Russian interests.

Although the Partnership does not operate in Ukraine, Russia or other parts of Europe, there are certain impacts arising from Russia's invasion of Ukraine that could have a potential effect on the Partnership, including, but not limited to, volatility in currencies and commodity prices, higher inflation, cost and supply chain pressures and availability and disruptions in banking systems and capital markets. As of the date of filing, there have been no material impacts.

Cost structure optimization and portfolio management. The Partnership intends to optimize its capital structure in the future by reducing its indebtedness with free cash flow, and when appropriate, it may pursue opportunistic transactions with the objective of increasing long term unitholder value. This may include opportunistic acquisitions, divestitures (such as the disposition of the Lane G&P System), re-allocation of capital to new or existing areas, and development of joint ventures involving our existing midstream assets or new investment opportunities. We believe that our internally generated cash flow, our ABL Facility, the Permian Term Loan Facility, and access to debt or equity will be adequate to finance our strategic initiatives. To attain our overall corporate strategic objectives, we may conduct an asset divestiture, or divestitures, at a transaction valuation that is less than the net book value of the divested asset.

How We Evaluate Our Operations

We conduct and report our operations in the midstream energy industry through five reportable segments: Northeast, Rockies, Permian, Piceance and Barnett. Each of our reportable segments provides midstream services in a specific geographic area and our reportable segments reflect the way in which we internally report the financial information used to make decisions and allocate resources in connection with our operations. For additional information see Note 15 - Segment Information.

Our management uses a variety of financial and operational metrics to analyze our consolidated and segment performance. We view these metrics as important factors in evaluating our profitability. These metrics include:

- throughput volume;
- revenues;
- · operation and maintenance expenses;
- · capital expenditures; and
- segment adjusted EBITDA.

We review these metrics on a regular basis for consistency and trend analysis. There have been no changes in the composition or characteristics of these metrics during the three and six months ended June 30, 2022.

Additional Information. For additional information, see the "Results of Operations" section herein and the notes to the unaudited condensed consolidated financial statements. For additional information on how these metrics help us manage our business, see the "How We Evaluate Our Operations" section of MD&A included in the 2021 Annual Report. For information on impending accounting changes that are expected to materially impact our financial results reported in future periods, see Note 2 – Summary of Significant Accounting Policies and Recently Issued Accounting Standards applicable to the Partnership.

Results of Operations

Consolidated Overview for the Three and Six Months Ended June 30, 2022 and 2021

The following table presents certain consolidated financial and operating data.

	Three Months	Ended June 30,	Six Months Ended June 30,			
	 2022	2021	2022	2021		
	(In tho	usands)	(In tho	usands)		
Revenues:						
Gathering services and related fees	\$ 61,631	\$ 74,233	\$ 125,651	\$ 144,	,580	
Natural gas, NGLs and condensate sales	28,278	16,416	50,736	37,	,180	
Other revenues	 9,154	9,392	 18,802	17,	,599	
Total revenues	99,063	100,041	195,189	199,	,359	
Costs and expenses:	_					
Cost of natural gas and NGLs	26,831	16,626	49,082	37,	,102	
Operation and maintenance	22,277	17,507	39,339	34,	,100	
General and administrative ⁽²⁾	10,473	29,360	23,433	39,	,938	
Depreciation and amortization	30,111	28,364	60,556	56,	,875	
Transaction costs	(13)	450	233		217	
Gain on asset sales, net	(313)	(4)	(310)	((140)	
Long-lived asset impairment	84,614	33	84,628	1,	,525	
Total costs and expenses	173,980	92,336	256,961	169,	,617	
Other income (expense), net	 (4)	352	(4)		408	
Gain (loss) on interest rate swaps	3,936	(2,686)	10,964	(2,	,692)	
Loss on ECP Warrants	_	(12,159)	_	(13,	,634)	
Interest expense	(24,887)	(15,502)	(49,050)	(29,	,455)	
Loss before income taxes and equity method investment income	 (95,872)	(22,290)	 (99,862)	(15,	,631)	
Income tax (expense) benefit	(325)	248	(375)		262	
Income from equity method investees	4,393	2,304	8,428	4,	,619	
Net loss	\$ (91,804)	\$ (19,738)	\$ (91,809)	\$ (10,	,750)	
V. 1 (1)						
Volume throughput (1):						
Aggregate average daily throughput - natural gas (MMcf/d)	1,200	1,441	1,254	1,	,393	
Aggregate average daily throughput - liquids (Mbbl/d)	54	63	60		64	

⁽¹⁾ Excludes volume throughput for Ohio Gathering and Double E. For additional information, see the Northeast and Permian sections herein under the caption "Segment Overview for the Three and Six Months Ended June 30, 2022 and 2021".

⁽²⁾ Includes a \$19.3 million incremental loss contingency recognized during the three and six months ended June 30, 2021 for the 2015 Blacktail Release (See Note 14 - Commitments and Contingencies for additional information).

Volumes - Gas.

Natural gas throughput volumes decreased 241 MMcf/d for the three months ended June 30, 2022 compared to the three months ended June 30, 2021, primarily reflecting:

- a volume throughput decrease of 221 MMcf/d for the Northeast segment.
- a volume throughput decrease of 14 MMcf/d for the Piceance segment.
- a volume throughput decrease of 2 MMcf/d for the Permian segment.
- a volume throughput decrease of 6 MMcf/d for the Rockies segment.
- a volume throughput increase of 2 MMcf/d for the Barnett segment.

Natural gas throughput volumes decreased 139 MMcf/d for the six months ended June 30, 2022 compared to the six months ended June 30, 2021, primarily reflecting:

- a volume throughput decrease of 113 MMcf/d for the Northeast segment.
- a volume throughput decrease of 22 MMcf/d for the Piceance segment.
- a volume throughput decrease of 2 MMcf/d for the Permian segment.
- a volume throughput decrease of 6 MMcf/d for the Rockies segment.
- a volume throughput increase of 4 MMcf/d for the Barnett segment.

Volumes - Liquids.

Crude oil and produced water throughput volumes at the Rockies segment decreased for the three and six months ended June 30, 2022, compared to the three and six months ended June 30, 2021, primarily as a result of natural production declines and weather related downtime, offset by 31 new well connections that came online subsequent to June 30, 2021.

For additional information on volumes, see the "Segment Overview for the Three and Six Months Ended June 30, 2022 and 2021" section herein.

Revenues. Total revenues decreased \$1.0 million during the three months ended June 30, 2022 compared to the prior year period, comprised of a \$11.9 million increase in natural gas, NGLs and condensate sales, offset by a \$12.6 million decrease in gathering services and related fees and a \$0.2 million decrease in Other Revenue.

<u>Gathering Services and Related Fees</u>. Gathering services and related fees decreased \$12.6 million compared to the three months ended June 30, 2021, primarily reflecting:

- a \$5.5 million decrease in the Piceance, primarily due to decreased volume throughput and the expiration of a customer's minimum volume commitment;
- a \$5.0 million decrease in the Northeast, primarily due to decreased volume throughput;
- a \$1.9 million decrease in the Rockies, primarily due to decreased volume throughput and the expiration of a customer's minimum volume commitment contract in the DJ basin.

<u>Natural Gas, NGLs and Condensate Sales</u>. Natural gas, NGLs and condensate revenues increased \$11.9 million compared to the three months ended June 30, 2021, primarily reflecting:

- a \$6.4 million increase in revenues in the Rockies, primarily as a result of increased commodity prices;
- a \$3.6 million increase in revenues from the Permian, primarily as a result of increased commodity prices.

Total revenues decreased \$4.2 million during the six months ended June 30, 2022 compared to the prior year period, comprised of an \$18.9 million decrease in gathering services and related fees; offset by a \$13.6 million increase in natural gas, NGLs and condensate sales, a \$1.2 million increase in Other Revenue.

<u>Gathering Services and Related Fees</u>. Gathering services and related fees decreased \$18.9 million compared to the six months ended June 30, 2021, primarily reflecting:

- a \$10.3 million decrease in the Piceance, primarily due to decreased volume throughput and the expiration of a customer's minimum volume commitment;
- a \$5.2 million decrease in the Northeast, primarily due to decreased volume throughput;

 a \$3.0 million decrease in the Rockies, primarily due to decreased volume throughput and the expiration of a customer's minimum volume commitment contract in the DJ basin.

<u>Natural Gas, NGLs and Condensate Sales</u>. Natural gas, NGLs and condensate revenues increased \$13.6 million compared to the six months ended June 30, 2021, primarily reflecting:

- a \$7.7 million increase in revenues in the Rockies, primarily as a result of increased commodity prices; and
- a \$4.0 million increase in the Permian, primarily due to increased commodity prices.

Costs and Expenses. Total costs and expenses increased \$81.6 million during the three months ended June 30, 2022 compared to the three months ended June 30, 2021.

Total costs and expenses increased \$87.3 million during the six months ended June 30, 2022 compared to the six months ended June 30, 2021.

<u>Cost of Natural Gas and NGLs</u>. Cost of natural gas and NGLs increased \$10.2 million for the three months ended June 30, 2022 compared to the three months ended June 30, 2021, primarily driven by an increase in commodity prices.

Cost of natural gas and NGLs increased \$12.0 million for the six months ended June 30, 2022 compared to the six months ended June 30, 2021, primarily driven by an increase in commodity prices.

<u>Operation and Maintenance</u>. Operation and maintenance expense increased \$4.8 million for the three months ended June 30, 2022 compared to the three months ended June 30, 2021 primarily as a result of the 2021 recognition of certain commercial settlement benefits and the employee retention tax credit ("ERC Tax Credit") that were not repeated in 2022.

Operation and maintenance expense increased \$5.2 million for the six months ended June 30, 2022 compared to the six months ended June 30, 2021, primarily as a result of the 2021 recognition of certain commercial settlement benefits and the ERC Tax Credit that were not repeated in 2022 and totaled \$5.6 million for the six months ended June 2021.

<u>General and Administrative</u>. General and administrative expense decreased \$18.9 million for the three months ended June 30, 2022 compared to the three months ended June 30, 2021, primarily related to the nonrecurrence of a \$19.3 million loss contingency recognized in 2021 (for the 2015 Blacktail Release) and partially offset by \$1.0 million of deal costs.

General and administrative expense decreased \$16.5 million for the six months ended June 30, 2022 compared to the six months ended June 30, 2021, primarily related to the nonrecurrence of a \$19.3 million loss contingency recognized in 2021 (for the 2015 Blacktail Release), partially offset by \$2.4 million of employee severance costs and \$1.1 million of deal costs incurred during the six months ended June 30, 2022.

<u>Depreciation and Amortization</u>. Depreciation and amortization expense increased \$1.7 million for the three months ended June 30, 2022 compared to the three months ended June 30, 2021 primarily as a result of increased depreciation expense on certain gas processing equipment.

Depreciation and amortization expense increased \$3.7 million for the six months ended June 30, 2022 compared to the six months ended June 30, 2021 primarily as a result of increased depreciation expense on certain gas processing equipment.

Asset Impairments. During the quarterly period ended June 30, 2022, the Partnership recognized an impairment of \$84.5 million related to the disposition of the Lane G&P System.

<u>Interest Expense</u>. Interest expense increased \$9.4 million for the three months ended June 30, 2022, compared to the three months ended June 30, 2021, primarily due to higher interest costs resulting from the issuance of the 2026 Secured Notes and borrowings on the Permian Transmission Term Loan, partially offset by principal reduction payments to our credit facilities (ABL Facility and Revolving Credit Facility).

Interest expense increased \$19.6 million for the six months ended June 30, 2022, compared to the six months ended June 30, 2021, primarily due to higher interest costs resulting from the issuance of the 2026 Secured Notes and borrowings on the Permian Transmission Term Loan, partially offset by principal reduction payments to our credit facilities (ABL Facility and Revolving Credit Facility).

Segment Overview for the Three and Six Months Ended June 30, 2022 and 2021

Northeast.

Volume throughput for the Northeast reportable segment follows.

			Northeas	st		
	Three Months End	led June 30,		Six Months Ende	ed June 30,	
rerage daily throughput (MMcf/d) rerage daily throughput (MMcf/d) (Ohio	2022	2021	Percentage Change	2022	2021	Percentage Change
Average daily throughput (MMcf/d)	632	853	(26)%	687	800	(14)%
Average daily throughput (MMcf/d) (Ohio Gathering)	562	514	9%	580	536	8%

Volume throughput for the Northeast, excluding Ohio Gathering, decreased 26% and 14% compared to the three and six months ended June 30, 2021, respectively, primarily due to natural production declines, maintenance related downtime, and frac-protect activities, partially offset by 18 well connections that came online subsequent to June 30, 2021.

Volume throughput for the Ohio Gathering system increased 9% and 8% compared to the three and six months ended June 30, 2021, respectively, primarily as a result of 16 new well connections that came online subsequent to June 30, 2021, partially offset by natural production declines and frac-protect activities.

Financial data for our Northeast reportable segment follows.

					No	rtheast				
	7	Three Months	Ended Jun	ie 30,			Six Months E	Inded Ju	ne 30,	
		2022	20	21	Percentage Change		2022		2021	Percentage Change
		(In tho	ousands)				(In tho	usands)		
Revenues:										
Gathering services and related fees	\$	12,940	\$	17,961	(28)%	\$	27,576	\$	32,733	(16)%
Total revenues		12,940		17,961	(28)%		27,576		32,733	(16)%
Costs and expenses:									,	
Operation and maintenance		1,588		1,316	21%		3,235		2,596	25%
General and administrative		237		104	128%		420		254	65%
Depreciation and amortization		4,338		4,229	3%		8,638		8,459	2%
(Gain) loss on asset sales, net		_		8	*		(10)		(54)	*
Long-lived asset impairment		_		(8)	*		_		130	*
Total costs and expenses		6,163		5,649	9%		12,283		11,385	8%
Add:										
Depreciation and amortization		4,338		4,229			8,638		8,459	
Adjustments related to capital										
reimbursement activity		(20)		(21)			(40)		(42)	
(Gain) loss on asset sales, net		_		8			(10)		(54)	
Long-lived asset impairment		_		(8)			_		130	
Proportional adjusted EBITDA for Ohio Gathering		7,473		6,841			14,749		13,713	
Other		_		_			6		_	
Segment adjusted EBITDA	\$	18,568	\$	23,361	(21)%	\$	38,636	\$	43,554	(11)%

^{*} Not considered meaningful

Three and six months ended June 30, 2022. Segment adjusted EBITDA decreased \$4.8 million and \$4.9 million, respectively, compared to the three and six months ended June 30, 2021 primarily as a result of a \$5.0 million and \$5.2 million, respectively, decrease in revenues from gathering services and related fees for the three and six months ended June 30, 2022, partially offset by a \$0.6 million and \$1.0 million, respectively, increase in proportional adjusted EBITDA from Ohio Gathering.

Rockies.

Volume throughput for our Rockies reportable segment follows.

			Rock	ies		
	Three Months	Ended June 30,		Six Months H	Ended June 30,	
	2022	2021	Percentage Change	2022	2021	Percentage Change
Aggregate average daily throughput - natural gas (MMcf/d)	29	35	(17%)	29	35	(17%)
Aggregate average daily throughput - liquids (Mbbl/d)	54	63	(14%)	60	64	(6%)

<u>Natural gas</u>. Natural gas volume throughput decreased 17% compared to the three and six months ended June 30, 2021, primarily reflecting natural production declines and weather related downtime, partially offset by five new well connections that came online subsequent to June 30, 2021.

<u>Liquids</u>. Liquids volume throughput decreased 14% and 6% compared to the three and six months ended June 30, 2021, primarily associated with natural product declines and operational and weather related downtime, partially offset by 31 new well connections that came online subsequent to June 30, 2021.

Financial data for our Rockies reportable segment follows.

				Re	ockies				
	Three Months	Ended	June 30,			Six Months E	nded Ju	ıne 30,	
	 2022		2021	Percentage Change		2022		2021	Percentage Change
	(In tho	usand	s)			(In tho	usands)		
Revenues:									
Gathering services and related fees	\$ 16,510	\$	18,407	(10)%	\$	34,299	\$	37,303	(8%)
Natural gas, NGLs and condensate sales	14,927		8,506	75%		28,586		20,843	37%
Other revenues	4,728		6,098	(22)%		9,885		11,309	(13)%
Total revenues	 36,165		33,011	10%		72,770		69,455	5%
Costs and expenses:									
Cost of natural gas and NGLs	14,207		8,762	62%		27,629		21,103	31%
Operation and maintenance	7,214		7,365	(2%)		13,426		14,201	(5%)
General and administrative	515		1,682	(69%)		1,199		2,355	(49%)
Depreciation and amortization	7,471		7,459	0%		14,919		14,933	0%
(Gain) loss on asset sales, net	_		(5)	*		14		(22)	*
Long-lived asset impairment	98		41	*		111		136	*
Total costs and expenses	 29,505		25,304	17%		57,298		52,706	9%
Add:									
Depreciation and amortization	7,471		7,459			14,919		14,933	
Adjustments related to capital	(462)		(450)			(0.40)		(025)	
reimbursement activity	(462)		(470)			(940)		(935)	
(Gain) loss on asset sales, net	_		(5)			14		(22)	
Long-lived asset impairment	98		41			111		136	
Other	132 -		_			153 –	-	23	
Segment adjusted EBITDA	\$ 13,899	\$	14,732	(6)%	\$	29,729	\$	30,884	(4%)

^{*} Not considered meaningful

Three and six months ended June 30, 2022. Segment adjusted EBITDA decreased \$0.8 million and \$1.2 million, respectively, compared to the three and six months ended June 30, 2021 primarily due to lower volume throughput on our systems as previously discussed, as well as the expiration of a customer's minimum volume commitment contract in the DJ basin, partially offset by \$0.2 million and \$0.8 million, respectively, of lower operating and maintenance expenses.

Table of Contents

Permian.

Volume throughput for our Permian reportable segment follows.

			Pern	nian		
	Three Months End	ded June 30,		Six Months End	ed June 30,	
	2022	2021	Percentage Change	2022	2021	Percentage Change
Average daily throughput (MMcf/d)	27	29	(7%)	27	29	(7%)
Average daily throughput (MMcf/d) (Double E)	314	_	n/a	251	_	n/a

On June 30, 2022, we completed the sale of our Lane G&P System. The volumes above include daily volume throughput data through the date of sale. Subsequent to the date of sale, the Permian reportable segment will consist solely of our equity method investment in Double E.

Volume throughput, excluding Double E, decreased 7% compared to the three and six months ended June 30, 2021, primarily as a result of natural production declines, partially offset by four new connections that came online subsequent to June 30, 2021.

Double E commenced operations during November 2021. Volume throughput for Double E for the three and six months ended June 30, 2021 averaged 314 MMcf per day and 251 MMcf per day, respectively.

The following table presents the MVC quantities that Double E's shippers have contracted to with firm transportation service agreements and related negotiated rate agreements.

(Amounts in MMBTU/day)	Weighted average MVC quantities for the year ended December 31,
2022	622,603
2023	839,247
2024	990,205
2025	1,000,000
2026	1,000,178
2027	1,000,000
2028	1,002,562
2029	1,000,000
2030	1,000,000
2031	832,877

Financial data for our Permian reportable segment follows.

	Permian									
	Th	ree Months	Ended Ju	me 30,			Six Months E	nded Ju	ıne 30,	
	:	2022	2	2021	Percentage Change		2022		2021	Percentage Change
		(In tho	usands)				(In tho	usands)		
Revenues:										
Gathering services and related fees	\$	1,822	\$	2,262	(19%)	\$	3,669	\$	4,461	(18%)
Natural gas, NGLs and condensate sales		10,514		6,875	53%		17,381		13,393	30%
Other revenues		1,298		1,049	24%		2,317		1,809	28%
Total revenues		13,634		10,186	34%		23,367		19,663	19%
Costs and expenses:										
Cost of natural gas and NGLs		10,721		7,167	50%		17,813		14,183	26%
Operation and maintenance		1,779		1,527	17%		3,083		2,519	22%
General and administrative		187		166	13%		550		384	43%
Depreciation and amortization		1,236		1,464	(16)%		2,733		2,933	(7)%
Gain on asset sales, net		(13)		_	*		(13)		_	*
Long-lived asset impairment		84,516		_	*		84,516		_	*
Total costs and expenses	,	98,426		10,324	853%		108,682		20,019	443%
Add:										
Depreciation and amortization		1,236		1,464			2,733		2,933	
Adjustments related to capital reimbursement activity		(63)					(63)			
, and the second		` ′		-			` ´		-	
Gain on asset sales, net		(13)		_			(13)		_	
Long-lived asset impairment		84,516					84,516		_	
Proportional adjusted EBITDA for Double E		3,933					7,108			
Other				15					15	
Segment adjusted EBITDA	\$	4,817	\$	1,341	259%	\$	8,966	\$	2,592	246%

^{*}Not considered meaningful

<u>Three and six months ended June 30, 2022</u>. Segment adjusted EBITDA increased \$3.5 million and \$6.4 million compared to the three and six months ended June 30, 2021, primarily reflecting an increase in proportional adjusted EBITDA from Double E.

Table of Contents

Piceance.

Volume throughput for our Piceance reportable segment follows.

			Pice	ance		
	Three Months En	ded June 30,		Six Months Endo	ed June 30,	
	2022	2021	Percentage Change	2022	2021	Percentage Change
Aggregate average daily throughput (MMcf/d)	312	326	(4%)	312	334	(7%)

Volume throughput decreased 4% and 7%, respectively, compared to the three and six months ended June 30, 2021, as a result of natural production declines, partially offset by 9 new well connections that came online in October 2021.

Financial data for our Piceance reportable segment follows.

				Pie	ceance				
	Three Months I	Ended J	June 30,			Six Months E	nded J	une 30,	
	2022		2021	Percentage Change		2022		2021	Percentage Change
	(In thou	ısands)			(In thousands)	
Revenues:									
Gathering services and related fees	\$ 19,981	\$	25,527	(22%)	\$	40,052	\$	50,311	(20%)
Natural gas, NGLs and condensate sales	2,067		1,025	102%		3,962		2,878	38%
Other revenues	1,276		1,233	3%		2,551		2,409	6%
Total revenues	23,324		27,785	(16%)		46,565		55,598	(16%)
Costs and expenses:									
Cost of natural gas and NGLs	1,266		697	82%		2,374		1,816	31%
Operation and maintenance	5,785		5,367	8%		11,058		10,309	7%
General and administrative	326		345	(6)%		654		643	2%
Depreciation and amortization	12,846		10,757	19%		25,626		21,631	18%
(Gain) loss on asset sales, net	(300)		4	*		(300)		(53)	*
Long-lived asset impairment	_		_	*		_		970	*
Total costs and expenses	 19,923		17,170	16%		39,412		35,316	12%
Add:									
Depreciation and amortization	12,846		10,757			25,626		21,631	
Adjustments related to capital reimbursement activity	(702)		(1,403)			(1,601)		(1,831)	
(Gain) loss on asset sales, net	(300)		4			(300)		(53)	
Long-lived asset impairment	_		_			_		970	
Other	105		351			240		359	
Segment adjusted EBITDA	\$ 15,350	\$	20,324	(24%)	\$	31,118	\$	41,358	(25%)

^{*}Not considered meaningful

Three and six months ended June 30, 2022. Segment adjusted EBITDA decreased \$5.0 million and \$10.2 million, respectively, compared to the three and six months ended June 30, 2021, primarily reflecting a decrease in volume throughput as a result of natural production declines as discussed above as well as \$3.6 million and \$7.0 million in reduced revenue for the three and six month periods, respectively, related to the expiration of a customer's minimum volume commitment contract in 2021.

Barnett.

Volume throughput for our Barnett reportable segment follows.

			Bar	nett		
	Three Months En	ded June 30,		Six Months End	ed June 30,	
	2022	2021	Percentage Change	2022	2021	Percentage Change
Average daily throughput (MMcf/d)	200	198	1%	199	195	2%

Volume throughput increased 1% and 2% compared to the three and six months ended June 30, 2021, primarily as a result of eleven wells that came online subsequent to June 30, 2021, partially offset by natural production declines.

Financial data for our Barnett reportable segment follows.

					В	arnett				
	T	hree Months	Ende	d June 30,			Six Months I	Ended	June 30,	
		2022		2021	Percentage Change		2022		2021	Percentage Change
		(In tho	usano	ls)			(In the	usanc	ls)	
Revenues:										
Gathering services and related fees	\$	10,378	\$	10,076	3%	\$	20,055	\$	19,772	1%
Natural gas, NGLs and condensate sales		786		10	7,760%		786		66	1,091%
Other revenues (1)		1,671		1,012	65%		3,734		2,072	80%
Total revenues		12,835		11,098	16%		24,575		21,910	12%
Costs and expenses:										
Operation and maintenance		5,220		1,852	182%		7,344		4,316	70%
General and administrative		272		260	5%		514		495	4%
Depreciation and amortization		3,790		3,798	_		7,582		7,596	_
Gain on asset sales, net		_		(11)	*		_		(11)	*
Long-lived asset impairment		_		_	*		_		289	*
Total costs and expenses		9,282		5,899	57%		15,440		12,685	22%
Add:										
Depreciation and amortization		4,025		4,032			8,051		8,064	
Adjustments related to capital reimbursement activity		(331)		(331)			(658)		(662)	
Gain on asset sales, net		_		(11)			_		(11)	
Long-lived asset impairment		_		_			_		289	
Other		_		_			5		_	
Segment adjusted EBITDA	\$	7,247	\$	8,889	(18)%	\$	16,533	\$	16,905	(2)%

^{*}Not considered meaningful

Three and six months ended June 30, 2022. Segment adjusted EBITDA decreased \$1.6 million and \$0.4 million, respectively, compared to the three and six months ended June 30, 2021, primarily as a result of higher general operating expenses that resulted from the nonrecurrence of commercial settlements that benefited the segment's financial results in 2021, partially offset by a \$0.7 million and \$1.7 million increase in other revenues for the three and six months ended June 30, 2022.

⁽¹⁾ Includes the amortization expense associated with our favorable gas gathering contracts as reported in Other revenues.

Corporate and Other Overview for the Three and Six Months Ended June 30, 2022 and 2021

Corporate and Other represents those results that are not specifically attributable to a reportable segment or that have not been allocated to our reportable segments, including certain general and administrative expense items, natural gas and crude oil marketing services, transaction costs and interest expense.

				Corporat	te and C	Other			
	 Three Months	June 30,		Six Months Ended June 30,					
	2022		2021	Percentage Change		2022		2021	Percentage Change
	 (In tho	usands	s)			(In the	usand	ls)	
Revenues:									
Total revenues	\$ 165	\$	_	*	\$	336	\$	_	*
Costs and expenses:									
Operation and maintenance	\$ 691	\$	80	*	\$	1,193	\$	159	*
General and administrative ⁽¹⁾	8,936		26,803	(67)%		20,096		35,808	(44)%
Transaction costs	(13)		450	*		233		217	*
Interest expense	24,887		15,502	61%		49,050		29,455	67%

^{*} Not considered meaningful

General and Administrative. General and administrative expense decreased by \$17.9 million, compared to the three months ended June 30, 2021, primarily related to the nonrecurrence of a \$19.3 million loss contingency recognized in 2021 (for the 2015 Blacktail Release) and partially offset by \$1.0 million of deal costs.

General and administrative expense decreased by \$15.7 million, compared to the six months ended June 30, 2021, primarily related to primarily related to the nonrecurrence of a \$19.3 million loss contingency recognized in 2021 (for the 2015 Blacktail Release), partially offset by \$2.4 million of employee severance costs and \$1.1 million of deal costs incurred during the six months ended June 30, 2022.

<u>Interest Expense</u>. The increase in interest expense for the three and six months ended June 30, 2022, compared to three and six months ended June 30, 2021, was primarily due to higher interest costs resulting from the issuance of the 2026 Secured Notes and borrowings on the Permian Transmission Term Loan, partially offset by principal reduction payments to our credit facilities (ABL Facility and Revolving Credit Facility).

Liquidity and Capital Resources

We rely primarily on internally generated cash flows as well as external financing sources, including our ABL Facility, and the issuance of debt, equity and preferred equity securities, and proceeds from potential asset divestitures to fund our capital expenditures. We believe that our ABL Facility and Permian Transmission Credit Facility, together with internally generated cash flows and access to debt or equity capital markets, will be adequate to finance our operations for the next twelve months without adversely impacting our liquidity.

We may enter into off-balance sheet arrangements and transactions that can give rise to material off-balance sheet obligations. As of June 30, 2022, our material off-balance sheet arrangements and transactions include (i) letters of credit outstanding against our ABL Facility aggregating to \$5.9 million, (ii) letters of credit outstanding against our Permian Transmission Credit Facility aggregating to \$7.5 million. There are no other transactions, arrangements or other relationships with unconsolidated entities or other persons that are reasonably likely to materially affect our liquidity or availability of our capital resources.

⁽¹⁾ Inclusive of a \$19.3 million incremental loss contingency accrual during the three months ended June 30, 2021 related to the 2015 Blacktail Release (See Note 14 - Commitments and Contingencies for additional information).

Indebtedness Compliance as of June 30, 2022. As of June 30, 2022, we were in compliance with all covenants contained in the Senior Notes, ABL Facility and the Permian Transmission Credit Facility. The ABL Facility requires that Summit Holdings not permit (i) the First Lien Net Leverage Ratio (as defined in the ABL Agreement) as of the last day of any fiscal quarter to be greater than 2.50:1.00, or (ii) the Interest Coverage Ratio (as defined in the ABL Agreement) as of the last day of any fiscal quarter to be less than 2.00:1.00. As of June 30, 2022, the First Lien Net Leverage Ratio and Interest Coverage Ratio was 0.64:1.00 and 2.73:1.00, respectively.

Credit Agreements and Financing Activities

ABL Facility. As of June 30, 2022, we had a \$400.0 million revolving ABL Facility with a maturity date of May 1, 2026. As of June 30, 2022, the outstanding balance of the ABL Facility was \$151.0 million and the available borrowing capacity totaled \$243.1 million after giving effect to the issuance thereunder of \$5.9 million of outstanding but undrawn irrevocable standby letters of credit.

We may in the future use a combination of cash, secured or unsecured borrowings and issuances of our common units or other securities and the proceeds from asset sales to retire or refinance our outstanding debt or Series A Preferred Units through privately negotiated transactions, open market repurchases, redemptions, exchange offers, tender offers or otherwise, but we are under no obligation to do so.

LIBOR Transition

LIBOR is the basic rate of interest widely used as a reference for setting the interest rates on loans globally. In 2017, the United Kingdom's Financial Conduct Authority, which regulates LIBOR, announced that it intended to phase out LIBOR by the end of 2021. The U.S. Federal Reserve, in conjunction with the Alternative Reference Rates Committee, a steering committee comprised of large U.S. financial institutions, is considering replacing U.S. dollar LIBOR with a new index, the Secured Overnight Financing Rate ("SOFR"), calculated using short-term repurchase agreements backed by Treasury securities. After December 31, 2021, the United Kingdom ceased publications of certain LIBOR tenors and expects to cease publications of all USD LIBOR tenors after June 30, 2023. We are evaluating the potential impact of the eventual replacement of the LIBOR benchmark interest rate, however, we are not able to predict when LIBOR will cease to be available, whether SOFR will become a widely accepted benchmark in place of LIBOR or what the impact of such a possible transition to SOFR may be on our business, financial condition and results of operations.

The ABL Facility includes an ability of the administrative agent to transition loans based on LIBOR to loans based on term SOFR or daily simple SOFR. The adjustment rate upon a transition to term SOFR is 11.448 bps for a one-month tenor, 26.161 bps for a three-month tenor, 42.826 for a six-month tenor, and 71.513 for a twelve-month tenor. This transition has not yet occurred with the ABL Facility.

Cash Flows

The components of the net change in cash and cash equivalents were as follows:

	Six Months Ended June 30,			
	 2022		2021	
	 (In thousands)			
Net cash provided by operating activities	\$ 60,159	\$	86,217	
Net cash provided by (used in) investing activities	56,882		(46,905)	
Net cash used in financing activities	(120,913)		(47,331)	
Net change in cash, cash equivalents and restricted cash	\$ (3,872)	\$	(8,019)	

Operating activities.

Cash flows provided by operating activities for the six months ended June 30, 2022 primarily reflected:

a net loss of \$91.8 million plus positive adjustments of \$152.9 million for non-cash operating items.

Cash flows provided by operating activities for the six months ended June 30, 2021 primarily reflected:

- net loss of \$10.8 million plus adjustments of \$90.5 million for non-cash items; and
- \$6.4 million increase in working capital accounts.

Investing activities.

Cash flows provided by investing activities during the six months ended June 30, 2022 primarily reflected:

\$75.5 million of cash proceeds from the disposition of the Lane G&P System, net of cash sold in the transaction;

- \$4.6 million of cash proceeds from the sale of compressors; offset by
- \$14.8 million cash outflow for capital expenditures; and
- \$8.4 million of cash investments in the Double E Project.

Cash flows used in investing activities during the six months ended June 30, 2021 primarily reflected:

- \$48.9 million for cash investments in the Double E Project;
- \$6.0 million cash outflow for capital expenditures; offset by
- \$8.0 million of cash proceeds from the sale of compressors.

Financing activities.

Cash flows used in financing activities during the six months ended June 30, 2022 primarily reflected:

- \$139.0 million of cash outflow for repayments on the ABL Facility, offset by
- \$23.0 million from borrowings on ABL Facility.

Cash flows used in financing activities during the six months ended June 30, 2021 primarily reflected:

- \$95.0 million of cash outflow for repayments on the Revolving Credit Facility;
- \$5.2 million of cash payments related to debt issuance costs; offset by
- \$53.5 million from borrowings under the Permian Transmission Credit Facility.

Capital Requirements

Overall.

Our business is capital intensive, requiring significant investment for the maintenance of existing gathering systems and the acquisition or construction and development of new gathering systems and other midstream assets and facilities. Our Partnership Agreement requires that we categorize our capital expenditures as either:

- maintenance capital expenditures, which are cash expenditures (including expenditures for the addition or improvement to, or the replacement
 of, our capital assets or for the acquisition of existing, or the construction or development of, new capital assets) made to maintain our longterm operating income or operating capacity; or
- expansion capital expenditures, which are cash expenditures incurred for acquisitions or capital improvements that we expect will increase our operating income or operating capacity over the long term.

For the six months ended June 30, 2022, cash paid for capital expenditures totaled \$14.8 million which included \$4.8 million of maintenance capital expenditures. For the six months ended June 30, 2022, there were no contributions to Ohio Gathering and we contributed \$8.4 million to Double E.

We rely primarily on internally generated cash flows as well as external financing sources, including commercial bank borrowings and the issuance of debt, equity and preferred equity securities, and proceeds from potential asset divestitures to fund our capital expenditures. We believe that our internally generated cash flows, our ABL Facility and the Permian Transmission Credit Facility, and access to debt or equity capital markets, will be adequate to finance our operations for the next twelve months without adversely impacting our liquidity.

There are a number of risks and uncertainties that could cause our current expectations to change, including, but not limited to, (i) the ability to reach agreements with third parties; (ii) prevailing conditions and outlook in the natural gas, crude oil and NGLs and markets, and (iii) our ability to obtain financing from commercial banks, the capital markets, or other financing sources.

Excess Cash Flow Offers to Purchase.

Starting in the first quarter of 2023 with respect to the fiscal year ended 2022, and continuing annually through fiscal year 2025, the Partnership is required under the terms of the 2026 Secured Notes Indenture to, if it has Excess Cash Flow (as defined in the 2026 Secured Notes Indenture), and subject to its ability to make such an offer under the ABL Credit Facility, offer to purchase an amount of the 2026 Secured Notes, at 100% of the principal amount plus accrued and unpaid interest, equal to 100% of the Excess Cash Flow generated in the prior year.

Generally, if the Partnership does not offer to purchase designated annual amounts of its 2026 Secured Notes or reduce its first lien capacity under the 2026 Secured Notes Indenture per annum from 2023 through 2025, the interest rate on the 2026 Secured Notes are subject to certain rate escalations. Per the terms of the 2026 Secured Notes Indenture, the designated amounts are; (i)

\$50.0 million in aggregate by April 1, 2023, otherwise the interest rate shall automatically increase by 50 basis points per annum; (ii) \$100.0 million in aggregate by April 1, 2024, otherwise the interest rate shall automatically increase by 100 basis points per annum (minus any amount previously increased); and (iii) \$200.0 million in aggregate by April 1, 2025, otherwise the interest rate shall automatically increase by 200 basis points per annum (minus any amount previously increased).

To the extent the Partnership makes an offer to purchase, and the offer is not fully accepted by the holders of the 2026 Secured Notes, the Partnership may use any remaining amount not accepted for any purpose not prohibited by the 2026 Secured Notes Indenture or the ABL Facility.

Credit and Counterparty Concentration Risks

We examine the creditworthiness of counterparties to whom we extend credit and manage our exposure to credit risk through credit analysis, credit approval, credit limits and monitoring procedures, and for certain transactions, we may request letters of credit, prepayments or guarantees.

Certain of our customers may be temporarily unable to meet their current obligations. While this may cause disruption to cash flows, we believe that we are properly positioned to deal with the potential disruption because the vast majority of our gathering assets are strategically positioned at the beginning of the midstream value chain. The majority of our infrastructure is connected directly to our customers' wellheads and pad sites, which means our gathering systems are typically the first third-party infrastructure through which our customers' commodities flow and, in many cases, the only way for our customers to get their production to market.

We have exposure due to nonperformance under our MVC contracts whereby a potential customer, may not have the wherewithal to make its MVC shortfall payments when they become due. We typically receive payment for all prior-year MVC shortfall billings in the quarter immediately following billing. Therefore, our exposure to risk of nonperformance is limited to and accumulates during the current year-to-date contracted measurement period.

Off-Balance Sheet Arrangements

During the three and six months ended June 30, 2022, there were no material changes to the off-balance sheet obligations disclosed in our 2021 Annual Report other than the following. On June 30, 2022, in connection with the disposition of the Lane G&P System, a subsidiary of Matador Resources assumed our commitment related to the take-or-pay firm capacity on the Double E Pipeline. The assumption of the take-or-pay contract relieved the Partnership of \$29.2 million of remaining firm transportation obligations to Double E as of June 30, 2022.

Summarized Financial Information

The supplemental summarized financial information below reflects SMLP's separate accounts, the combined accounts of Summit Holdings and Finance Corp. (together, the "Co-Issuers") and its guarantor subsidiaries (the "Guarantor Subsidiaries" and together with the Co-Issuers, the "Obligor Group") for the dates and periods indicated. The financial information of the Obligor Group is presented on a combined basis and intercompany balances and transactions between the Co-Issuers and Guarantor Subsidiaries have been eliminated. There were no reportable transactions between the Co-Issuers and Obligor Group and the subsidiaries that were not issuers or guarantors of the Senior Notes.

Payments to holders of the Senior Notes are affected by the composition of and relationships among the Co-Issuers, the Guarantor Subsidiaries and Permian Holdco and Summit Permian Transmission, both of which are unrestricted subsidiaries of SMLP and are not issuers or guarantors of the Senior Notes. The assets of our unrestricted subsidiaries are not available to satisfy the demands of the holders of the Senior Notes. In addition, our unrestricted subsidiaries are subject to certain contractual restrictions related to the payment of dividends, and other rights in favor of their non-affiliated stakeholders, that limit their ability to satisfy the demands of the holders of the Senior Notes.

A list of each of SMLP's subsidiaries that is a guarantor, issuer or co-issuer of our registered securities subject to the reporting requirements in Release 33-10762 is filed as Exhibit 22.1 to this report.

On June 30, 2022, the Partnership completed the disposition of all the equity interests in Summit Permian and Summit Permian Finance for a cash sale price of \$75.0 million. In connection with the sale, the status of Summit Permian and Summit Permian Finance as guarantor subsidiaries was modified prior to the disposition. The summarized financial information below presents the activities and balances of Summit Permian and Summit Permian Finance as guarantor subsidiaries for all summarized income statement periods and balance sheet dates presented in which they were owned by the Partnership. Summit Permian and Summit Permian Finance were not included in the Partnership's balance sheet as of June 30, 2022 and their assets and liabilities are not included in the June 30, 2022 summarized balance sheet below.

Summarized Balance Sheet Information. Summarized balance sheet information as of June 30, 2022 and December 31, 2021 follows.

	June 30, 2022		
	SMLP	(Obligor Group
	(In thousands)		
\$	3,577	\$	66,579
	5,424		1,930,417
\$	11,388	\$	57,259
	2,470		1,158,937
	December 31, 2021		
	SMLP	(Obligor Group
	(In tho	usand	s)
\$	2,495	\$	70,483
	4,776		2,149,300
\$	12,463	\$	58,658
	1,771		1,274,803

Summarized Statements of Operations Information. For the purposes of the following summarized statements of operations, we allocate a portion of general and administrative expenses recognized at the SMLP parent to the Obligor Group to reflect what those entities' results would have been had they operated on a stand-alone basis. Summarized statements of operations for the six months ended June 30, 2022 and for the year ended December 31, 2021 follow.

Six Months Ended June 30, 2022

(37,291) \$

Obligor Group

23,047

SMLP

	SIVILI		nigor Group
	 (In tho	usands)	
Total revenues	\$ _	\$	195,189
Total costs and expenses	1,020		255,077
Loss before income taxes and income from equity method investees	(1,020)		(106,113)
Income from equity method investees	_		6,280
Net loss	\$ (1,393)	\$	(99,833)
	Year Ended December 31, 2021		
	Year Ended De	cember 3	31, 2021
	 Year Ended De SMLP		31, 2021 oligor Group
	 SMLP		
Total revenues	\$ SMLP	Ob	
Total revenues Total costs and expenses	\$ SMLP (In tho	Ob usands)	oligor Group
	\$ SMLP (In tho	Ob usands)	400,619

Critical Accounting Estimates

Net income (loss)

We prepare our financial statements in accordance with GAAP. These principles are established by the FASB. We employ methods, estimates and assumptions based on currently available information when recording transactions resulting from business operations. There have been no significant changes to our critical accounting estimates from those disclosed on Form 10-K for the fiscal year ended December 31, 2021.

Forward-Looking Statements

Investors are cautioned that certain statements contained in this report as well as in periodic press releases and certain oral statements made by our officers and employees during our presentations are "forward-looking" statements. Forward-looking statements include, without limitation, any statement that may project, indicate or imply future results, events, performance or achievements and may contain the words "expect," "intend," "plan," "anticipate," "estimate," "believe," "will be," "will continue," "will likely result," and similar expressions, or future conditional verbs such as "may," "will," "should," "would," and "could." In addition, any statement concerning future financial performance (including future revenues, earnings or growth rates), ongoing business strategies or prospects, and possible actions taken by us or our subsidiaries are also forward-looking statements. These forward-looking statements involve various risks and uncertainties, including, but not limited to, those described in Part II. Item 1A. Risk Factors included in this report.

Forward-looking statements are based on current expectations and projections about future events and are inherently subject to a variety of risks and uncertainties, many of which are beyond the control of our management team. All forward-looking statements in this report and subsequent written and oral forward-looking statements attributable to us, or to persons acting on our behalf, are expressly qualified in their entirety by the cautionary statements in this paragraph. These risks and uncertainties include, among others:

- our decision whether to pay, or our ability to grow, our cash distributions;
- fluctuations in natural gas, NGLs and crude oil prices, including as a result of political or economic measures taken by various countries or OPEC or as a result of the Russia-Ukraine conflict;
- the extent and success of our customers' drilling and completion efforts, as well as the quantity of natural gas, crude oil and produced water volumes produced within proximity of our assets;
- the current and potential future impact of the COVID-19 pandemic on our business, results of operations, financial position or cash flows;
- · failure or delays by our customers in achieving expected production in their natural gas, crude oil and produced water projects;
- competitive conditions in our industry and their impact on our ability to connect hydrocarbon supplies to our gathering and processing assets or systems;
- actions or inactions taken or nonperformance by third parties, including suppliers, contractors, operators, processors, transporters and
 customers, including the inability or failure of our shipper customers to meet their financial obligations under our gathering agreements and
 our ability to enforce the terms and conditions of certain of our gathering agreements in the event of a bankruptcy of one or more of our
 customers;
- our ability to divest of certain of our assets to third parties on attractive terms, which is subject to a number of factors, including prevailing conditions and outlook in the natural gas, NGL and crude oil industries and markets;
- the ability to attract and retain key management personnel;
- commercial bank and capital market conditions and the potential impact of changes or disruptions in the credit and/or capital markets;
- changes in the availability and cost of capital and the results of our financing efforts, including availability of funds in the credit and/or capital markets;
- our ability to refinance near-term maturities on favorable terms or at all and the related impact on our ability to continue as a going concern;
- restrictions placed on us by the agreements governing our debt and preferred equity instruments;
- the availability, terms and cost of downstream transportation and processing services;
- natural disasters, accidents, weather-related delays, casualty losses and other matters beyond our control;
- · operational risks and hazards inherent in the gathering, compression, treating and/or processing of natural gas, crude oil and produced water;
- our ability to comply with the terms of the agreements comprising the Global Settlement;
- weather conditions and terrain in certain areas in which we operate;
- any other issues that can result in deficiencies in the design, installation or operation of our gathering, compression, treating and processing facilities;

Table of Contents

- timely receipt of necessary government approvals and permits, our ability to control the costs of construction, including costs of materials, labor and rights-of-way and other factors that may impact our ability to complete projects within budget and on schedule;
- our ability to finance our obligations related to capital expenditures, including through opportunistic asset divestitures or joint ventures and the impact any such divestitures or joint ventures could have on our results;
- the effects of existing and future laws and governmental regulations, including environmental, safety and climate change requirements and federal, state and local restrictions or requirements applicable to oil and/or gas drilling, production or transportation;
- changes in tax status;
- the effects of litigation;
- changes in general economic conditions; and
- certain factors discussed elsewhere in this report.

Developments in any of these areas could cause actual results to differ materially from those anticipated or projected or cause a significant reduction in the market price of our common units, preferred units and senior notes.

The foregoing list of risks and uncertainties may not contain all of the risks and uncertainties that could affect us. In addition, in light of these risks and uncertainties, the matters referred to in the forward-looking statements contained in this document may not in fact occur. Accordingly, undue reliance should not be placed on these statements. We undertake no obligation to publicly update or revise any forward-looking statements as a result of new information, future events or otherwise, except as otherwise required by law.

Information About Us

Investors should note that we make available, free of charge on our website at www.summitmidstream.com, our annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K, and any amendments to those reports as soon as reasonably practicable after we electronically file such material with, or furnish it to, the SEC. We also post announcements, updates, events, investor information and presentations on our website in addition to copies of all recent news releases. We may use the Investors section of our website to communicate with investors. It is possible that the financial and other information posted there could be deemed to be material information. Documents and information on our website are not incorporated by reference herein.

The SEC maintains a website at www.sec.gov that contains reports, proxy and information statements, and other information regarding issuers, including us, that file electronically with the SEC.

Item 3. Quantitative and Qualitative Disclosures About Market Risk.

Interest Rate Risk

Our interest rate risk exposure, which is largely related to our indebtedness, has not changed materially since December 31, 2021. As of June 30, 2022, we had approximately \$959.5 million principal of fixed-rate debt, \$151.0 million outstanding under our variable rate ABL Facility and \$157.8 million outstanding under the variable rate Permian Transmission Term Loan (see Note 8 - Debt). As of June 30, 2022, we had \$142.0 million of interest rate exposure hedged to offset the impact of changes in interest rates on our Permian Transmission Term Loan. While existing fixed-rate debt mitigates the downside impact of fluctuations in interest rates, future issuances of long-term debt could be impacted by increases in interest rates, which could result in higher overall interest costs. In addition, the borrowings under our ABL Facility, which have a variable interest rate, also expose us to the risk of increasing interest rates. For additional information, see the "Interest Rate Risk" section included in Item 7A. Quantitative and Qualitative Disclosures About Market Risk of the 2021 Annual Report and updates to our risk factors included herein.

Commodity Price Risk

We generate a majority of our revenues pursuant to primarily long-term and fee-based gathering agreements, many of which include MVCs and areas of mutual interest. Our direct commodity price exposure relates to (i) the sale of physical natural gas and/or NGLs purchased under percentage-of-proceeds and other processing arrangements with certain of our customers in the Rockies, and Piceance segments, (ii) the sale of natural gas we retain from certain Barnett segment customers and (iii) the sale of condensate we retain from certain gathering services in the Piceance segment. Our gathering agreements with certain Barnett customers permit us to retain a certain quantity of natural gas that we sell to offset the power costs we incur to operate our electric-drive compression assets. We manage our direct exposure to natural gas and power prices through the use of forward power purchase contracts with wholesale power providers that require us to purchase a fixed quantity of power at a fixed heat rate based on prevailing natural gas prices on the Henry Hub Index. We sell retainage natural gas at prices that are based on the Atmos Zone 3 Index. By basing the power prices on a system and basin-relevant market, we are able to closely associate the relationship between the compression electricity expense and natural gas retainage sales. We do not enter into risk management contracts for speculative purposes. Our current commodity price risk exposure has not changed materially since December 31, 2021. For additional information, see the "Commodity Price Risk" section included in Item 7A. Quantitative and Qualitative Disclosures About Market Risk of the 2021 Annual Report.

Item 4. Controls and Procedures.

Under the direction of our General Partner's Chief Executive Officer and Chief Financial Officer, we evaluated our disclosure controls and procedures and internal control over financial reporting and concluded that (i) our disclosure controls and procedures were effective as of June 30, 2022 and (ii) no change in internal control over financial reporting occurred during the quarter ended June 30, 2022, that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings.

Although we may, from time to time, be involved in litigation and claims arising out of our operations in the ordinary course of business, we are not currently a party to any significant legal or governmental proceedings, except as described below. In addition, we are not aware of any significant legal or governmental proceedings contemplated to be brought against us, under the various environmental protection statutes to which we are subject.

Moore Control Systems

Moore Control Systems, Inc. ("MCSI") initiated an arbitration in December 2020 related to the construction of Summit's Lane Gas Processing Plant in Eddy County, New Mexico (the "Lane Plant"). MCSI was the EPC contractor on the Lane Plant under a lump sum contract. During the course of construction, the parties executed change orders that increased the lump sum price. Summit paid the adjusted lump sum amount in full upon the plant's completion in June 2019.

A merits hearing is currently scheduled for March 6, 2023 in Houston, Texas. MCSI seeks alleged damages totaling approximately \$12 million on various theories, including breach of contract, quantum meruit and fraud, plus interest, fees and costs. Summit denies all liability and seeks to recover all costs and fees associated with defending the arbitration. We do not currently believe that the eventual outcome of this matter could have a material adverse effect on our business, financial condition, results of operations or cash flows. In conjunction with the disposition of Lane Gathering and Processing system on June 30, 2022, Summit Holdings provided the buyer with an indemnity for and hold harmless from these claims.

Fiberspar Corporation

On May 3, 2022, Fiberspar Corporation filed a petition in state court alleging \$5 million or more owed but not paid by Summit Midstream Partners, LLC for orders of pipeline product from Fiberspar. The petition asserts causes of action for breach of contract and suit on sworn account. A civil action on the same claims had been filed by Fiberspar on January 29, 2016, which was non-suited on September 26, 2016 pursuant to an agreement that included a standstill period through September 30, 2021. We have not been served with the petition, which is pending in the District Court of Harris County, Texas.

Item 1A. Risk Factors.

The risk factors contained in the Item 1A. Risk Factors of the 2021 Annual Report are incorporated herein by reference except to the extent they address risks arising from or relating to the failure of events described therein to occur, which events have since occurred.

Item 5. Other Information.

None.

Item 6. Exhibits.

Exhibit number	Description
3.1	Fourth Amended and Restated Agreement of Limited Partnership of Summit Midstream Partners, LP, dated May 28, 2020 (Incorporated herein by reference to Exhibit 3.1 to SMLP's Current Report on Form 8-K dated June 2, 2020 (Commission File No. 001-35666))
<u>3.2</u>	Second Amended and Restated Limited Liability Company Agreement of Summit Midstream GP, LLC, dated May 28, 2020 (Incorporated herein by reference to Exhibit 3.2 to SMLP's Current Report on Form 8-K filed June 2, 2020 (Commission File No. 001-35666))
<u>3.3</u>	Certificate of Limited Partnership of Summit Midstream Partners, LP (Incorporated herein by reference to Exhibit 3.1 to SMLP's Form S-1 Registration Statement dated August 21, 2012 (Commission File No. 333-183466))
<u>3.4</u>	Certificate of Formation of Summit Midstream GP, LLC (Incorporated herein by reference to Exhibit 3.4 to SMLP's Form S-1 Registration Statement dated August 21, 2012 (Commission File No. 333-183466))
<u>10.1</u>	+ <u>Purchase and Sale Agreement, dated as of June 9, 2022, among Summit Midstream Holdings, LLC as Seller, Longwood</u> <u>Gathering and Disposal Systems, LP as Buyer, and Summit Midstream Partners LP</u>
<u>10.2</u>	Form of Summit Midstream Partners, LP 2012 Long-Term Incentive Plan Grant Award Agreement (Incorporated herein by reference to Exhibit 10.54 to SMLP's Annual Report on Form 10-K filed February 28, 2022 (Commission File No. 001-35666))
<u>10.3</u>	Summit Midstream Partners, LP 2022 Long-Term Incentive Plan (incorporated herein by reference to Appendix B to SMLP's definitive proxy statement on Schedule 14A filed March 31, 2022 (Commission File No. 001-35666))
<u>22.1</u>	Summit Midstream Partners, LP Subsidiary Issuers and Guarantors of Registered Securities (Incorporated herein by reference to Exhibit 22.1 to SMLP's Report on Form 10-Q filed May 8, 2020 (Commission File No. 001-35666))
<u>31.1</u>	+ Rule 13a-14(a)/15d-14(a) Certification, executed by Heath Deneke, President, Chief Executive Officer and Director
<u>31.2</u>	+ Rule 13a-14(a)/15d-14(a) Certification, executed by William J. Mault, Executive Vice President and Chief Financial Officer
<u>32.1</u>	+ Certifications required by Rule 13a-14(b) or Rule 15d-14(b) and Section 1350 of Chapter 63 of Title 18 of the United States Code (18 U.S.C. 1350), executed by Heath Deneke, President, Chief Executive Officer and Director, and William J. Mault, Executive Vice President and Chief Financial Officer
101.INS	* Inline XBRL Instance Document – the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document
101.SCH	* Inline XBRL Taxonomy Extension Schema
101.CAL	* Inline XBRL Taxonomy Extension Calculation Linkbase
101.DEF	* Inline XBRL Taxonomy Extension Definition Linkbase
101.LAB	* Inline XBRL Taxonomy Extension Label Linkbase
101.PRE	* Inline XBRL Taxonomy Extension Presentation Linkbase
104	* Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101).

⁺ Filed herewith.

^{*} Pursuant to Rule 406T of Regulation S-T, the Interactive Data Files on Exhibit 101 hereto are deemed not filed or part of a registration statement or prospectus for purposes of Sections 11 or 12 of the Securities Act of 1933, as amended, are deemed not filed for purposes of Section 18 of the Securities and Exchange Act of 1934, as amended, and otherwise are not subject to liability under those sections. The financial information contained in the XBRL (eXtensible Business Reporting Language)-related documents is unaudited and unreviewed.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Summit Midstream Partners, LP

(Registrant)

By: Summit Midstream GP, LLC (its General Partner)

August 5, 2022

/s/ WILLIAM J. MAULT

William J. Mault, Executive Vice President and Chief Financial Officer (Principal Financial Officer)

PURCHASE AND SALE AGREEMENT

among

SUMMIT MIDSTREAM HOLDINGS, LLC

as Seller,

LONGWOOD GATHERING AND DISPOSAL SYSTEMS, LP

as Buyer, and

SOLELY FOR THE LIMITED PURPOSES AS SET FORTH IN SECTION 10.27, SUMMIT MIDSTREAM PARTNERS, LP

as Summit Parent, dated as of June 9, 2022

Table of Contents

		Page
	ARTICLE I DEFINITIONS AND CONSTRUCTION	1
1.1 1.2	Definitions 1 Rules of Construction 12	
	ARTICLE II PURCHASE AND SALE AND CLOSING	13
2.1 2.2 2.3 2.4 2.5 2.6 2.7 2.8	Purchase and Sale 13 Purchase Price 13 Closing 14 Closing Deliveries by Seller to Buyer 14 Closing Deliveries by Buyer to Seller 15 Working Capital Adjustment 15 Tax Withholding 16 Express Exclusion of Operations Services from the Purchase and Sale 17	
ARTICLE III	REPRESENTATIONS AND WARRANTIES REGARDING THE COMPA	NIES 17
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9 3.10 3.11 3.12 3.13	Organization; Good Standing; Authority 17 Capitalization of Each Company 18 No Conflicts; Consents and Approvals 18 Financial Statements 19 Absence of Changes 19 Compliance with Applicable Laws 20 Intellectual Property 20 Absence of Litigation 20 Real Property 20 Personal Property 22 Regulatory Status 22 Environmental Matters 22 Taxes 23 Contracts 24	
3.14 3.15 3.16 3.17 3.18	Employees 26 Employee Benefit Plans 26 Transactions with Affiliates 26	
3.19 3.20 3.21	Surety Bonds and Credit 26 No Bankruptcy 26 No Undisclosed Material Liabilities; Indebtedness 26 Insurance 26	
3.22 3.23	Bank Accounts 27 Throughput Data 27	
3.24 3.25 3.26	Capital Commitments; Capital Expenditures 27 Imbalances 27 Preferential Rights 27	
3.27	Data Privacy and Security 27	

Confidential i

	Condition and Sufficiency of the Assets 28	
	ARTICLE IV REPRESENTATIONS AND WARRANTIES REGARDING SELLER	28
4. 4. 4. 4. 4. 4. 4.	Authority 28 Ownership of the Interests 29 No Conflicts; Consents and Approvals 29 Legal Proceedings 29 Broker's Commissions 29 No Bankruptcy 29	
	ARTICLE V REPRESENTATIONS AND WARRANTIES REGARDING BUYER	30
5. 5. 5. 5. 5. 5.	Authority 30 No Conflicts; Consents and Approvals 30 Legal Proceedings 30 Acquisition as Investment 30 Financial Resources 31 Opportunity for Independent Investigation 31 Broker's Commissions 32	
	ARTICLE VI COVENANTS 32	
6. 6. 6. 6. 6. 6.	Access of Buyer 34 Regulatory and Other Approvals 35 Resignation and Removal 35 Payments; Distributions 36 Tax Matters 36 Public Announcements 39 Removal of Name 39	
	22 Termination of Summit Marketing Transaction Confirmation 44	
	Confidential	

ii

6.23	Post-Closing Matters 44
ARTICLE VII	BUYER'S CONDITIONS TO CLOSING 45
7.1 7.2 7.3 7.4 7.5 7.6	Representations and Warranties 45 Performance 45 No Material Adverse Effect 45 Officer's Certificate 45 Orders and Laws 45 Deliverables 45
ARTICLE VII	II SELLER'S CONDITIONS TO CLOSING 45
	Representations and Warranties 46 Performance 46 Officer's Certificate 46 Orders and Laws 46 Deliverables 46 Double E FTSA 46
ARTICLE IX	TERMINATION 46
9.1 9.2 9.3	Right of Termination 46 Effect of Termination 47 Return of Documentation 48
ARTICLE X	MISCELLANEOUS 48
10.13 10.14 10.15 10.16 10.17 10.18 10.19 10.20	Exclusive Remedy 54
10.23	Governing Law; Enforcement 58

10.24 Specific Performance 59 10.25 No Recourse 59 10.26 Legal Representation 60 10.27 Summit Parent 60

Exhibits

Exhibit A - Assignment Agreement Exhibit B - Summit Permian System Exhibit C Transition Services Agreement Exhibit D – Excluded Liability

Exhibit E – R&W Policy

Exhibit F - Double E Release Agreement Exhibit G - IRB Project Site

Exhibit H – Post-Closing Matters

Schedules

Schedule 1.1-PI – Permitted Indebtedness

Schedule 1.1-NWC — Sample Net Working Capital Calculation Schedule 1.1-PL — Permitted

Schedule 2.2(c) — Assumed Indebtedness Schedule 2.4(e) — Required Consents Schedule 3.2(a) — Capitalization Schedule 3.2(f) — Equity Interests Schedule

3.3 – Consents

Schedule 3.5 – Absence of Changes

Schedule 3.6 — Compliance with Applicable Laws Schedule 3.6-P — Permits Schedule 3.8 — Absence of Litigation Schedule 3.9(a) — Owned Real Property Schedule 3.9(b) — Real Property Leases Schedule 3.9(c) — Real Property

Easements Schedule 3.9(e) – Gaps

Schedule 3.9(g) — Certain Real Property and Easements Schedule 3.10 — Personal Property Schedule 3.12 — Environmental Matters Schedule 3.12(a)-P — Environmental

Permits Schedule 3.13 - Taxes

Schedule 3.14 - Material Contracts Schedule 3.17 - Transactions with Affiliates

Schedule 3.18 - Surety Bonds and Credit

Schedule 3.20 – Material Liabilities; Indebtedness Schedule 3.21 – Insurance

Schedule 3.22 - Bank Accounts; Powers of Attorney Schedule 3.23 - Throughput Data

Schedule 3.24 – Capital Commitments Schedule 3.25 – Imbalances

Schedule 3.28 — Open Orders Schedule 4.4 — Conflicts; Consents and Approvals Schedule 6.1 — Interim Period

Operations

Schedule 6.17 – Continuing Employees

PURCHASE AND SALE AGREEMENT

This PURCHASE AND SALE AGREEMENT, dated as of June 9, 2022 (this "*Agreement*"), is made and entered into by and among Summit Midstream Holdings, LLC, a Delaware limited liability company ("*Seller*"), Longwood Gathering and Disposal Systems, LP, a Texas limited partnership ("*Buyer*"), and solely for the limited purposes as set forth in <u>Section 10.27</u>, Summit Midstream Partners, LP, a Delaware limited partnership ("*Summit Parent*").

RECITALS

WHEREAS, Seller desires to sell to Buyer, and Buyer desires to purchase from Seller, 100% of the issued and outstanding Equity Interests (as defined herein) (the "*Interests*") in Summit Midstream Permian, LLC, a Delaware limited liability company ("*Summit Permian*"), on the terms and subject to the conditions set forth herein;

WHEREAS, Summit Permian owns 100% of the issued and outstanding limited liability company interests in Summit Midstream Permian Finance, LLC, a Delaware limited liability company ("*Permian Finance*"; Summit Permian and Permian Finance are each referred to herein as a "*Company*" and collectively referred to herein as the "*Companies*"); and

WHEREAS, contemporaneously with the Closing, Summit Marketing and MRC Permian, as Affiliates of the Seller and Buyer, respectively, shall enter into that certain Prearranged Permanent Capacity Release Agreement, dated as of the Closing Date, by and between Summit Marketing and MRC Permian, in substantially the form attached hereto as Exhibit F (the "Double E Release Agreement").

AGREEMENT

NOW, THEREFORE, in consideration of the premises and the mutual representations, warranties, covenants and agreements in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties hereby agree as follows:

ARTICLE I DEFINITIONS AND CONSTRUCTION

1.1 *Definitions*. As used in this Agreement, the following capitalized terms have the meanings set forth below:

"1933 Act" means the United States Securities Act of 1933, as amended. "Acquired Business" has the meaning

given to it in Section 6.14(c).

"Actual Fraud" means (i) an Executive Person made an oral representation or any Representative of the Seller made a written representation in this Agreement, any Transaction Document or in the Data Room, either of which was false, with the actual and present knowledge on the part of the Person making such representation that such representation was false, (ii) the Person making such representation intended to induce another Person party to this Agreement to

act upon such representation, (iii) such Person party to this Agreement to whom such representation was made actually and justifiably relied upon such representation, and (iv) such Person to whom such representation was made suffered injury as a direct result of relying upon such representation.

- "Affiliate" means, with respect to any Person, a Person directly or indirectly controlling, controlled by or under common control with such Person. The Companies shall be considered Affiliates of Seller prior to and at Closing and Affiliates of Buyer after Closing. In this context "control" means the possession, directly or indirectly, through one or more intermediaries, by any Person or group (within the meaning of Section 13(d)(3) under the United States Securities Exchange Act of 1934, as amended) of one or both of the following:
- (a) (i) in the case of a corporation, more than 50% of the direct or indirect economic interest in the outstanding equity securities thereof; (ii) in the case of a limited liability company, partnership, limited partnership or venture, the right to more than 50% of the distributions therefrom (including liquidating distributions); (iii) in the case of a trust or estate, including a business trust, more than 50% of the beneficial interest therein; and (iv) in the case of any other entity, more than 50% of the economic or beneficial interest therein; or
- (b) in the case of any entity, the power or authority, through ownership of voting securities, by contract or otherwise, to control or direct the management and policies of the entity.
 - "Agreement" has the meaning given to it in the Preamble.
 - "Allocation Schedule" has the meaning given to it in Section 6.6(f)(i).
 - "Arbitration" is defined in Exhibit D.
- "Assets" of any Person means all properties of every kind, nature, character and description (whether real, personal or mixed, whether tangible or intangible and wherever situated), including the goodwill related thereto, operated, owned or leased by such Person, and with respect to the Companies, includes all such properties used in connection with the Business.
 - "Assignment Agreement" has the meaning given to it in Section 2.4(a).
 - "Base Purchase Price" has the meaning given to it in Section 2.2(a).
- "*Business*" means the development, ownership and operation by Seller of the Summit Permian System, and other activities that are incidental thereto, and all other activities and operations of the Companies conducted during the 12-month period prior to the date hereof.
- "Business Day" means a day other than Saturday, Sunday or any day on which banks located in the State of Texas are authorized or obligated to close.
 - "Buyer" has the meaning given to it in the Preamble.

- "Buyer Fundamental Representations" means the representations and warranties of Buyer in Article V; provided, however, that Section 5.3(b), Section 5.3(c) and Section 5.3(d) (No Conflicts; Consents and Approvals) shall not constitute Buyer Fundamental Representations.
 - "Buyer Indemnified Parties" has the meaning given to it in Section 10.3.
 - "Buyer Released Claim" has the meaning given to it in Section 6.13(b).
 - "Buyer Released Party" has the meaning given to it in Section 6.13(a).
- "Cash" means money, currency or a credit balance in a deposit account at a financial institution, net of checks outstanding as of the time of determination.
- "Change of Control" means the occurrence of (a) any consolidation or merger of Summit Parent with or into any other entity, or any other corporate reorganization or transaction (including the acquisition or issuance of common equity (stock, partnership units, or otherwise) of Summit Parent), whether or not Summit Parent is a party thereto, after giving effect to which (i) in excess of 50% of the Common Units or Summit Parent's voting power is beneficially owned by any Person or "group" (as such term is used in Rule 13d-5 under the Exchange Act) that is not an Affiliate of Summit Parent at the time of such Change of Control or (ii) the common equity holders of Summit Parent (through stock, partnership units, or otherwise) and their respective Affiliates prior to such consolidation, merger, reorganization or other transaction shall cease to have a majority of the voting power of, or the power to elect a majority of the entire board of the directors of, Summit Parent or other surviving entity; or (b) a sale of all or substantially all of the assets of Summit Parent and its Affiliates to any Person or "group" (as such term is used in Rule 13d-5 under the Exchange Act) that is not an Affiliate of Summit Parent at the time of the Change of Control.
- "Charter Documents" means with respect to any Person that is not a natural person, the articles of incorporation or organization, memorandum of association, articles of association and by-laws, the limited partnership agreement, the partnership agreement or the limited liability company agreement and/or such other organizational documents of such Person which establish the legal personality of such Person.
- "*Claim*" means any demand, claim, action, cause of action, inquiry, audit, notice of violation, citation, summons, subpoena, investigation, review or Proceeding.
 - "Closing" has the meaning given to it in Section 2.3.
 - "Closing Date" means June 30, 2022, unless the Parties mutually agree to an earlier date for the Closing.
 - "Closing Purchase Price" means the Purchase Price determined at the Closing pursuant to Section 2.2.
 - "Code" means the United States Internal Revenue Code of 1986, as amended.

"Common Units" means the common units representing limited partner interests in Summit Parent.

"Company" or "Companies" has the meaning given to it in the Recitals. "Competitive Activities" has the

meaning given to it in Section 6.14(c).

"Confidential Material" means all information (written or oral) that is confidential, proprietary or not otherwise generally available to the public that relates to any Company or the Business, whether disclosed prior to, on or after the date hereof. The term "Confidential Material" shall not include (a) information that is or becomes generally available to the public, other than as a result of disclosure by Seller, its Affiliates or Representatives in violation of this Agreement,

(b) information that is or becomes available to Seller, its Affiliates or its or their Representatives from a Person other than Buyer or any Company on a non-confidential basis, *provided*, that such Person was not reasonably known by Seller, its Affiliates or its Representatives to be bound by a confidentiality agreement with or other contractual, legal or fiduciary obligation of confidentiality to Buyer, any Company or any of the Representatives of a Company with respect to such materials, or (c) information developed by Seller or any of its Affiliates independently of and without use or reference to any Confidential Material.

"*Confidentiality Agreement*" means that certain Confidentiality Agreement by and between Summit Parent and MRC Permian Company, dated as of March 8, 2022.

"Continuing Employee" has the meaning given it in Section 6.17.

"Contract" means any legally binding contract, lease, license, evidence of Indebtedness, mortgage, indenture, purchase order, binding bid, letter of credit, security agreement, deed of trust or arrangement, in each case, whether written or oral.

"*Data Room*" shall mean the data site maintained by Seller and its Representatives in connection with the transactions completed in this Agreement.

"Data Security Requirements" means the following, to the extent applicable to the Companies, in each case solely to the extent relating to any privileged or proprietary information that, if compromised through any theft, interruption, modification, corruption, loss, misuse or unauthorized access or disclosure, would cause serious harm to the organization owning it, or any matters relating to data privacy, protection or security or the access, collection, use, processing, storage, sharing, distribution, transfer, disclosure, destruction or disposal of any such information:

(i) all Privacy and Security Laws and other applicable Laws; (ii) the Companies' respective rules, policies and procedures; (iii) Contracts to which any Company is bound.

"Dispute" has the meaning given to it in Section 10.11.

"Double E" shall mean Double E Pipeline, LLC, a Delaware limited liability company.

"*Double E FTSA*" shall mean, collectively, (i) that certain Firm Transportation Services Agreement No. FTS004, by and between Double E and Summit Marketing; and (ii) that certain

Negotiated Rate Letter Agreement to Firm Transportation Services Agreement No. FTS004, by and between Double E and Summit Marketing.

"Double E Pipeline" means the pipeline system owned by Double E as of the date hereof. "Double E Release

Agreement" has the meaning given to it in the Recitals.

"Due Diligence Information" has the meaning given to it in Section 5.7(b).

"Environmental Laws" means any and all federal, state and local Laws pertaining to pollution, protection of the environment or natural resources, or the Release of, or exposure to, Hazardous Materials, including the Clean Air Act, the Federal Water Pollution Control Act, the Oil Pollution Act of 1990, the Rivers and Harbors Act of 1899, the Safe Drinking Water Act, the Comprehensive Environmental Response, Compensation and Liability Act, the Superfund Amendments and Reauthorization Act of 1986, the Resource Conservation and Recovery Act, the Hazardous and Solid Waste Amendments Act of 1984, the Toxic Substances Control Act, and comparable state and local counterparts.

"Environmental Permits" means any Permit issued pursuant to Environmental Laws. "Equity Interests" means capital

stock, partnership or membership interests or units

(whether general or limited), and any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distribution of assets of, the issuing entity.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"*Estimated Net Working Capital*" means Seller's good faith estimate of the Net Working Capital as of the close of business on the Business Day immediately preceding the Closing Date.

"Executive Person" means each of J. Heath Deneke, Ryan Simmons, James Johnston, Hugo Guerrero, Bill Mault and Dakota Lee.

"Excluded Claims" are defined in Exhibit D.

"*Excluded Liabilities*" means all Losses arising from, based upon, related to or associated with the matters set forth on Exhibit D.

"FERC" means the Federal Energy Regulatory Commission.

"*Final Purchase Price*" means the final adjusted Purchase Price determined as of the Final Purchase Price Determination Date pursuant to <u>Section 2.6(b)</u>.

"Final Purchase Price Determination Date" has the meaning given to it in Section 2.6(b).

"Financial Statements" has the meaning given to it in Section 3.4.

"*FTSA*" means that certain Firm Transportation Service Agreement attached as <u>Attachment A</u> to the Double E Release Agreement by and between Double E and MRC Permian.

"Fundamental Representations" means the Seller Fundamental Representations and the Buyer Fundamental Representations, collectively.

"GAAP" means generally accepted accounting principles in the United States, applied on a consistent basis.

"Governmental Authority" means any applicable federal, state, tribal or local governmental authority, agency, board, commission, council court or official in the United States or political subdivision thereof, or any agency or instrumentality of such government or political subdivision, or any self-regulated organization or other non-governmental regulatory authority or quasi-governmental authority (to the extent that the rules, regulations or orders of such organization or authority have the force of Law), or any arbitrator, court or tribunal of competent jurisdiction.

"Hazardous Materials" means each substance designated or classified as a hazardous waste, hazardous substance, hazardous material, pollutant, contaminant or toxic substance under any Environmental Laws, including any petroleum or petroleum-derived substance, waste or additive, asbestos, asbestos-containing material, polychlorinated biphenyl, per- or polyfluoroalkyl substance or radioactive compound.

"Indebtedness" means, with respect to any Person, at any date, without duplication, (a) all obligations of such Person for borrowed money, including all principal, interest, premiums, fees, expenses, overdrafts and, to the extent required to be carried on a balance sheet prepared in accordance with GAAP, penalties with respect thereto, whether short-term or long-term, and whether secured or unsecured, or with respect to deposits or advances of any kind (other than deposits and advances of any Person relating to the purchase of products or services from any Company in the ordinary course of business), (b) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments or debt securities, (c) all obligations of such Person to reimburse any bank or other Person in respect of amounts paid under a letter of credit or bankers' acceptances or similar instruments, (d) all obligations of such Person under conditional sale or other title retention agreements relating to property or assets purchased by such Person,

(e) all guarantees, whether direct or indirect, by such Person of Indebtedness of others, (f) obligations of such Person under a lease and other similar arrangements conveying the right to use that are required to be classified under GAAP as a liability (including pursuant to Accounting Standards Update, Leases (Topic 842)), (g) with respect to the Companies, any Transaction Expenses that remain unpaid as of the Closing and (h) all other obligations of a Person which would be required to be shown as indebtedness on a balance sheet of such Person prepared in accordance with GAAP; provided, that Indebtedness shall not include accounts payable to trade creditors and accrued expenses arising in the ordinary course of business and shall not include the endorsement of negotiable instruments for collection in the ordinary course of business that are reflected on the Financial Statements and included in the calculation of Net Working Capital; provided, further, that any items set forth on Schedule 1.1-PI shall not be treated as items of Indebtedness.

"Indemnification Claim" has the meaning given to it in Section 10.7(a).

- "*Indemnified Parties*" means (a) with respect to the Losses described in <u>Section 10.3</u>, the Buyer Indemnified Parties, and (b) with respect to the Losses described in <u>Section 10.4</u>, the Seller Indemnified Parties.
- "*Indemnifying Party*" means (a) with respect to the Losses described in <u>Section 10.3</u>, the Seller, and (b) with respect to the Losses described in <u>Section 10.4</u>, the Buyer.
- "*Intellectual Property Rights*" means material rights in any of the following to the extent subject to protection under applicable Law: (a) trademarks, service marks and trade names; (b) patents; (c) copyrights; (d) internet domain names; (e) licenses, software, systems, trade secrets and other proprietary and
- confidential information; and (f) any registrations or applications for registration for any of the foregoing.

"Interests" has the meaning given to it in the Recitals. "Interim Period" has the meaning

given to it in Section 6.1(a).

- "*IRB Arrangement*" means that certain Industrial Revenue Bond (Summit Midstream Permian, LLC Project) Series 2017 issued by the State of New Mexico and the related Lease Agreement by and between Eddy County, New Mexico and Summit Permian, dated as of December 1, 2017.
- "IRB Assets" means the various real property located in the unincorporated portion of Eddy County, New Mexico and all improvements installed and to be installed thereon or therein, all equipment, furniture, furnishings, machines, packaging, process and other systems and all other personal property of any kind which is subject to depreciation for federal income tax purposes and is suitable for use and used in the IRB Facility, including the natural gas processing facility, three gathering compressor stations and certain gathering pipelines and residue gas outlet and takeaway gas pipelines, including, without limitation, the IRB Project Site.
- "*IRB Facility*" means a natural gas processing facility operated or to be operated by Summit Permian and located approximately 25 miles east-northeast of Carlsbad, New Mexico, and related pipelines, including gathering pipelines and compressor stations.
 - "IRB Project Site" is defined on Exhibit G.
 - "*Knowledge*" when used in a particular representation or warranty in this Agreement with respect to Seller, means the actual knowledge of an Executive Person, after reasonable due inquiry.
- "*Laws*" means all laws, statutes, rules, regulations, ordinances, court or other orders of a Governmental Authority and other pronouncements having the effect of law of any Governmental Authority.
- "*Lien*" means any mortgage, pledge, deed of trust, assessment, security interest, charge, lien (statutory or other), encumbrance, equitable interest, option, warranty, purchase right, right of first refusal or offer, lease, or other similar burden or property interest.

"Loss" means any and all judgments, losses, liabilities, amounts paid in settlement, awards, damages, Taxes, Claims, fines, penalties, deficiencies, costs or expenses (including interest, court costs, reasonable fees of attorneys, accountants and other experts, other reasonable expenses of litigation or other Proceedings or of any Claim, default or assessment and the cost of investigation and enforcement of any right to indemnification hereunder and the cost of pursuing any insurance providers). For the avoidance of doubt, any retention amount owed under the R&W Policy shall not be considered in the calculation of Loss.

"Material Adverse Effect" when used with respect to the Companies, the Business, or the Companies' Assets, means any change, circumstance or effect that is materially adverse to (a) the business, operations, results of operations or financial condition of the Companies taken as a whole, or (b) the validity or enforceability of this Agreement, but excluding (in the case of clause

(a) hereof) any of the foregoing resulting from (i) general economic, financial and credit conditions, (ii) changes or conditions in financial or securities markets and interest and exchange rates, (iii) changes in, or conditions affecting, the price of hydrocarbons, the oil and gas midstream industry generally or the Delaware and Permian Basins specifically (including any reduction in drilling activity or associated reduction in production), (iv) the execution of this Agreement and the announcement or completion thereof, (v) any act or omission by Seller or the Companies taken as required by or permitted under this Agreement or with prior written consent of Buyer, (vi) any act or omission of Buyer, (vii) changes in national or international political conditions, or any shutdown of the United Sates federal government or any default on the debt obligations of any sovereign entity, (viii) the outbreak or escalation of hostilities involving the United States, the declaration by the United States of a national emergency, acts of war (whether or not declared) or any similar disorder, including acts of terrorism, sanctions, interruptions of trade, embargoes and pandemics (whether regional, national or international), and any actions taken in response thereto.

(ix) acts of God, including any hurricane, flood, tornado, earthquake or other natural disaster or climate change and any escalation or worsening thereof and any actions taken in response thereto,

(x) any change in applicable Laws or accounting requirements or principles imposed upon the Companies or their respective businesses, including GAAP, or the interpretations thereof, (xi) any change in the credit rating and/or outlook of Summit Parent or any of the Companies or any of their respective securities (except that the underlying causes of any such changes may be considered in determining whether a Material Adverse Effect has occurred), (xii) any decline in market price of Summit Parent's common units or a failure of Summit Parent or the Companies to meet any internal or external projections or forecasts or any estimates of revenue or earnings for any period (except that the underlying causes of any such changes may be considered in determining whether a Material Adverse Effect has occurred), (xiii) natural declines in well performance or reclassification or recalculation of reserves, (xiv) the failure of FERC to have or maintain a quorum that allows it to render orders and (xv) any matters or disclosures set forth in the Schedules as of the date hereof; but, solely in the case of the foregoing clauses (viii) and (ix), only to the extent that the Companies, taken as a whole, are adversely affected in a materially disproportionate manner as compared to other participants in the oil and gas midstream industry in the United States.

"Material Contracts" has the meaning given to it in Section 3.14(a).

"MRC Permian" means MRC Permian Company, a Texas corporation.

- "NRA" means that certain Negotiated Rate Agreement attached as Attachment B to the Double E Release Agreement by and between Double E and MRC Permian.
- "Net Working Capital" means, as of a particular date or time, (a) the sum of the current assets of the Companies other than Cash less (b) the sum of the current liabilities of the Companies. Notwithstanding anything in this Agreement to the contrary, all Tax assets and liabilities will be excluded from the calculation of Net Working Capital. For illustrative purposes only, attached as <u>Schedule 1.1-NWC</u> is a sample calculation of Net Working Capital prepared by the Parties as of March 31, 2022.
 - "NGA" means the Natural Gas Act of 1938.
 - "Non-Company Affiliate" means any Affiliate of Seller, except for the Companies.

 - "Non-Recourse Party" has the meaning given to it in <u>Section 10.25</u>. "Operations Services" has the meaning given to it in <u>Section 2.8</u>.
 - "Outside Date" has the meaning given to it in Section 9.1(b).
 - "Party" means each of Buyer and Seller and "Parties" means, together, Buyer and Seller. "Payoff Letter" has the meaning given to it in Section 2.4(g).
 - "Permian Finance" has the meaning given to it in the Recitals.
 - "Permits" means all permits, licenses or authorizations from any Governmental Authority.
- "Permitted Lien" means, as to any Company, (a) any Lien for Taxes not yet due or payable or being contested in good faith by appropriate proceedings, (b) any Lien arising in the ordinary course of business by operation of Law with respect to a liability that is not yet due or payable or which is being contested in good faith by or on behalf of such Company, (c) all matters that are disclosed in the deed or instrument conveying such property have either been disclosed on Schedule 1.1-PL or are not material to the Business, (d) purchase money Liens arising in the ordinary course of business, (e) zoning, planning and other similar limitations and restrictions and all rights of any Governmental Authority to regulate a property, (f) the terms and conditions of the Permits or the Contracts of such Company set forth in the Schedules, (g) pledges and deposits made in the ordinary course of business in compliance with workers' compensation, unemployment insurance and other social security Laws, (h) any Lien to be released on or prior to Closing and (i) the other matters identified on Schedule 1.1-PL.
- "Person" means any natural person, corporation, general partnership, limited partnership, limited liability company, unlimited liability corporation, proprietorship, other business organization, trust, union, association or Governmental Authority.
 - "Personal Property" has the meaning given to it in Section 3.10.

"Plan" means, whether written or oral, each "employee benefit plan" within the meaning of Section 3(3) of ERISA (including "multiemployer plans" within the meaning of Section 3(37) of ERISA) and any and all employment, deferred compensation, change in control, severance, termination, loan, employee benefit, retention, bonus, pension, profit sharing, savings, retirement, welfare, incentive compensation, stock or equity-based compensation, stock purchase, stock appreciation, collective bargaining, fringe benefit, vacation, paid time off, sick leave or other similar agreements, plans, programs, policies, understandings or arrangements.

"Pre-Closing Period" means any Tax period of either Company ending on or before the Closing Date.

"Privacy and Security Laws" means all applicable Laws concerning data protection, privacy, security, the access, collection, use, processing, storage, sale, distribution, transfer, disclosure, destruction, or disposal of data including personal information and trade secrets or other similar Laws (including any security breach notification requirements), including HIPAA, HITECH, the Gramm-Leach-Bliley Act, the Fair Credit Reporting Act, the Fair and Accurate Credit Transaction Act, the Privacy Act of 1974, the CAN-SPAM Act, the Telephone Consumer Protection Act, the Telemarketing and Consumer Fraud and Abuse Prevention Act, Children's Online Privacy Protection Act, state Social Security number protection Laws, state data breach notification Laws, and state consumer protection Laws.

"Privileged Communications" has the meaning given to it in Section 10.26.

"*Proceeding*" means any complaint, lawsuit, action, suit, arbitration or other proceeding at Law or in equity or order or ruling, in each case by or before any Governmental Authority or arbitral tribunal.

"Purchase Price" has the meaning given to it in Section 2.2.

"*R&W Policy*" means the buyer-side representations and warranties insurance policy, obtained by Buyer as of the date hereof and attached hereto as Exhibit E.

"*Real Property*" means the real property owned in fee or leased, used or held for use by a Person together with all buildings, structures, improvements, fixtures and other interests in real property located thereon, and all easements, rights of way and other rights and interests appurtenant thereto.

"*Records*" has the meaning given to it in <u>Section 10.10</u>.

"*Release*" means any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, seeping, dumping, or disposing.

"*Representatives*" means, as to any Person, its officers, directors, employees, managers, members, partners, shareholders, owners, counsel, accountants, financial advisers and consultants.

"Restricted Party" and "Restricted Parties" have the meanings given to such terms in Section 6.14(a).

- "*Schedules*" means the disclosure schedules prepared by Seller and attached to this Agreement, as such may be updated pursuant to <u>Section 6.15</u>.
 - "Seller" has the meaning given to it in the Preamble.
- "Seller Fundamental Representations" means the representations and warranties of Seller in Section 3.1 (Organization; Good Standing; Authority), Section 3.2 (Capitalization of Each Company), Section 3.3(a) (No Conflicts; Consents and Approvals), and Article IV (Representations and Warranties Regarding Seller); provided, however, that Section 4.4(b), Section 4.4(c), and Section 4.4(d) (No Conflicts; Consents and Approvals) and Section 4.8 (Financial Resources) shall not constitute Seller Fundamental Representations.
 - "Seller Group" means Seller and any of its Affiliates.
 - "Seller Indemnified Party" has the meaning given to it in Section 10.4.
 - "Seller Released Claim" has the meaning given to it in Section 6.13(a).
 - "Seller Released Party" has the meaning given to it in Section 6.13(b).
 - "Straddle Period" means any Tax period of either Company that begins before and ends after the Closing Date.
- "Subject Employee" means those certain employees whose principal employment duties involve providing services with respect to the Companies in furtherance of the Business.
 - "Summit Marketing" means Summit Midstream Marketing, LLC, a Delaware limited liability company.
 - "Summit Parent" has the meaning given to it in the Preamble. "Summit Permian" has the

meaning given to it in the Recitals.

- "Summit Permian System" means that certain natural gas gathering pipeline system and related compression, dehydration and processing equipment located in the northern Delaware Basin in Eddy and Lea Counties, New Mexico, including, but not limited to, the Lane Gas Processing Facility, a gas processing facility located in Eddy County, all as more specifically depicted in Exhibit B, and all interconnects with upstream producers and downstream providers.
 - "Supplement" has the meaning given to it in Section 6.15.
- "Tax" or "Taxes" means (a) all federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Code), customs, duties, capital stock, franchise, margins, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, wealth, net wealth, net worth, alternative or add-on minimum, estimated or any other taxes of any kind, whatsoever, including any interest, penalty, fines, or addition thereto, whether disputed or not and including

any obligations to indemnify or otherwise assume or succeed to the Tax liability of any other Person or (b) any liability for the payment of any taxes, interest, penalty, addition to tax or like additional amount resulting from the application of Treasury Regulation Section 1.1502-6 or comparable federal, state or local requirement of Law (including, but not limited to, Texas Tax Code Chapter 171).

"Tax Proceeding" has the meaning given to it in Section 6.6(d).

- "*Tax Return*" means any return, report, rendition, claim for refund, statement, information return or other document (including any related or supporting information or schedule thereto, or amendment thereof) filed or required to be filed with any Governmental Authority in connection with the determination, assessment, collection or administration of any Taxes or the administration of any Laws relating to any Taxes.
- "*Transaction Confirmation*" that certain Transaction Confirmation by and between Summit Permian and Summit Marketing, dated as of November 18, 2021.
- "*Transaction Document*" means each of this Agreement, the Assignment Agreement, the Transition Services Agreement and each other agreement, document and instrument required to be executed and delivered in accordance with this Agreement.
- "*Transaction Expenses*" means the amounts owed by the Companies to any third party who provides services to Seller or the Companies in connection with the transactions contemplated by this Agreement.
 - "Transfer Taxes" has the meaning given to it in Section 6.6(c).
- "*Transition Services Agreement*" means a Transition Services Agreement to be entered into at Closing by and among the Buyer, the Companies and the Seller, in substantially the form attached hereto as <u>Exhibit C</u>.

1.1 **Rules of Construction**.

- (a) The exhibits and Schedules attached to this Agreement constitute a part of this Agreement and are incorporated herein for all purposes. All article, section, subsection and exhibit references used in this Agreement are to articles, sections, subsections and exhibits to this Agreement unless otherwise specified.
- (b) The headings preceding the text of articles and sections included in this Agreement and the headings to the Schedules attached to this Agreement are for convenience only and shall not be deemed part of this Agreement or be given any effect in interpreting this Agreement.
- (c) If a term is defined as one part of speech (such as a noun), it shall have a corresponding meaning when used as another part of speech (such as a verb). Unless the context of this Agreement clearly requires otherwise, words importing the masculine gender shall include the feminine and neutral genders and vice versa. The words "includes" or "including" shall mean "including without limitation," the words "hereof," "hereby," "herein," "hereunder" and similar

terms in this Agreement shall refer to this Agreement as a whole and not any particular section or article in which such words appear. All currency amounts referenced herein are in United States Dollars unless otherwise specified. The singular shall include the plural and the plural shall include the singular wherever and as often as may be appropriate. The words "shall" and "will" are interchangeably used throughout this Agreement and shall accordingly be given the same meaning, regardless of which word is used.

- (d) Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. Whenever any action must be taken hereunder on or by a day that is not a Business Day, then such action may be validly taken on or by the next day that is a Business Day.
- (e) All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP and shall be calculated in a manner consistent with that used in preparing the Financial Statements to the extent that the Financial Statements are prepared in accordance with GAAP, except as otherwise expressly stated in this Agreement.
- (f) Any reference herein to any Law, statute, rule or regulation shall be construed as referring to such Law, statute, rule or regulation as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time and references to particular provisions of a Law include a reference to the corresponding provisions of any prior or succeeding Law.
- (g) Each Party acknowledges that it and its attorneys have been given an equal opportunity to negotiate the terms and conditions of this Agreement and that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party or any similar rule operating against the drafter of an agreement shall not be applicable to the construction or interpretation of this Agreement.

ARTICLE II PURCHASE AND SALE AND CLOSING

- 2.1 **Purchase and Sale**. On the terms and subject to the conditions set forth in this Agreement, Buyer agrees to purchase from Seller, and Seller agrees to sell and to convey to Buyer, at Closing, the Interests free and clear of all Liens other than those under state or federal securities Laws.
- 2.2 **Purchase Price**. The purchase price for the purchase and sale of the Interests is equal to the sum of the following (the "**Purchase Price**"):
 - (a) \$75,000,000 (the "Base Purchase Price");
 - (b) *plus* (if positive) or *minus* (if negative) the Net Working Capital as of the Closing Date;
- (c) *minus* Indebtedness of the Companies, which shall be paid by Buyer on behalf of the Companies to all holders of such Indebtedness, by wire transfer of immediately

available funds, in such amounts and to such accounts as set forth in the Payoff Letters, save and except for such Indebtedness set forth on Schedule 2.2(c); and

- (d) *plus* the amount of Cash held by the Companies as of the Closing.
- 2.3 *Closing*. The consummation of the purchase and sale of the Interests (the "*Closing*") shall take place either (a) via electronic exchange of documents or (b) at the offices of Summit Midstream Partners, LP, 910 Louisiana St., Suite 4200, Houston, Texas 77002 at 10:00

A.M. local time, on the Closing Date; *provided*, *however*, the Closing shall be deemed effective as of 11:59 PM in Houston, Texas on the Closing Date. All actions listed in <u>Section 2.4</u> or <u>Section 2.5</u> that occur on the Closing Date shall be deemed to occur simultaneously at the Closing.

- 2.4 *Closing Deliveries by Seller to Buyer*. At the Closing, Seller shall deliver, or shall cause to be delivered, to Buyer the following:
- (a) a counterpart duly executed by Seller of an assignment in the form attached hereto as <u>Exhibit A</u> evidencing the assignment and transfer to Buyer of the Interests (the "*Assignment Agreement*");
 - (b) a duly completed and executed Internal Revenue Service Form W-9 with respect to Summit Parent;
- (c) the resignation or removal (effective as of Closing) of managers, officers and directors of each of the Companies, as applicable, nominated or appointed by Seller or its Affiliates;
- (d) a certificate, dated as of the Closing Date, in form and substance reasonably satisfactory to Buyer, duly executed by an officer of Seller certifying (A) the authenticity and effectiveness of the actions of the members, managers or officers of the Seller, as appropriate, authorizing the execution of this Agreement and the Transaction Documents to which Seller is a party; (B) the Charter Documents of each Company; and (C) the incumbency of the officers signing this Agreement or any of the Transaction Documents on behalf of Seller (together with specimen signatures);
- (e) fully executed copies of all registrations, filings, applications, notices, consents, approvals, orders, qualifications and waivers listed on <u>Schedule 2.4(e)</u> and <u>Schedule 3.3</u>, in form and substance reasonably satisfactory to Buyer;
- (f) a certificate of good standing dated not more than ten (10) Business Days prior to the Closing Date with respect to each Company and Seller issued by the Secretary of State of the State of their formation;
- (g) payoff letters evidencing the repayment of all of the Indebtedness, including customary contingent lien release and commitment termination language, subject only to payment of the payoff amounts included therein, in each case in form and substance reasonably satisfactory to Buyer, duly executed and delivered by the applicable payees identified therein (the "Payoff Letters");

- (h) the Transition Services Agreement, duly executed by the Seller;
- (i) subject to Buyer's compliance with <u>Section 6.20</u>, Seller shall cause Summit Marketing to deliver an executed counterpart to the Double E Release Agreement, and Seller shall cause Double E to deliver an executed counterpart to the the FTSA and the NRA;
- (j) evidence of the full repayment and termination of the IRB Arrangement and the purchase of the IRB Assets by Summit Permian; and
 - (k) evidence of the termination of the Transaction Confirmation.
- 2.5 *Closing Deliveries by Buyer to Seller*. At the Closing, Buyer shall deliver, or shall cause to be delivered, to Seller the following:
- (a) a wire transfer or transfers of immediately available funds (to such account or accounts of Seller as Seller shall have notified Buyer of at least two Business Days prior to the Closing Date) in an amount or amounts in the aggregate equal to the Purchase Price;
- (b) to the holders of Indebtedness, the amounts to such accounts as set forth in the Payoff Letters in accordance with Section 2.2(c);
 - (c) a counterpart of the Transition Services Agreement, duly executed by the
- (d) Buyer shall cause MRC Permian to deliver a counterpart of the Double E Release Agreement, the FTSA and the NRA; and
 - (e) a counterpart duly executed by Buyer of the Assignment Agreement.

2.6 Working Capital Adjustment.

- a. Seller and Buyer shall cooperate and provide each other access, including through electronic means, to the Companies' books and records as is reasonably requested in connection with the matters addressed in this <u>Section 2.6</u>. Within three Business Days prior to the expected Closing Date, Seller shall in good faith estimate (i) the Estimated Net Working Capital and its components, (ii) the amount of Cash held by the Companies as of the close of business on the Business Day immediately preceding the Closing Date, (iii) the amount of Indebtedness owed by the Companies as of the close of business on the Business Day immediately preceding the Closing Date, and (iv) the resulting calculation of the Closing Purchase Price. Prior to the Closing, Seller shall provide such estimates to Buyer together with reasonable supporting documentation. For purposes of calculating Net Working Capital, all payments made at Closing pursuant to <u>Section 6.5</u> shall be deemed to have been paid immediately prior to the close of business on the Business Day immediately preceding the Closing Date.
- b. Within 60 days after Closing, Seller shall provide Buyer with its good faith final calculation of the actual amounts for each of the estimated amounts required by <u>Section 2.6(a)</u>, which delineates any differences from such estimates together with reasonable supporting documentation. If Buyer disagrees with any of the calculations provided by Seller

pursuant to the notice referenced in the foregoing sentence, then Buyer shall provide Seller with written notice thereof within 30 days after Buyer's receipt of notice of such calculations and shall include reasonable detail regarding such specific objections. If Buyer and Seller working in good faith are unable to agree on such disputed items on or prior to the 120th day following the Closing Date, then either Party may refer such dispute to PricewaterhouseCoopers LLP or, if that firm declines to act as provided in this Section 2.6(b), another firm of independent public accountants, mutually acceptable to Buyer and Seller, which firm shall make a final and binding determination as to all matters in dispute on a timely basis and promptly shall notify the Parties in writing of its resolution. Such accounting firm handling the dispute resolution shall not have the power to modify or amend any term or provision of this Agreement. The costs of such review and report shall be borne by Buyer, on the one hand, and the Seller, on the other hand, in proportion to the aggregate dollar amounts of such dispute(s) resolved in favor of the Buyer compared to the aggregate dollar amounts of such dispute(s) resolved in favor of the Seller (as determined by the accounting firm). For example, if Seller claims the appropriate adjustments are \$100,000 more than the amount determined by Buyer, and Buyer contests only \$50,000 of the amount claimed by Seller, and if the accounting firm ultimately resolves the dispute by awarding Seller \$30,000 of the \$50,000 contested, then the fees and expenses of the accounting firm will be allocated 60% (i.e. 30/50) to Buyer and 40% (i.e., 20/50) to Seller. Except as provided in the immediately preceding two sentences, all other costs and expenses incurred by the Parties in connection with resolving any dispute hereunder before the accounting firm shall be borne by the Party incurring such cost and expense. If Buyer does not object to Seller's calculations within the time period and in the manner set forth in the first sentence of this Section 2.6(b) or accepts Seller's calculations, then such calculations as set forth in Seller's notice shall become final and binding upon the Parties for all purposes hereunder. The date that such calculations pursuant to this <u>Section 2.6(b)</u> are either

- (i) deemed to be final and binding as provided in the immediately preceding sentence, or (ii) finally determined by the independent public accounting firm as provided in this <u>Section 2.6(b)</u>, is referred to herein as the "*Final Purchase Price Determination Date*."
- (c) If the Final Purchase Price is greater than the Closing Purchase Price, then Buyer shall pay such difference to Seller within five Business Days after the Final Purchase Price Determination Date, by wire transfer of immediately available funds to an account or accounts of Seller designated by Seller. If the Final Purchase Price is less than the Closing Purchase Price, then Seller shall pay such difference to Buyer within five Business Days after the Final Purchase Price Determination Date, by wire transfer of immediately available funds to an account or accounts of Buyer designated by Buyer.
- 2.7 *Tax Withholding*. Notwithstanding any other provision of this Agreement, Buyer shall be entitled to deduct and withhold from the cash otherwise deliverable under this Agreement, and from any other consideration otherwise paid or delivered in connection with the transactions contemplated by this Agreement, to any Person such amounts Buyer is required to deduct and withhold with respect to any such deliveries and payments under the Code or any provision of state, local or foreign Law; *provided*, that Buyer shall give written notice to Seller at least five (5) days prior to any such withholding. Buyer agrees to use its commercially

reasonable efforts to provide Seller the opportunity to furnish any applicable certificates or exemptions to reduce or eliminate any such withholding. To the extent such amounts are so withheld by Buyer, such withheld amounts shall be treated for purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made, and Buyer shall disburse such withheld amounts to the applicable Governmental Authority.

2.8 Express Exclusion of Operations Services from the Purchase and Sale. The Companies utilize, and the Business is currently dependent upon, certain operations, scheduling and distribution, accounting, contract administration, information technology, environmental, health and safety, land management and administration support services and software, including, but not limited to, gas and liquids monitoring and leak detection programs, processes and programs for gas and liquids measurement, allocation, settlement and distribution of monthly statements to producers, collection and disbursement of applicable Taxes, supervisory control and data acquisition, and various management systems for maintaining and monitoring compliance with applicable regulatory requirements and contractual obligations (collectively, the "Operations Services") that have been implemented in connection with Summit Parent's and its Affiliates' enterprise-wide management program for midstream operations. The Operations Services are proprietary, and non-transferable; therefore, it is expressly agreed and acknowledged that Buyer is not receiving the Operations Services in connection with this Agreement and Buyer will rely solely upon its own enterprise-wide management program for operations, accounting, contract administration, information technology, environmental, health and safety, land management and administration support services and software. Notwithstanding the foregoing, Buyer shall receive all the tangible hardware, communications, networking and measurement instrumentation and equipment associated with the Business operations conducted in the field.

ARTICLE III REPRESENTATIONS AND WARRANTIES REGARDING THE COMPANIES

Seller hereby represents and warrants (subject to any disclosures in the Schedules) to Buyer as of the date hereof, and if the Closing occurs, as of the Closing (or, if a representation and warranty speaks to an earlier date, then made at the Closing as of such earlier date), as follows:

3.1 Organization; Good Standing; Authority.

- (a) Each Company is a limited liability company duly formed, validly existing and in good standing under the Laws of the State of Delaware, and has all requisite limited liability company power and authority to own its properties and conduct the Business as it is now being conducted. The Company is duly qualified and licensed to do business in New Mexico, which is the only jurisdiction in which the ownership or operation of its Assets makes such qualification or licensing necessary.
- (b) True and complete copies of the Charter Documents of each Company and all amendments thereto have been furnished to Buyer.
- (c) Each Company has the requisite power and authority to execute and deliver the Transaction Documents to which it is, or will be, a party, to perform its obligations thereunder and to consummate the transactions contemplated thereby. The execution and delivery by each Company of the Transaction Documents to which it is, or will be, a party, and the performance by such Company of its obligations thereunder, have been duly and validly authorized by all necessary limited liability company action. The Transaction Documents to which each Company is, or will

be, a party have been (or will be, when executed and delivered at the Closing) duly and validly executed and delivered by such Company and will constitute (when so executed and delivered) the legal, valid and binding obligation of such Company enforceable against such Company in accordance with its terms and conditions, except that the enforcement hereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, arrangement or other similar Laws relating to or affecting the rights of creditors generally, or by general equitable principles.

3.2 Capitalization of Each Company.

- (a) The current equity capitalization of each Company is set forth in <u>Schedule 3.2(a)</u>.
- (b) The Interests in the Companies are uncertificated.
- (c) There are no interests of any Company convertible into, or exchangeable or exercisable for Equity Interests of such Company, or options, warrants, calls, rights, commitments or Contracts to which such Company is a party or by which it is bound, in any case obligating such Company to issue, deliver, sell, purchase, redeem or acquire, or cause to be issued, delivered, sold, purchased, redeemed or acquired, Equity Interests of such Company, or obligating such Company to grant, extend or enter into any such option, warrant, call, right, commitment or Contract.

- (d) The Interests are duly authorized, validly issued, fully paid (to the extent required by the Charter Documents of the Companies) and, subject to the Laws of the State of Delaware, non-assessable (except as such non-assessability may be affected by Section 18-607 of the Delaware Limited Liability Company Act) and were not issued in violation of any purchase option, call option, right of first refusal, preemptive right or other similar right.
- (e) There are no outstanding bonds, debentures, notes or other instruments or evidence of Indebtedness of any Company having the right to vote (or convertible into, or exercisable or exchangeable for, securities having the right to vote).
 - (f) Each Company owns the Equity Interests in the Persons set forth in Schedule 3.2(f).
- (g) Except as set forth in the Charter Documents, the Interests in the Companies are not subject to any voting agreement, voting trust, or any other agreement, right, instrument, or understanding with respect to any purchase, sale, transfer, repurchase, issuance, redemption, voting, distribution rights or disposition of such Interests.
- (h) When the Interests are transferred to Buyer from Seller at Closing, the Interests shall be owned by Buyer free and clear of all Liens, other than those under state or federal securities Laws or Liens arising by, through or under Buyer or any of its Affiliates.
- 3.3 *No Conflicts; Consents and Approvals.* The execution and delivery by Seller of this Agreement, and the performance by Seller of its obligations under this Agreement do not and will not:

- (a) violate or result in a breach of the Charter Documents of any Company;
- (b) violate or result in a default, or give rise to any right of termination, cancellation, amendment or acceleration under any Material Contract to which any Company is a party;
 - (c) violate or result in a breach of any Law applicable to any Company;
 - (d) result in the creation of any Lien on any of the Assets of the Companies, other than Permitted Liens; or
- (e) require any consent or approval of (i) any Governmental Authority under any Law applicable to Seller and (ii) any counterparty to any Contract to which the Company is a party or by which any of its Assets or properties are bound, except as set forth on <u>Schedule 3.3</u>.
- 3.4 *Financial Statements*. Buyer has been provided copies of the unaudited balance sheet of Summit Permian as of December 31, 2020 and 2021 and the unaudited income statement of Summit Permian for the years ended December 31, 2020 and 2021 and the three months ended March 31, 2022 (the "*Financial Statements*"). The Financial Statements have been prepared in accordance with GAAP (except that the Financial Statements do not contain the financial statement footnotes required by GAAP and are subject to normal year-end adjustments) applied on a consistent basis throughout the periods covered thereby and fairly present, in all material respects, the financial position of the Companies at the dates thereof.
 - 3.5 **Absence of Changes.** Except as set forth in Schedule 3.5, since March 31, 2022:
- (a) no fact, event, change, occurrence or circumstance has occurred that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect;
 - (b) no Company has amended its Charter Documents;
- (c) no Company has sold, transferred or disposed of any of its Assets, including any right under any lease or Contract or any proprietary right or other intangible Asset, (i) within the ordinary course of business in each case having a value in excess of \$250,000 or (ii) outside of the ordinary course of business;
- (d) no Company has waived, released, canceled, settled or compromised any debt, Claim or right having a value in excess of \$100,000:
- (e) except as may be required to meet the requirements of applicable Law or GAAP, no Company has changed any accounting method or practice in a manner that is inconsistent with past practice in a way that would materially and adversely affect the Business and such Company;
- (f) no Company has failed to maintain its limited liability company existence or consolidated with any other Person or acquired all or substantially all of the Assets of any other Person;

- (g) no Company has issued or sold any Equity Interests in itself;
- (h) no Company has liquidated, dissolved, recapitalized, reorganized or otherwise wound up the Business;
- (i) no Company has purchased any securities of any Person, except for short- term investments made in the ordinary course of business; and
 - (j) no Company has agreed or committed to do any of the foregoing.
- 3.6 *Compliance with Applicable Laws*. Except as set forth on <u>Schedule 3.6</u>, and for the prior three (3) year period, each Company is, or has been, in material compliance with all Laws (and has not received any written notice of violation from any Governmental Authority or any other Person with respect to any Laws) applicable to it, such Company holds all material Permits necessary for the lawful conduct of the Business and operation and ownership of the Assets and, to the Knowledge of Seller, a complete list of all such Permits is set forth on <u>Schedule 3.6-P</u>; *provided, however*, that this <u>Section 3.6</u> does not address compliance with Environmental Laws, which is exclusively addressed by <u>Section 3.12</u>, matters relating to Taxes, which are exclusively addressed by <u>Section 3.13</u> and <u>Section 3.16</u> or employee matters, which are exclusively addressed by <u>Section 3.15</u>.

3.7 Intellectual Property.

- (a) No material registrations or applications for registration are included in any Intellectual Property Rights held by any Company. Subject to Section 2.8, each Company owns, licenses or otherwise has a valid right to use, free and clear of all Liens (other than Permitted Liens), all material Intellectual Property Rights necessary to conduct the Business as currently conducted.
- (b) To the Knowledge of Seller, the conduct of the Business as currently conducted has not infringed or misappropriated any Intellectual Property Right of any third party in any material respect.
- (c) The consummation of the transactions contemplated hereby will not result in the loss or impairment of any material right of any Company to own, use, practice or exploit any Intellectual Property Rights held by or licensed to such Company (excluding licenses for commercially available, "off-the-shelf" software).
- 3.8 **Absence of Litigation**. Except as set forth on <u>Schedule 3.8</u>, there is no Claim or Proceeding (a) pending against any Company by or before any arbitrator or Governmental Authority, or (b) to the Knowledge of Seller, threatened against such Company by or before any arbitrator or Governmental Authority.

3.9 Real Property.

(a) Set forth on <u>Schedule 3.9(a)</u> is a true and complete list of each parcel of Real Property owned in fee title by each Company and its subsidiaries. Seller has provided Buyer with true and complete copies of the conveyance documents to any Company for each such parcel

of Real Property owned in fee. As of the date hereof, except with respect to the IRB Assets, each Company or its subsidiaries has good and indefeasible fee title to all applicable Real Property listed on <u>Schedule 3.9(a)</u>, free and clear of all Liens, except for Permitted Liens and those Liens set forth on <u>Schedule 3.9(a)</u>. As of the Closing Date, each Company or its subsidiaries will have good and indefeasible fee title to all applicable Real Property listed on <u>Schedule 3.9(a)</u>, including the IRB Assets, free and clear of all Liens, except for Permitted Liens and those Liens set forth on <u>Schedule 3.9(a)</u>.

- (b) Set forth on Schedule 3.9(b) is a true and complete list of all leases pursuant to which each Company or its subsidiaries is granted a right to use or occupy all or any portion of Real Property. Seller has provided Buyer with true and complete copies of such leases, and any amendments thereto. Each lease set forth on Schedule 3.9(b) is a legal, valid and binding obligation of the applicable Company or its subsidiaries. Except as set forth on Schedule 3.9(b), (i) none of the Companies or their subsidiaries is in material default under any lease set forth on Schedule 3.9(b), (ii) to the Knowledge of Seller, no landlord is in material default under any lease set forth on Schedule 3.9(b), and (iii) no event has occurred which constitutes a material default or, with lapse of time or giving of notice or both, would constitute a material default under any of the leases set forth on Schedule 3.9(b).
- (c) Set forth on <u>Schedule 3.9(c)</u> are all easements (including amendments thereto) on which any portion of the Summit Permian System is located, except in such case that any omission would not be material to the Business. Seller has provided Buyer with true and complete copies of the documents creating such easements, and any amendments thereto. Each easement set forth on <u>Schedule 3.9(c)</u> is a legal, valid and binding obligation of the applicable Company or its subsidiaries. Except as set forth on <u>Schedule 3.9(c)</u>, (i) none of the Companies or their subsidiaries is in material default under any easement set forth on <u>Schedule 3.9(c)</u>, (ii) to the Knowledge of Seller, no owner of the Real Property granting any easement set forth on <u>Schedule 3.9(c)</u> is in material default under any such easement, and (iii) no event has occurred which constitutes a material default or, with lapse of time or giving of notice or both, would constitute a material default under any of the easements set forth on <u>Schedule 3.9(c)</u>.
- (d) No Company is obligated under any right of first refusal, option or other contractual right to sell, assign or otherwise dispose of any Real Property, or any interest therein, to any Person other than Buyer.
- (e) Except as set forth on <u>Schedule 3.9(e)</u>, the easements granted to Summit Permian and its subsidiaries that cover the Summit Permian System, together with any owned or leased Real Property related to the Summit Permian System, establish a continuous right of way for the Summit Permian System, and the Summit Permian System and the buildings and improvements used in connection therewith are located entirely on Real Property of the Companies and their subsidiaries except in each case as would not reasonably be expected to be material to the Business.
- (f) No Company has received any written notice of any eminent domain Proceeding or taking, nor, to the Knowledge of Seller, is any such Proceeding or taking contemplated with respect to all or any material portion of the Real Property.

- (g) Except as set forth on <u>Schedule 3.9(g)</u>, no member of the Seller Group other than the Companies and their subsidiaries owns or leases any real property or easements that constitute part of the Business or that is necessary in connection with the ownership or operation thereof.
- (h) Other than the assets associated with the Operations Services, the Assets of the Companies constitute all of the assets, rights and properties, tangible or intangible, real or personal, that are used or necessary for use in connection with the operation of the Business as currently operated.
- (i) None of the assets retained by any member of the Seller Group (other than the Companies) as part of the Operations Services (other than the vehicles) are physically located on any Real Property, Personal Property, easements set forth on $\underline{Schedule 3.9(c)}$, or the Summit Permian System.
- 3.10 **Personal Property**. Schedule 3.10 lists all plants, processing units and other equipment (other than the individual components of any plants or processing units or installed pipeline) owned or leased by each Company and its subsidiaries or used or held for use by such Company and its subsidiaries in the conduct of the Business valued above \$100,000 (if required to be listed on Schedule 3.10, the "**Personal Property**"). Each Company and its subsidiaries have good and valid title to all Personal Property free of all Liens except for Permitted Liens.

3.11 Regulatory Status.

- (a) None of the Assets of the Companies is currently subject to regulation by FERC under the NGA and (ii) as of the date of this Agreement, the Companies are in material compliance with the applicable laws, regulations and orders of the New Mexico Public Regulation Commission.
- (b) (i) None of the Summit Permian System is currently regulated, and (ii) no portion of the Summit Permian System would be regulated as of the date hereof, by FERC as a "natural gas company" under the NGA. Without limiting the foregoing, the rates charged by the Companies are not currently regulated by FERC under the Interstate Commerce Act or the Natural Gas Policy Act of 1978. The transfer of the Interests as contemplated by this Agreement does not require the approval of FERC.
- (c) No rate refunds, rebates, offsets or like obligations are accrued or owed by any Company to FERC, the New Mexico Public Regulation Commission or any other Governmental Authority or third party with respect to services related to the Assets of any Company.
- (d) Each Company has timely made all material regulatory filings necessary to conduct its Business, and has provided Buyer with copies of all such filings made in the past three years.
 - 3.12 *Environmental Matters*. Except as set forth in <u>Schedule 3.12</u>:

- (a) Each Company is and the Assets of each Company are, and since January 1, 2019, has and have been, in compliance in all material respects with applicable Environmental Laws, including by timely possessing and complying with the terms and conditions of, and, to the extent applicable, filing timely application to renew, all Environmental Permits, and, to the Knowledge of Seller, a complete list of such Environmental Permits is set forth on Schedule 3.12(a)-P;
- (b) No Company has received from any Governmental Authority or any other Person any written notice alleging a material violation of, non-compliance with, or liability or potential or alleged liability pursuant to, any Environmental Laws other than notices with respect to matters that have been resolved to the satisfaction of any relevant Governmental Authority and for which such Company has no further material obligations outstanding;
- (c) None of the Companies or the Assets of the Companies or the Summit Permian System is subject to any outstanding order, consent decree or other agreement concerning alleged non-compliance with or liability under Environmental Laws;
- (d) There has been no Release of Hazardous Materials by any Company at, under, on or from any Real Property currently or, to the Knowledge of the Seller, formerly owned, leased, occupied or operated by any Company or any other location, and, to the Knowledge of the Seller, no Hazardous Materials are present at, under or on any Real Property currently or formerly owned, leased, occupied or operated by any Company, in each case, in violation of any Environmental Laws or in a quantity, concentration or manner that could reasonably be expected to give rise to a material liability, including any remedial or corrective action obligation pursuant to Environmental Laws.
- (e) The Seller has provided to Buyer copies of final reports of all environmental audits, assessments, investigations, studies or other analysis relating to the Companies or the Companies' Assets that are in the possession of the Seller, any Company, or any of their respective Affiliates or Representatives.

3.13 *Taxes.* Except, in each case, as set forth on <u>Schedule 3.13</u>:

- (a) All material Tax Returns required to be filed by each Company and Seller with respect to the Companies have been duly and timely filed (taking into account any applicable extensions). Each such Tax Return is true, correct and complete in all material respects. All material Taxes required to be paid by each Company and Seller with respect to the Companies (whether or not shown due on a filed Tax Return) have been paid in full. All withholding Tax requirements imposed on each Company and Seller with respect to the Companies have been satisfied in all material respects. There are no Liens (other than Permitted Liens) on any of the Assets of any Company or Seller's membership interest in Summit Permian that arose in connection with any failure (or alleged failure) to pay any Tax.
- (b) No assessment, deficiency, or adjustment has been asserted, proposed, or threatened in writing with respect to any Taxes or Tax Returns of any Company or Seller with respect to the Companies, in each case, that remains outstanding.

- (c) Neither Seller, with respect to the Taxes owed by or with respect to the Companies nor any Company is a party to or bound by any Tax allocation, sharing or indemnity agreements or arrangements (excluding, for the avoidance of doubt, any commercial agreements or contracts that are not primarily related to Taxes). Neither Seller nor any Company is liable for the Taxes of any other Person (other than the Companies) by virtue of Treasury Regulation Section 1.1502-6 or any similar provision of state, local or foreign applicable Law (other than the members of any combined group of which such Company, Seller or an Affiliate of Seller is the primary filer for Texas franchise tax purposes).
- (d) Each Company is classified as an entity disregarded as separate from Summit Parent for U.S. federal income Tax purposes.
- (e) Each of Seller (with respect to the Companies) and each Company has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, shareholder, member or creditor of such entity, or other third party, and all forms (including forms W-2 and 1099) required with respect thereto have been properly completed and timely filed.
- (f) No Claim has ever been made by a Governmental Authority in a jurisdiction where any of Seller with respect to the Companies or the Companies do not file Tax Returns that Seller with respect to the Companies or any Company is or may be subject to taxation by that jurisdiction or any of Seller's or the Companies' assets are or may be subject to such taxation.
- (g) Neither Seller with respect to the Companies nor any Company has been a party to a transaction that is or is substantially similar to a "reportable transaction," within the meaning of Treasury Regulation Section 1.6011-4(b), or any other transaction requiring disclosure under analogous provisions of the United States, state, local or foreign Tax Law.
 - (h) Seller is not a foreign person within the meaning of Section 1445 of the Code.

3.14 *Contracts*.

- (a) <u>Schedule 3.14</u> contains a true and complete listing of each of the following Contracts to which any Company is a party (the Contracts listed in <u>Schedule 3.14</u> being "*Material Contracts*"):
 - (i) each hydrocarbon purchase and sale, gathering, transportation, treating, dehydration, processing or similar Contract and any Contract for the provision of services relating to gathering, compression, collection, processing, treating, disposal or transportation of natural gas, other hydrocarbons and saltwater involving annual expenditures or revenues in excess of \$100,000;
 - (ii) each Contract that constitutes a pipeline interconnect or facility operating agreement;
 - (iii) each Contract involving a remaining commitment to pay capital expenditures in excess of \$100,000 in the aggregate;

- (iv) each Contract for lease of personal property or Real Property involving aggregate payments in excess of \$100,000 in any future calendar year;
 - (v) each partnership or joint venture agreement;
- (vi) each agreement relating to the acquisition or disposition of any business (whether by merger, sale of stock, sale of Assets or otherwise) or granting to any Person a right of first refusal, first offer or right to purchase any of the Assets material to the conduct of the Business;
- (vii) each Contract that provides for a limit on the ability of any Company to compete in any line of business or in any geographic area during any period of time after the Closing;
- (viii) each Contract evidencing Indebtedness, whether secured or unsecured, including all loan agreements, line of credit agreements, indentures, mortgages, promissory notes, guarantees, agreements concerning long and short-term debt, together with all security agreements or other lien documents related to or binding on the Assets of any Company or requires the owner of the Assets to conduct business exclusively with any Person, including any Contract containing an area of mutual interest;
- (ix) any contract that licenses material Intellectual Property Rights from a third party, other than "off the shelf" software licenses that are generally commercially available;
 - (x) any swap, option, hedge or future Contracts;
- (xi) each Contract that provides for the indemnification by any Company of any Person outside of the ordinary course of business, which ordinary course of business arrangements shall include customary indemnities in services agreements, customer agreements or similar agreements;
- (xii) any contracts between the Seller and its Affiliates, on the one hand, and any Company, on the other hand;
 - (xiii) any Contracts with employees or officers of any Company; and
- (xiv) except for Contracts of the nature described in <u>clauses (i)</u> through (<u>iv</u>) above, each Contract involving aggregate payments or receipts in excess of \$100,000 in any future calendar year that cannot be terminated by such Person or any Company, as applicable, upon 30 days or less notice without payment of a penalty or other liability.
 - (b) True and complete copies of all Material Contracts have been made available to Buyer.
- (c) Each of the Material Contracts is in full force and effect in all material respects and constitutes a legal, valid and binding obligation of the applicable Company and, to the Knowledge of Seller, of the counterparties to such Material Contracts. No Company nor, to the

Knowledge of Seller, any counterparty thereto, is in (or has received written notice that it is in) default or breach (or has taken or failed to take any action such that with notice, the passage of time or both it would be in default or breach) under the terms of any such Material Contract except any such defaults or breaches that would not reasonably be expected to be material to any Company or the Business.

- 3.15 *Employees*. No Company has, or has ever had, any employees on its payroll. The Company has never had any agreement with any individual providing services as an independent contractor or as a consultant.
- 3.16 *Employee Benefit Plans*. No Company currently or has ever maintained or contributed to any Plan or been a participating employer in any Plan, and no Company has any current or contingent liability or obligation under or with respect to any Plan. There does not now exist, nor do any circumstances exist that could reasonably be expected to result in, any liabilities of any Company under (i) Title IV of ERISA, (ii) Sections 206(g), 302 or 303 of ERISA, or (iii) Sections 412, 430, 431, 436 or 4971 of the Code.
- 3.17 *Transactions with Affiliates*. Except as set forth on <u>Schedule 3.17</u>, no Company is owed any amount from, and does not owe any amount to, guarantee any amount owed by, have any Contracts with or have any commitments to any Affiliate, any officer or director of Seller, such Company or any member of a family group of any of the foregoing.
- 3.18 **Surety Bonds and Credit.** Except as listed on <u>Schedule 3.18</u>, none of Seller or the Companies has any obligation to post any material surety bond, letter of credit, guarantee or other form of support (credit or otherwise) in respect of the Companies or their Business.
- 3.19 *No Bankruptcy*. There are no bankruptcy Proceedings pending against or, to the Knowledge of Seller, threatened against any Company.
- 3.20 *No Undisclosed Material Liabilities; Indebtedness.* Except as set forth on Schedule 3.20, there are no liabilities of any Company of any kind (whether accrued or fixed, absolute or contingent, matured or unmatured, determined or determinable or otherwise), other than: (a) liabilities disclosed, reflected, reserved against or otherwise provided for in the Financial Statements or disclosed in the notes thereto; (b) liabilities not required under GAAP to be disclosed, reflected, reserved against or otherwise provided for in the Financial Statements or disclosed in the notes thereto; and (c) liabilities, other than Indebtedness, incurred in the ordinary course of business since March 31, 2022. Except as set forth in Schedule 3.20, no Company has any Indebtedness.
- 3.21 *Insurance*. Schedule 3.21 sets forth a list of each insurance policy maintained by any Company or by Seller or any of its Affiliates on behalf of such Company. Except as set forth in Schedule 3.21, there is no material claim pending under any such insurance policy with respect to the Business or the Assets of any Company, and none of Seller, such Company or any of their respective Affiliates has received written notice of cancellation of any such insurance policy. Each policy listed on Schedule 3.21 is valid and binding and in full force and effect, and all premiums under such policies that are due and payable on or prior to the date of this Agreement have been paid in full. Except as would not individually or in the aggregate reasonably be expected to be

material to the Business, operations or financial performance of the Companies, there is no default under any such policy and no insurer has advised Seller or the Companies in writing that it intends to reduce coverage, increase premiums or fail to renew any existing policy or binder. Neither the Companies nor any of their Affiliates have received any written notice from any insurer of any reservation of rights with respect to any pending or paid claims.

- 3.22 *Bank Accounts.* Set forth on <u>Schedule 3.22</u> is an accurate and complete list showing (a) the name and address of each bank in which any Company has an account or safe deposit box, the number of any such account or any such box and the names of all Persons authorized to draw thereon or to have access thereto and (b) the names of all Persons, if any, holding powers of attorney from such Company and a summary statement of the terms thereof.
- 3.23 *Throughput Data*. Schedule 3.23 sets forth historical throughput data and information for the calendar year 2020 and 2021 and for the year to date to March 31, 2022, relating to the Business. Such throughput data and information is accurate and complete in all material respects with respect to the information for each applicable period, respectively, without representations as to any specific monthly volume. Neither Company has received any written, or to Knowledge of Seller, oral, notices regarding a material reduction or increase of throughput from any customers.
- 3.24 *Capital Commitments; Capital Expenditures*. Except for each Company's obligations set forth on <u>Schedule 3.24</u>, no material capital expenditures are required to enable any Company to operate the Summit Permian System and the Business in a manner consistent with past practice and as currently operated. Except as set forth on <u>Schedule 3.24</u>, no Company has made any capital expenditures in excess of \$100,000 since April 30, 2022 and no Company has any obligations for capital commitments or expenditures in excess of \$100,000 as of the date hereof.
- 3.25 *Imbalances*. As of the date hereof, except for the hydrocarbon imbalances reflected on <u>Schedule 3.25</u> (for which Buyer and the Companies shall not be liable) and for normal and customary gathering imbalances occurring after the date hereof, there do not exist any material hydrocarbon imbalances as of the date hereof (a) under any Material Contracts or (b) for which any Company has received a quantity of hydrocarbons prior to the date hereof for which such Company will have a duty to deliver an equivalent quantity of hydrocarbons after the Closing.
- 3.26 *Preferential Rights*. No Person has a preferential right to purchase with respect to any of the Assets owned by, or Equity Interests in, any Company as a result of the consummation of the transactions contemplated by this Agreement and the other Transaction Documents.

3.27 Data Privacy and Security.

(a) To the Knowledge of Seller, no third party has gained unauthorized access to or made any unauthorized use of any information maintained by (or by third parties on behalf of) the Companies in a manner that constitutes a breach under any applicable Privacy and Security Law, if any, except unsuccessful security incidents (such as, but not limited to, pings, phishing exercises or other broadcast attacks on the Companies' firewall, port scan, unsuccessful log-on attempts, denials of service or any combination of same).

- (b) To the Knowledge of Seller, there are no facts or other information indicating that there have occurred any unauthorized intrusions or breaches of security of the Companies' computer systems or other information technology used or provided by the Companies. The Companies have adopted and use commercially reasonable security procedures and systems to identify attempted or successful intrusions or breaches of security of the Companies' systems or other information technology used by any Company.
- (c) The Companies currently are and have been in compliance in all material respects with applicable Privacy and Security Laws and Data Security Requirements.

3.28 Accounts Receivable; Payable and Purchase Orders.

- (a) All of the accounts receivable owing to the Companies ("Accounts Receivable") constitute valid and enforceable claims arising from bona fide transactions for goods sold or services performed in the ordinary course of business, and, to the Knowledge of Sellers, the goods and/or services involved have been sold, delivered and/or performed to the account obligors. The Companies have not received any written (or, to the Knowledge of Sellers, oral) notice of any claims, refusals to pay or other claimed rights of set off against any Accounts Receivables other than in the ordinary course of business and consistent with past practices. To the Knowledge of Seller, no account debtor is insolvent or bankrupt.
- (b) All accounts payable of the Company are legal, valid and binding obligations of the Companies and were incurred in the ordinary course of business.
- (c) <u>Schedule 3.28</u> sets for a true and complete list as of the date hereof of all open orders for the purchase of goods or services by the Companies in an amount in excess of \$100,000 relating to the Business.
- 3.29 *Condition and Sufficiency of the Assets*. All Assets used in the Business of the Companies are in good operating condition and repair (subject to ordinary wear and tear), are adequate for the uses to which they are currently being put, and are sufficient for the operation of the Business as currently conducted in all material respects. None of the Assets are in need of material maintenance or repairs except for ordinary, routine maintenance and repairs that are not material in nature or cost.

ARTICLE IV REPRESENTATIONS AND WARRANTIES REGARDING SELLER

Seller hereby represents and warrants (subject to any disclosures in the Schedules) to Buyer as of the date hereof, and if the Closing occurs, as of the Closing (or, if a representation and warranty speaks to an earlier date, then made at the Closing as of such earlier date), as follows:

- 4.1 *Organization; Good Standing*. Seller is duly formed, validly existing and in good standing under the Laws of its jurisdiction of formation.
- 4.2 **Authority**. Seller has the requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution, delivery and performance by Seller of this Agreement, and the performance

by Seller of its obligations hereunder, have been duly and validly authorized by all necessary limited liability company action. This Agreement has been duly and validly executed and delivered by Seller and constitutes the legal, valid and binding obligation of Seller enforceable against Seller in accordance with its terms and conditions, except that the enforcement hereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, arrangement or other similar Laws relating to or affecting the rights of creditors generally, or by general equitable principles.

- 4.3 *Ownership of the Interests*. The Interests are owned beneficially and of record by Seller, free and clear of all Liens other than those under state or federal securities Laws or the Charter Documents of the Companies.
- 4.4 *No Conflicts; Consents and Approvals*. Except as set forth in <u>Schedule 4.4</u>, the execution and delivery by Seller of this Agreement, and the performance by Seller of its obligations under this Agreement, do not and will not:
 - (a) violate or result in a breach of the Charter Documents of Seller;
- (b) violate or result in a default (or any event that with notice or passage of time or both would give rise to a default) under, or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both) under any material Contract to which Seller is a party, except for any such violation or default which would not reasonably be expected to result in a material impairment on Seller's ability to perform its obligations hereunder;
- (c) violate or result in a breach of any Law applicable to Seller, except for such violations or breaches as would not be material; or
- (d) require any consent or approval of any Governmental Authority under any Law applicable to Seller, other than immaterial consents or approvals.
- 4.5 **Legal Proceedings**. There is no Proceeding pending or to the Knowledge of Seller, threatened against Seller by or before any arbitrator or Governmental Authority, which seeks a writ, judgment, order or decree restraining, enjoining or otherwise rendering illegal any of the transactions contemplated by this Agreement.
- 4.6 *Broker's Commissions*. None of Seller or the Companies has, directly or indirectly, entered into any Contract with any Person that would obligate Buyer or any of its Affiliates (including the Companies after Closing) to pay any commission, brokerage fee or "finder's fee" in connection with the transactions contemplated herein.
- 4.7 *No Bankruptcy*. There are no bankruptcy Proceedings pending against, being contemplated by or, to the Knowledge of Seller, threatened against Seller or its Affiliates.
- 4.8 *Financial Resources*. Seller has sufficient cash on hand, available lines of credit or other sources of immediately available funds to enable it to make any post-Closing payments which may be required under this Agreement and to otherwise perform its obligations under this Agreement.

ARTICLE V REPRESENTATIONS AND WARRANTIES REGARDING BUYER

Buyer hereby represents and warrants to Seller as of the date hereof and, if the Closing occurs, at the Closing shall be deemed to have represented and warranted (subject to any disclosures in the Schedules) to Seller as of the Closing (or, if a representation and warranty speaks to an earlier date, then made at the Closing as of such earlier date), as follows:

- 5.1 *Organization; Good Standing*. Buyer is duly formed, validly existing and in good standing under the Laws of its jurisdiction of formation.
- 5.2 **Authority**. Buyer has all requisite power and authority to enter into this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery by Buyer of this Agreement, and the performance by Buyer of its obligations hereunder, have been duly and validly authorized by all necessary partnership action on behalf of Buyer. This Agreement has been duly and validly executed and delivered by Buyer and constitutes the legal, valid and binding obligation of Buyer enforceable against Buyer in accordance with its terms except as the same may be limited by bankruptcy, insolvency, reorganization, fraudulent conveyance, arrangement, moratorium or other similar Laws relating to or affecting the rights of creditors generally or by general equitable principles.
- 5.3 *No Conflicts; Consents and Approvals*. The execution and delivery by Buyer of this Agreement does not, and the performance by Buyer of its obligations under this Agreement does not and will not:
 - (a) violate or result in a breach of its Charter Documents;
- (b) violate or result in a default (or any event that with notice or passage of time or both would give rise to a default) under, or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both)under any material Contract to which Buyer is a party, except for any such violation or default which would not reasonably be expected to result in a material impairment on Buyer's ability to perform its obligations hereunder:
- (c) violate or result in a breach of any Law applicable to Buyer, except for such violations or breaches as would not be material; or
- (d) require any consent or approval of any Governmental Authority under any Law applicable to Buyer, other than immaterial consents or approvals.
- 5.4 *Legal Proceedings*. There is no Proceeding pending or, to Buyer's knowledge, threatened, against Buyer before or by any Governmental Authority, which seeks a writ, judgment, order or decree restraining, enjoining or otherwise prohibiting or making illegal any of the transactions contemplated by this Agreement.
- 5.5 **Acquisition as Investment**. Buyer is acquiring the Interests for its own account as an investment without the present intent to sell, transfer or otherwise distribute the same to any other Person in violation of any state or federal securities laws. Buyer has made, independently

and without reliance on Seller, its own analysis of the Interests, the Companies and their Assets for the purpose of acquiring the Interests, and Buyer has had reasonable and sufficient access to documents, other information and materials as it considers appropriate to make its evaluations. Buyer acknowledges that the Interests are not registered pursuant to the 1933 Act and that none of the Interests may be transferred, except pursuant to an effective registration statement or an applicable exemption from registration under the 1933 Act. Buyer is an "accredited investor" as defined under Rule 501 promulgated under the 1933 Act.

- 5.6 *Financial Resources*. Buyer has and will have on the Closing Date, sufficient cash on hand, available lines of credit or other sources of immediately available funds to enable it (a) to pay the Purchase Price and (b) to otherwise perform its obligations under this Agreement.
- 5.7 **Opportunity for Independent Investigation**. Buyer is an experienced and knowledgeable investor in the United States. Buyer has conducted its own independent review and analysis of the Business and of the Assets, liabilities, results of operations, financial condition, technology and prospects of the Companies and acknowledges that Buyer has been provided access to personnel, properties, premises and records of the Companies for such purpose. In entering into this Agreement, Buyer has relied solely upon the representations, warranties and covenants contained herein and upon its own investigation and analysis of the Companies and the Business (such investigation and analysis having been performed by Buyer), and Buyer:
- (a) absent Actual Fraud, acknowledges and agrees that it has not been induced by and has not relied upon any Due Diligence Information, representations, warranties or statements, whether oral or written, express or implied, made by Seller, the Companies or any of their Representatives, Affiliates or agents except for the representations and warranties expressly set forth in this Agreement and those items delivered to Buyer pursuant to Section 2.4 and Section 7.4;
- (b) acknowledges and agrees that, except for the representations and warranties expressly set forth in this Agreement and those items delivered to Buyer pursuant to Section 2.4 and Section 7.4, none of Seller, the Companies or any of their Representatives, Affiliates or agents makes or has made any representation or warranty, either express or implied, as to the accuracy or completeness of any of the information provided or made available to Buyer or its Representatives, Affiliates or agents, including any information, document or material provided or made available, or statements made, to Buyer (including its Representatives, Affiliates and agents) during site or office visits, in any "data rooms," management presentations or supplemental due diligence information provided to Buyer (including its Representatives, Affiliates and agents), in connection with discussions with management or in any other form in expectation of the transactions contemplated by this Agreement (collectively, the "**Due Diligence Information**");
- (c) acknowledges and agrees that, except for the representations and warranties expressly set forth in this Agreement and those items delivered to Buyer pursuant to <u>Section 2.4</u> and <u>Section 7.4</u>, (i) the Due Diligence Information includes certain projections, estimates and other forecasts and certain business plan information, (ii) there are uncertainties inherent in attempting to make such projections, estimates and other forecasts and plans and Buyer is aware of such uncertainties, and (iii) Buyer is taking full responsibility for making its own evaluation of the adequacy and accuracy of all projections, estimates and other forecasts and plans so furnished to

it and any use of or reliance by Buyer on such projections, estimates and other forecasts and plans shall be at its sole risk; and

- (d) agrees, to the fullest extent permitted by Law, that none of Seller, the Companies or any of their respective Representatives, Affiliates or agents shall have any liability or responsibility whatsoever to Buyer or its Representatives, Affiliates or agents on any basis (including in contract or tort, under federal or state securities Laws or otherwise) resulting from the furnishing to Buyer, or from Buyer's use of, any Due Diligence Information, except for liability or responsibility for the representations and warranties expressly set forth in this Agreement and those items delivered to Buyer pursuant to Section 2.4 and Section 7.4.
- 5.8 *Broker's Commissions*. Buyer has not, directly or indirectly, entered into any Contract with any Person that would obligate Seller or any of its Affiliates to pay any commission, brokerage fee or "finder's fee" in connection with the transactions contemplated herein.
- 5.9 **Solvency**. Immediately after giving effect to the transactions contemplated by this Agreement, assuming the accuracy of the representations and warranties set forth in <u>Article III</u>, none of Buyer or the Companies, individually or in the aggregate, will (a) be insolvent (either because its financial condition is such that the sum of its debts is greater than the fair value of its assets or because the fair salable value of its assets is less than the amount required to pay its probable liability on its existing debts as they mature), (b) have unreasonably small capital with which to engage in its business or (c) have incurred debts beyond its ability to pay as they become due. In completing the transactions contemplated by this Agreement, Buyer does not intend to hinder, delay or defraud any present or future creditors of any of the Companies.

ARTICLE VI COVENANTS

6.1 Interim Period Operations.

- (a) Except as required or permitted hereby, or as consented to by Buyer (which consent shall not be unreasonably withheld, conditioned or delayed), or as otherwise set forth in <u>Schedule 6.1</u>, from the date of this Agreement until the earlier of the Closing or the termination of this Agreement in accordance with the terms hereof (the "*Interim Period*"), Seller will use commercially reasonable efforts to cause each Company (x) to operate in the ordinary course of business; and (y) to preserve intact its material relationships with third parties. Without limiting the foregoing, except as otherwise required or permitted hereby or required by the terms of any material Permit or any Material Contract, Seller will cause each Company not to undertake any of the following during the Interim Period without the consent of Buyer (which consent shall not be unreasonably withheld, conditioned or delayed):
 - (i) adopt any change in any of its Charter Documents;
 - (ii) (A) fail to maintain its limited liability company existence, or (B) consolidate with any other Person or acquire (by merger, consolidation, acquisition of stock or assets or otherwise) all or substantially all of the Assets of any other Person;

- (iii) sell, transfer or otherwise dispose of any material Assets of such Company, other than (A) sales of inventory in the ordinary course of business, (B) any abandonment of Intellectual Property Rights that such Company determines in the exercise of its reasonable business judgment to abandon in the ordinary course of business, (C)) replacement of Assets of such Company in the ordinary course of business and/or dispositions of obsolete or worthless assets, and (D) Permitted Liens;
 - (iv) split, combine or reclassify any Equity Interests in such Company;
- (v) transfer, issue, sell or otherwise dispose, or repurchase, redeem or otherwise acquire, any Equity Interests in such Company;
- (vi) liquidate, dissolve, recapitalize, reorganize or otherwise wind up all or any portion of the Business;
 - (vii) purchase any securities of any Person;
- (viii) other than in the ordinary course of business, terminate, renew, amend, restate, supplement or waive any material rights under any Material Contract involving total consideration throughout its term in excess of \$100,000;
- (ix) enter into any Contract that would have been a Material Contract had it been entered into prior to the date of this Agreement;
 - (x) make any loans, advances or capital contributions to, or investments in, any other Person;
- (xi) incur (or enter into any agreement for the incurrence of) any Indebtedness (except for Indebtedness entered into in the ordinary course of business);
- (xii) cause the Assets of the Companies or any Equity Interests in the Company to be encumbered by any Liens, except for Permitted Liens;
- (xiii) incur any capital expenditures or operating expenditures other than in the ordinary course of business consistent with past practice and not to exceed, individually or in the aggregate, \$100,000;
- (xiv) cancel or compromise any debt or claim or waive or release any material right of such Company that could be reasonably expected (due to the nature of the claims involved or the scope of their applicability to such Company's business or operations) to involve amounts of \$100,000 or more in value;
- (xv) settle or compromise any action, suit, investigation or proceeding against such Company that could be reasonably expected (due to the nature of the claims involved or the scope of their applicability to such Company's business or operations) to involve amounts of \$100,000 or more in value, except for any cases in which the amount paid in settlement does not exceed the amount reserved against such matter in the Financial Statements (or the notes thereto);

(xvi) enter into or modify in any material respect, terminate, cancel, renew or assign any Permit, or make any regulatory filing that is not either expressly contemplated by this Agreement or made in the ordinary course of business;

(xvii) except as may be required to meet the requirements of applicable Law or GAAP, change any accounting method or practice in a manner that is inconsistent with past practice in a way that would materially and adversely affect such Company; or

(xviii) agree or commit to do any of the foregoing.

Notwithstanding the foregoing, each Company may take commercially reasonable actions with respect to (i) emergency situations or (ii) requirements of Law, and each Company shall promptly inform Buyer of any such actions taken outside the ordinary course of business. For the avoidance of doubt, and notwithstanding anything in this Agreement to the contrary, nothing in this Agreement shall prohibit or otherwise restrict in any way the operation of the business of Summit Parent and its Affiliates (including any actions that may be outside the ordinary course of business of Summit Parent) so long as the Business is not materially and disproportionately affected by any such action as compared to the other businesses of Summit Parent.

6.2 Access of Buyer.

During the Interim Period, Seller will provide (or will cause each Company to provide) Buyer and its Representatives with reasonable access, upon reasonable prior notice and during normal business hours, to the Assets of such Company and the officers and employees of such Company (or, as appropriate, its Affiliates) who have significant responsibility for the Assets of such Company, but only to the extent that such access (i) does not unreasonably interfere with the Business or the safe commercial operations of the Assets of such Company (or, as appropriate, its Affiliates) and (ii) is reasonably related to Buyer's obligations and rights hereunder; provided, however, that (A) Seller shall have the right to have a Representative present for any communication with employees or officers of such Company or its Affiliates, (B) Buyer shall and shall cause its Representatives to observe and comply with all generally applicable health, safety and security requirements of such Company and (C) Buyer shall not conduct any environmental sampling or testing, except for the completion of a Phase I environmental review, or otherwise make direct contact identifying the Company with regulators of such Company, without receiving the prior written consent of the Seller; provided, however, Buyer shall be permitted to contact the New Mexico State Land Office and Bureau of Land Management after reasonable prior notice to and consultation with Seller, such notice to include a general description of the matters Buyer desires to discuss and address with the New Mexico State Land Office and Bureau of Land Management. Buyer shall hold in confidence all such information on the terms and subject to the conditions contained in Section 6.11. Notwithstanding the foregoing, Buyer shall have no right of access to, and none of the Companies, Seller nor any of their respective Affiliates shall have any obligation to provide any information, the disclosure of which (1) would reasonably be expected in the opinion of counsel to jeopardize any privilege available to Seller, the Companies or any of their respective Affiliates, (2) would cause Seller, the Companies or any of their respective Affiliates to breach a Material Contract or express confidentiality obligation, or (3) would result in a violation of Law.

- (b) Buyer understands that Seller has had discussions regarding a potential sale of the Interests and the preparation and negotiation of this Agreement, the Schedules hereto and the other documents contemplated herein, and that, (x) except to the extent discovery rules would otherwise permit review of such information by Buyer or the Companies if such information were in the possession of Seller,(y) except in the case of any allegation of fraud and (z) excluding information related to this Agreement (including the representations and warranties and covenants set forth herein and the Schedules and Exhibits attached hereto), (i) Buyer and the Companies shall not be entitled to use in connection with any disputes against Seller or the Companies (before or after Closing) any of Seller's or the Companies' internal drafts of this Agreement or emails or other written information (including in electronic form) relating to any potential sales process involving the Interests, and (ii) Buyer hereby agrees that (A) it shall not have any rights to any such information and (B) it shall not request any of the Companies, its Affiliates or any of their respective Representatives to provide to it any such information.
- (c) During the Interim Period and for ninety (90) days following the Closing Date, the Buyer agrees to indemnify and hold harmless Seller and the Companies, any of their respective Affiliates and their respective Representatives for any and all liabilities, losses, costs or expenses incurred by Seller or the Companies, any of their respective Affiliates or their respective Representatives arising out of the access rights under this <u>Section 6.2</u>, including any Claims by any of Buyer's Representatives for any injuries or property damage while accessing any Assets; provided, however, THIS INDEMNITY SHALL NOT APPLY TO ANY SUCH DAMAGE, INJURY, SICKNESS OR DEATH WHICH WAS CAUSED OR CONTRIBUTED TO BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF SELLER AND SELLER'S REPRESENTATIVES.

6.3 *Regulatory and Other Approvals*. During the Interim Period:

- (a) Each of Seller and Buyer shall use commercially reasonable efforts to obtain as promptly as practicable all consents and approvals that Seller or Buyer is required to obtain in order to consummate the transactions contemplated hereby by the date that is 30 days after the date hereof; *provided* that for purposes of clarification, the obtaining of such consents and approvals shall not be a condition to Closing except to the extent expressly set forth in Section 2.4.
- (b) In connection with any such filings, each of Seller and Buyer shall use commercially reasonable efforts to undertake promptly any and all actions required to complete lawfully the transactions contemplated by this Agreement prior to the date that is 30 days after the date hereof.
- (c) Each of Seller and Buyer shall provide prompt notification to the other when it becomes aware that any consent or approval referred to in this <u>Section 6.3</u> is obtained, taken, made, given or denied, as applicable.
- 6.4 *Resignation and Removal*. Notwithstanding anything in this Agreement to the contrary, at or prior to the Closing, Seller shall cause or accept the resignation or removal of any manager, partner, officer and/or director, as applicable, nominated or appointed by Seller or its Affiliates to any board or operating, management or other committee of each Company.

- 6.5 *Payments; Distributions*. Notwithstanding anything in this Agreement to the contrary, at or prior to Closing:
 - (a) Seller will, or will cause the Companies to, pay all Transaction Expenses;
- (b) Seller will, or will cause the Companies to, deliver reasonable and customary evidence demonstrating the release of all Liens for Indebtedness of the Companies on the Assets of the Companies (other than Permitted Liens);
- (c) Seller will cause, any accounts payable and/or accounts receivable between any Company, on the one hand, and a Non-Company Affiliate, on the other hand, to be paid in full; and
 - (d) Seller may cause the Companies to pay cash dividends and/or cash distributions to Seller or its Affiliates.

6.6 Tax Matters.

(a) <u>Tax Returns</u>.

- (i) Seller shall prepare or cause to be prepared and file or cause to be filed a 2022 Form 1065, U.S. Return of Partnership Income, of which Summit Parent is the primary filer, which report shall include all items and activities of the Companies through and including the Closing Date.
- (ii) Seller shall prepare or cause to be prepared any Tax Return of any Company required to be filed after the Closing Date (other than the Tax Return set forth in Section 6.6(a)(i)) for all Tax periods ending on or before the Closing Date. Such Tax Returns shall be prepared on a basis consistent with the past practice of Seller and the Companies except to the extent otherwise required by applicable Law. Reasonably in advance of the due date for the filing of any such Tax Return, Seller shall deliver a draft of such Tax Return, together with all supporting documentation and workpapers, to Buyer for its review and reasonable comment. Seller will (x) cause such Tax Return (as revised to incorporate Buyer's reasonable comments) to be timely filed and will provide a copy thereof to Seller, and (y) timely pay the Taxes shown due thereon.
- (iii) Buyer shall prepare or cause to be prepared any Tax Return of any Company for any Straddle Periods. Such Tax Returns shall be prepared on a basis consistent with the past practice of the Seller and the Companies except to the extent otherwise required by applicable Law. Reasonably in advance of the due date for the filing of any such Tax Return, Buyer shall deliver a draft of such Tax Return, together with all supporting documentation and workpapers, to Seller for its review and reasonable comment. Buyer will (x) cause such Tax Return (as revised to incorporate Seller's reasonable comments) to be timely filed and will provide a copy thereof to Seller, and (y) timely pay the Taxes shown due thereon.
- (b) <u>Proration of Straddle Period Taxes</u>. In the case of Taxes that are payable with respect to any Straddle Period:

- (i) Taxes that are either (A) based upon or related to income or receipts, or (B) imposed in connection with any sale or other transfer or assignment of property (real or personal, tangible or intangible), shall be allocated between the portion of such Straddle Period ending immediately prior to the Closing Date and the portion of such Straddle Period beginning on the Closing Date on a "closing of the books" basis by assuming the books of the Companies were closed at the end of the Closing Date; *provided* that exemptions, allowances or deductions that are calculated on an annual basis (including depreciation and amortization deductions) shall be allocated between the portion of such Straddle Period ending at the end of the Closing Date and the portion of such Straddle Period beginning on the Closing Date in proportion to the number of days in each period.
- (ii) Taxes that are imposed on a periodic basis with respect to the assets of the Companies shall be allocated between the portion of such Straddle Period ending immediately prior to the Closing Date and the portion of such Straddle Period beginning on the Closing Date by prorating each such Tax based on the number of days in the applicable Straddle Period that occur before the Closing Date, on the one hand, and the number of days in such Straddle Period that occur on and after the Closing Date, on the other hand. Consistent with the foregoing, the amount of such Taxes that is allocable to the portion of the Straddle Period (1) ending immediately prior to the Closing Date shall be deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction the numerator of which is the number of calendar days in the entire Straddle Period, and (2) beginning on the Closing Date shall be deemed to be the amount of such Taxes for the entire Straddle Period multiplied by a fraction the numerator of which is the number of calendar days from (and including) the Closing Date through the end of the Straddle Period, and the denominator of which is the number of calendar days in the entire Straddle Period. Such proration shall, initially, be based on the most recent Tax statements, received by the Seller as of the Closing Date. Following the Closing Date, each party shall, upon request of the other party, immediately reimburse the other party for any such Taxes for which said party is responsible but have been paid by the other party.
- (c) <u>Transfer Taxes</u>. Buyer shall bear and pay all transfer, sales, use, stamp, registration or other similar Taxes and any documentary, filing or recording fees resulting from the transfer of the Interests to Buyer pursuant to this Agreement ("*Transfer Taxes*"). Buyer and Seller shall cooperate in good faith to minimize, to the extent permissible under applicable Law, the amount of any such Transfer Taxes.
- (d) <u>Cooperation</u>. Buyer and Seller shall cooperate fully as and to the extent reasonably requested by the other Party, in connection with the filing of Tax Returns and any audit, litigation or other proceeding with respect to Taxes imposed on or with respect to the assets, operations or activities of the Companies (each a "*Tax Proceeding*"). Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information which are reasonably relevant to any such Tax Return or Tax Proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.

(e) <u>Amended Tax Returns</u>. No amended Tax Return with respect to a Pre- Closing Period or Straddle Period shall be filed by or on behalf of any Company without the prior written consent of Seller.

(f) <u>Purchase Price Allocation</u>.

- Seller shall prepare or cause to be prepared an allocation schedule (the "Allocation Schedule"), allocating the Purchase Price (plus any other items that are properly treated as consideration for U.S. federal income Tax purposes) among the Companies' assets in a manner consistent with Section 1060 of the Code. Seller shall deliver the Allocation Schedule to Buyer within 120 days after the Closing Date for Buyer's review and reasonable comments. Within 30 days after receiving such Allocation Schedule, Buyer shall notify Seller in writing if Buyer has any objections to the allocations on the Allocation Schedule and shall specify the basis for any such objections. If Buyer does not notify Seller of any objection to the Allocation Schedule, then it shall be deemed agreed to by Seller and Buyer and the Allocation Schedule shall be final and binding. If Buyer objects to any allocations on the Allocation Schedule, then Seller and Buyer shall negotiate in good faith to resolve any disagreement regarding the Allocation Schedule as soon as practicable (taking into account the due date of any Tax Returns on which the allocation set forth in the Allocation Schedule is required to be reflected) and memorialize the agreed allocation in a final Allocation Schedule, which shall be final and binding. If Buyer and Seller working in good faith are unable to agree on such disputed items on or prior to the 15th day following the date Buyer provides notice to Seller of any objections to the Allocation Schedule, then either Party may refer such dispute to PricewaterhouseCoopers LLP or, if that firm declines to act as provided in this Section 6.6(f), another firm of independent public accountants, mutually acceptable to Buyer and Seller, which firm shall make a final and binding determination as to all matters in dispute on a timely basis and promptly shall notify the Parties in writing of its resolution. Such accounting firm handling the dispute resolution shall not have the power to modify or amend any term or provision of this Agreement. The costs of such review and report shall be borne by Buyer, on the one hand, and the Seller, on the other hand, in proportion to the aggregate dollar amounts of such dispute(s) resolved in favor of the Buyer compared to the aggregate dollar amounts of such dispute(s) resolved in favor of the Seller (as determined by the accounting firm). The final Allocation Schedule shall be revised to take into account subsequent adjustments to the Purchase Price, in accordance with the provisions of Section 1060 of the Code and the Treasury Regulations thereunder.
- (ii) Buyer and Seller shall file all Tax Returns for Tax purposes consistent with the Allocation Schedule, as adjusted. None of Buyer, Seller, the Companies or their respective Affiliates shall take any position on any filed Tax Return that is inconsistent with the Allocation Schedule unless required to do so by applicable Law. If any Governmental Authority disputes the allocation set forth in the Allocation Schedule, the Party receiving notice of the dispute shall promptly notify the other Party of such dispute. Notwithstanding anything in this $\underline{\text{Section } 6.6(\underline{f})}$ to the contrary, neither Party shall be unreasonably impeded in its ability and discretion to negotiate, compromise and/or settle any Tax audit, claim or similar proceedings in connection with such Allocation Schedule.

- 6.7 **Public Announcements.** The Parties will maintain the confidentiality of this Agreement and its terms except that any Party may disclose this Agreement or any of its terms to any of the following if advised of the confidentiality obligations of such information: (a) any direct and indirect holders of Equity Interests in such Party or any Affiliate of such Party and (b) any potential lender to such Party. No Party hereto shall issue any press release or make any public statement with respect to this Agreement or the transactions contemplated hereby without the prior written consent of Buyer and Seller (such consent not to be unreasonably withheld, conditioned or delayed); *provided*, *however*, that the foregoing shall not restrict disclosures (i) to the extent required (upon advice of counsel) by applicable Law or regulations or the applicable rules of any stock exchange having jurisdiction over the Parties or their respective Affiliates, or (ii) made by either Party without reference to the other Party or any of the material terms of this Agreement or the transactions contemplated hereby.
- 6.8 **Removal of Name**. Effective immediately after the Closing, Buyer shall change the name of each Company so as not to include "Summit" or any variants thereof and shall provide evidence thereof to Seller. As promptly as practicable, but in any case within 60 days after the Closing Date, Buyer shall take all commercially reasonable steps to eliminate the name "Summit" and any variants thereof from the Assets of each Company, and, except with respect to such grace period for eliminating existing usage, shall have no right to use any logos, trademarks or trade names belonging to Seller or any of its Affiliates.
- 6.9 *Further Assurances*. Subject to the terms and conditions of this Agreement, at any time or from time to time after the Closing, at any Party's request and without further consideration, the other Party shall (and in the case of Buyer, Buyer shall and shall cause the Companies, as applicable, to) execute and deliver to such Party such other instruments of sale, transfer, conveyance, assignment and confirmation, provide such materials and information and take such other actions as such Party may reasonably request in order to consummate the transactions contemplated by this Agreement.
- 6.10 *R&W Policy*. As of the date hereof, Buyer has obtained and conditionally bound the R&W Policy, which policy expressly provides that (a) the insurer(s) issuing such policy have waived or otherwise shall not pursue any subrogation, contribution or other rights against Seller or any of its Affiliates and/or any of their respective past, present or future Representatives, except in the case of Actual Fraud by Seller, and (b) each of Seller, its Affiliates, and their respective past, present or future Representatives may rely upon and enforce such waiver of subrogation, contribution and other rights as express third party beneficiaries of such waiver of subrogation, contribution and other rights provisions. The aggregate amount of the premium with respect to the R&W Policy, together with the related underwriting fees, due diligence fees, net brokerage fees, premium Taxes and any other fees of the insurer relating to the R&W Policy shall, in the aggregate, be paid 50% by Buyer, on the one hand, and 50% by Seller, on the other hand, with such amounts being paid by the Parties when such payments are due under the R&W Policy and under the engagement letter with AON Risk Services Northeast, Inc. related to is insurance brokerage services. From and after inception of the R&W Policy, unless terminated in its entirety, Buyer shall not (and shall cause its Affiliates not to) amend, modify, or otherwise change, terminate, or waive any provision of the R&W Policy in a manner materially adverse to Seller, any of its Affiliates or any of their respective Representatives, including any rights of the insurer to subrogate or seek recovery from Seller, any of Seller's Affiliates or any of Seller's Representatives.

- 6.11 *Confidentiality*. For a period of two years after Closing, Seller shall, and shall cause its Affiliates (excluding, for the avoidance of doubt, Buyer and the Companies) and direct its Representatives to, treat as confidential and not disclose or otherwise use (including for competitive purposes) any and all Confidential Material except (a) for use under this Agreement or as otherwise agreed to in writing by Buyer, (b) for disclosures to any Governmental Authority having jurisdiction to require disclosure or to any arbitral body or in connection with any routine audit, supervisory examination, regulatory sweep or other regulatory inquiry to the extent required by same, (c) as otherwise may be required by Law, (d) for disclosures during the course of litigation or in connection with enforcing any rights under this Agreement or any Transaction Document, (e) as required for internal audit, financial and tax purposes, or (f) for disclosures to its Representatives in the ordinary course of business.
- 6.12 *No Shop.* During the Interim Period, the Seller shall not, and shall cause the Companies and their respective Affiliates and the directors, officers, employees, managers, members, agents, advisors or other Representatives to not, solicit, entertain or accept any offer to purchase, or make any offer to sell the Companies. Furthermore, during the Interim Period, Seller shall, and shall cause the Companies and their respective Affiliates and Representatives to, deal exclusively with the Buyer, its Affiliates and their Representatives with respect to the transactions contemplated by this Agreement. The Seller shall not, and shall cause the Companies and their respective Affiliates and Representatives to not, directly or indirectly, facilitate, encourage, solicit or entertain offers or inquiries from, participate in negotiations or discussions with, enter into any agreement or letter of intent with, or disclose any non-public information to, any other person or entity, in each case, with respect to the transactions contemplated by this Agreement, any sale of the Companies, merger or similar transaction or any other transaction having substantially the same economic effect.

6.13 Release.

(a) Effective as of the Closing, absent any Actual Fraud, Seller (on its own and on behalf of its Affiliates and each of their respective direct and indirect equityholders, successors and assigns) hereby unconditionally and irrevocably and forever releases and discharges Buyer and the Companies and all of their respective past, present and future directors, officers, mangers, members, equityholders and employees, and all of the foregoing Persons' successors, assigns and agents (each, a "Buyer Released Party"), of and from, and hereby unconditionally and irrevocably waives, any and all claims, debts, losses, expenses, Proceedings, covenants, liabilities, suits, judgments, damages, actions and causes of action, obligations, accounts, and liabilities of any kind or character whatsoever, known or unknown, contingent or unconditional, matured or unmatured, fixed or variable, suspected or unsuspected, in contract or tort, at Law or in equity, direct or indirect that Seller (or any of its Affiliates and each of their respective direct and indirect equityholders, successors and assigns) ever had, now has or ever may have or claim to have against any Buyer Released Party, that arise out of, relate to, or are in any way connected with Seller's (or any of Seller's Affiliates or Seller's or such Affiliate's respective direct and indirect equityholders, successors and assigns) past or present (x) direct or indirect ownership of any equity securities issued by the Companies or (y) relationship with the Companies (each, a "Seller Released Claim"); provided, however, that this release shall not affect (a) any Seller Released Claims related to the rights of Seller or its Affiliates under this Agreement or any Transaction Document, including any breach or alleged breach of or liabilities under this Agreement or any other Transaction Documents

or any recourse or similar rights of Seller or its Affiliates set forth in this Agreement or (b) any right to indemnification, exculpation or advancement of expenses, whether under corporate indemnification, director and officer insurance or otherwise, to which any individual may be entitled as a result of such individual's service as a director, manager, officer, or employee of the Companies. For purposes of clarification and avoidance of doubt, notwithstanding any other provision of the release in this Section 6.13(a), this release does not (i) waive or release any claim for breach or enforcement of this release, or (ii) waive or release any right or claim that may not be waived or released by applicable Law.

Effective as of the Closing, absent any Actual Fraud, Buyer (on its own and on behalf of its Affiliates, including after the Closing, the Companies, and each of their respective direct and indirect equityholders, successors and assigns) hereby unconditionally and irrevocably and forever releases and discharges Seller and its Affiliates and all of their respective past, present and future directors, officers, mangers, members, equityholders and employees, and all of the foregoing Persons' successors, assigns and agents (each, a "Seller Released Party"), of and from, and hereby unconditionally and irrevocably waives, any and all claims, debts, losses, expenses, Proceedings, covenants, liabilities, suits, judgments, damages, actions and causes of action, obligations, accounts, and liabilities of any kind or character whatsoever, known or unknown, contingent or unconditional, matured or unmatured, fixed or variable, suspected or unsuspected, in contract or tort, at Law or in equity, direct or indirect that Buyer (or any of its Affiliates, including the Companies, and each of their respective direct and indirect equityholders, successors and assigns) ever had, now has or ever may have or claim to have against any Seller Released Party, that arise out of, relate to, or are in any way connected with the Companies, the Companies' Assets, or the Business, including the direct or indirect ownership of any equity securities issued by the Companies (each, a "Buyer Released Claim"); provided, however, that this release shall not affect any Buyer Released Claims related to the rights of Buyer under this Agreement or any Transaction Document, including any breach or alleged breach of or liabilities under this Agreement or any other Transaction Documents or any recourse or similar rights of Buyer set forth in this Agreement. For purposes of clarification and avoidance of doubt, notwithstanding any other provision of the release in this Section 6.13(b), this release does not (i) waive or release any claim for breach or enforcement of this release, or (ii) waive or release any right or claim that may not be waived or released by applicable Law.

6.14 Non-Solicitation and Non-Competition.

- (a) As a material inducement for Buyer to enter into this Agreement, and in order to protect the goodwill that is conveyed hereunder, Seller agrees to comply and shall cause its Affiliates (collectively, the "*Restricted Parties*" and each a "*Restricted Party*") to comply with the terms of this <u>Section 6.14</u>.
- (b) Until the 18-month anniversary of the Closing Date, Seller will not and will cause each Restricted Party to not, directly or indirectly through any Person (whether as an officer, director, manager, member, employee, partner or equityholder), solicit, induce or hire, or attempt to solicit, induce or hire, or otherwise encourage any Continuing Employee to terminate such Person's employment with (in the case of an employee) or cease providing such Person's services to (in the case of a consultant or agent) the Companies, Buyer or any of Buyer's Affiliates; provided that the Restricted Parties shall not be precluded from conducting general solicitations

for employment or other services contained in a newspaper, other periodical or on the internet or hiring any such Continuing Employee who responds thereto, so long as such solicitations are not targeted at employees, consultants or agents of the Companies, Buyer or any of Buyer's Affiliates or from hiring any such Continuing Employee whose employment with the Companies, Buyer or any of Buyer's Affiliates has been terminated for a period of sixty (60) days.

- (c) Until the 12-month anniversary of the Closing Date, the Restricted Parties shall not directly or indirectly manage, operate, own, or provide funding to or invest in any Person that manages or operates any gas gathering system or gas processing plant in each case that is located within twenty (20) miles of the Lane Gas Processing Facility (the "*Competitive Activities*"). Notwithstanding the foregoing, none of the following shall be deemed a violation of this <u>Section 6.14</u> nor shall the restrictive covenants in this <u>Section 6.14</u> prevent or prohibit the Restricted Parties from taking any of the following actions:
 - (i) engaging in or consummating a Change of Control, including if a Person or Persons become an Affiliate of Seller as a result of the Change of Control and such Person or Persons engages in Competitive Activities at the time of such Change of Control;
 - (ii) entering into or consummating any merger, joint venture or any other business transaction or combination with any Person that participates in the Competitive Activities at the time of the closing of such merger, joint venture or business transaction or combination, so long as neither the Seller nor its Affiliates manages or exercises control over the Competitive Activities of such Person;
 - (iii) acquiring a direct or indirect interest, by joint venture, merger, stock purchase, asset purchase or other business purchase or combination, in a business (the "*Acquired Business*") that is engaged in the Competitive Activities so long as the Competitive Activities do not represent more than fifty percent (50%) of the gross revenues of such Acquired Business;
 - (iv) engaging in any (A) expansion or extension of the Double E Pipeline, including interconnecting the Double E Pipeline with other systems, or (B) managing, operating, owning, funding, or investing in any Person or business development activity to increase Double E Pipeline's access to or receipts of natural gas supplies downstream of processing; or
 - (v) performing services for or engaging in any business with Buyer or any of its Affiliates or Subsidiaries, including performing any duties or obligations as may be required pursuant to the Transition Services Agreement.

Notwithstanding anything to the contrary in this <u>Section 6.14</u>, the Parties acknowledge and agree that in the event of a Change of Control, this <u>Section 6.14</u> shall terminate and Seller and its Affiliates shall no longer be bound by or subject to the restrictive covenants herein.

(d) Each Restricted Party acknowledges and agrees that the scope of activity restrictions, and duration of the covenants contained in Section 6.14(b) and Section 6.14(c) are the result of arm's-length bargaining and are fair and reasonable in light of (i) the Restricted Parties'

level of control over and contact with the Companies' and Buyer's (and any of its respective subsidiaries') business, and association with the Companies and Buyer's (and any of its respective subsidiaries') goodwill in all jurisdictions in which the Companies and Buyer (and any of its respective subsidiaries) conduct business; and (ii) the consideration that Seller is receiving in connection with the transactions contemplated by this Agreement and the goodwill being conveyed and for which Buyer is paying. Each Restricted Party further acknowledges and agrees that the restrictions set forth in Section 6.14(b) and <u>Section 6.14(c)</u> do not impose any greater restraint than is necessary to protect the legitimate business interests of Buyer, have been specifically negotiated by sophisticated commercial parties, and constitute a material inducement for Buyer enter into this Agreement and consummate the transactions contemplated by this Agreement. The covenants contained in <u>Section 6.14(b)</u> and <u>Section 6.14(c)</u> and each provision (and portion) hereof are severable and distinct covenants and provisions (and portions). The invalidity or unenforceability of any such covenant or provision (or portion) as written shall not invalidate or render unenforceable the remaining covenants or provisions (or portions) hereof, and any such invalidity or unenforceability in any jurisdiction shall not invalidate or render unenforceable such covenant or provision (or portion) in any other jurisdiction. It is the desire and intent of the Buyer and the Restricted Parties that the provisions of Section 6.14(b) and Section 6.14(c) (and all portions thereof) be enforced to the fullest extent permitted under applicable Law. Each Restricted Party acknowledges that Section 6.14(b) and Section 6.14(c) are reasonable and enforceable in all respects; nonetheless, if any of the restrictions of <u>Section 6.14(b)</u> and <u>Section 6.14(c)</u> (or any portions thereof) are found by a court of competent jurisdiction to be unreasonable, overly broad, or otherwise unenforceable, Buyer and the Restricted Parties intend that such restrictions (and the applicable portions thereof) shall not be thereby terminated but shall be modified by such court of competent jurisdiction so as to be valid and enforceable and, as so modified, to be fully enforced. Each Restricted Party further acknowledges that a breach or threatened breach of Section 6.14(b) and Section 6.14(c) would cause irreparable harm to Buyer for which it would have no adequate remedy at Law, that it would be impractical and extremely difficult to determine Buyer's and its Affiliates' damages in the event of a breach or threatened breach, and that Buyer and its Affiliates shall be entitled to seek immediate injunctive relief and specific performance as remedies for any such breach in accordance with Section 10.24. Such remedies shall not be the exclusive remedies for a breach or threatened breach of <u>Section 6.14(b)</u> and <u>Section 6.14(c)</u> but shall be in addition to all other remedies available at law an equity, including the recovery of damages.

- 6.15 **Notice of Events.** During the Interim Period, the Seller shall, on behalf of themselves and the Companies, deliver written notice to Buyer (each such notice, a "**Supplement**") if any Company or Seller become aware of any matter first arising or first occurring during the Interim Period as necessary to complete or correct any information in the Schedules with respect to any representation or warranty or any breach of any covenant set forth herein. For all purposes of this Agreement, the Schedules shall be deemed to include only that information contained therein on the date hereof and shall be deemed to exclude all information contained in any supplement or modification thereto. Notwithstanding the foregoing, the Seller shall only have an obligation to deliver a Supplement in the event that such matter has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- 6.16 *Assignment and Assumption of Certain Excluded Liabilities*. Prior to the Closing, Seller shall use commercially reasonable efforts to cause the Companies, as applicable, to convey and assign to Seller the Excluded Claims, including a partial assignment of any necessary

associated portions of any agreements related to the Excluded Claims, and Seller shall assume the Excluded Claims; provided, however, that such assignment and assumption shall not be a condition to Closing required under <u>Article VII</u>, nor shall compliance by Seller with this <u>Section 6.16</u> in any event constitute a breach or non-compliance by Seller for purposes of <u>Section 7.2</u> or for purposes of termination of this Agreement pursuant to <u>Article IX</u>.

- 6.17 *Continuing Employees.* Beginning on the date hereof, Seller shall make available to Buyer each of the Subject Employees listed on <u>Schedule 6.17</u> to discuss potential continuing employment following the Closing with a Company, Buyer or an Affiliate of Buyer. Seller agrees that it will not directly or indirectly discourage any of such Subject Employees from accepting an offer of employment from Buyer. Buyer shall provide Seller, in writing, not later than three (3) calendar days prior to the Closing Date, a list of those Subject Employees who have accepted offers to begin employment with a Company, Buyer or an Affiliate of Buyer as of and immediately after the Closing (each, a "*Continuing Employee*"). Further, Seller agrees that it will use commercially reasonable efforts to reasonably assist Buyer in communicating with the Continuing Employees regarding the transactions contemplated by this Agreement.
- 6.18 *Third Party Consents*. Seller shall obtain and deliver all consents set forth on <u>Schedule 6.18</u> in form and substance reasonably satisfactory to Buyer prior to the Closing.
- 6.19 *IRB Arrangement*. Immediately prior to or in connection with the Closing, the Seller shall cause the Companies, as applicable, to repay and terminate the IRB Arrangement and shall cause Summit Permian to purchase the IRB Assets.
- 6.20 *Double E*. Buyer shall, and shall cause MRC Permian to, reasonably cooperate with Seller in connection with the capacity release from the Double E FTSA from Summit Marketing to MRC Permian.
- 6.21 **Surety Bond Replacement**. From and after the date hereof, Buyer shall use commercially reasonable efforts to, and in any event within fifteen (15) days following Closing, Buyer shall cause Summit Permian to, obtain replacement surety bonds for those such bonds listed on <u>Schedule 3.18</u> and provide evidence of the same to Seller. After Seller receives evidence of the replacement surety bonds from Buyer, Seller shall terminate the existing surety bonds set forth on <u>Schedule 3.18</u>. From and after the Closing until such surety bonds have been so replaced and terminated in accordance with this <u>Section 6.21</u>, Buyer shall indemnify the Seller and its Affiliates from and against any losses actually incurred by any of them arising out of or with respect to any of the then-outstanding surety bonds.
- 6.22 *Termination of Summit Marketing Transaction Confirmation.* Immediately prior to or in connection with the Closing, the Seller shall cause Summit Permian and Summit Marketing to terminate that certain Transaction Confirmation.
- 6.23 *Post-Closing Matters.* Following the Closing, Seller shall use its commercially reasonable efforts to take the actions described on <u>Exhibit H</u>.

ARTICLE VII BUYER'S CONDITIONS TO CLOSING

The obligation of Buyer to consummate the Closing is subject to the fulfillment of each of the following conditions (except to the extent waived in writing by Buyer in its sole discretion); *provided*, that Buyer may not rely on the failure of any condition set forth in this <u>Article VII</u> to be satisfied if such failure has been primarily caused by, or is primarily the result of, Buyer's failure to comply with its obligations under this Agreement:

- 7.1 **Representations and Warranties.** (i) The Seller Fundamental Representations shall be true and correct in all respects as of the date hereof and as of the Closing as if made on the Closing Date (except to the extent that such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct as of such earlier date) except for any *de minimis* breaches; and (ii) all other representations and warranties of Seller made in this Agreement (other than representations and warranties described in clause (i) above) will be true and correct in all respects as of the date hereof and as of the Closing Date as if made on the Closing Date in all material respects (other than those representations and warranties of Seller that are qualified as to materiality or a Material Adverse Effect, which shall be true and correct in all respects) (except to the extent that such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct as of such earlier date) except in the case of this <u>clause (ii)</u> for all such breaches, if any, of such representations and warranties that individually or in the aggregate would not have a Material Adverse Effect.
- 7.2 **Performance**. Seller shall have performed and complied, in all material respects, with the agreements, covenants and obligations required by this Agreement to be performed or complied with by Seller at or before the Closing.
- 7.3 *No Material Adverse Effect*. During the Interim Period, no Material Adverse Effect shall have occurred and be continuing.
- 7.4 *Officer's Certificate*. Seller shall have delivered to Buyer at the Closing an officer's certificate, dated as of the Closing Date, certifying as to the matters set forth in <u>Section 7.1 Section 7.2</u> and <u>Section 7.3</u>.
- 7.5 *Orders and Laws*. There shall not be any Law or current Proceedings restraining, enjoining or otherwise prohibiting or making illegal the consummation of the transactions contemplated by this Agreement.
- 7.6 **Deliverables**. The Seller shall have delivered, caused to be delivered, or be ready, willing and able to deliver, to Buyer the deliverables contemplated by <u>Section 2.4</u>.

ARTICLE VIII SELLER'S CONDITIONS TO CLOSING

The obligation of Seller to consummate the Closing is subject to the fulfillment of each of the following conditions (except to the extent waived in writing by Seller in its sole discretion); *provided*, that Seller may not rely on the failure of any condition set forth in this Article VIII to be

satisfied if such failure has been primarily caused by, or is primarily the result of, Seller's failure to comply with its obligations under this Agreement:

- 8.1 **Representations and Warranties**. (i) The Buyer Fundamental Representations shall be true and correct in all respects as of the date hereof and as of the Closing as if made on the Closing Date (except to the extent that such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct as of such earlier date) except for any *de minimis* breaches; and (ii) all other representations and warranties of Buyer made in this Agreement (other than representations and warranties described in clause (i) above) will be true and correct in all respects as of the date hereof and as of the Closing Date as if made on the Closing Date in all material respects (other than those representations and warranties of Buyer that are qualified as to materiality or a material adverse effect, which shall be true and correct in all respects) (except to the extent that such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct as of such earlier date) except in the case of this <u>clause (ii)</u> for all such breaches, if any, of such representations and warranties that individually or in the aggregate would not have a material adverse effect on Buyer's ability to consummate the transactions contemplated herein.
- 8.2 *Performance*. Buyer shall have performed and complied, in all material respects, with the agreements, covenants and obligations required by this Agreement to be so performed or complied with by Buyer at or before the Closing.
- 8.3 *Officer's Certificate*. Buyer shall have delivered to Seller at the Closing a certificate of an officer of Buyer, dated as of the Closing Date, certifying as to the matters set forth in <u>Section 8.1</u> and <u>Section 8.2</u>.
- 8.4 *Orders and Laws*. There shall not be any Law or current Proceedings restraining, enjoining or otherwise prohibiting or making illegal the consummation of the transactions contemplated by this Agreement.
- 8.5 *Deliverables*. The Buyer shall have delivered, caused to be delivered, or be ready, willing and able to deliver, to Seller the deliverables contemplated by <u>Section 2.5</u>.
- 8.6 **Double E FTSA.** The Double E FTSA shall have been permanently released from Summit Marketing to MRC Permian Company.

ARTICLE IX TERMINATION

- 9.1 *Right of Termination*. Prior to Closing, this Agreement may be terminated:
 - (a) By written consent of Seller and Buyer;
- (b) By Seller or Buyer if the Closing has not occurred before 60 days after the date hereof (the "*Outside Date*"), unless such failure of the Closing to occur on or prior to the Outside Date is due to such Party's failure to perform or comply with any of its obligations under

this Agreement will have been the primary cause of, or will have resulted in, the failure of the Closing to occur on or before the Outside Date;

- (c) By Seller or Buyer if consummation of the transactions contemplated by this Agreement is permanently enjoined, prohibited or otherwise restrained by the terms of a final, non-appealable order that has not been vacated, withdrawn or overturned (other than a temporary restraining order) of a Governmental Authority having competent jurisdiction or the terms of any Law; *provided*, *however*, that the right to terminate this Agreement under this Section 9.1(c) shall not be available to a Party if the enactment, promulgation, issuance, entry, enforcement or adoption of such Law or issuance of such order was primarily due to the failure of such Party to perform or comply with any of its obligations under this Agreement;
- (d) By Seller, if Seller is not then in material breach of any provision of this Agreement, if satisfaction of any of the conditions in <u>Article VIII</u> on or prior to the Outside Date is or becomes impossible (other than through failure of Seller to comply with its obligations under this Agreement), and such breach, if curable, is not cured within 10 days after receipt of written notice thereof from Seller; *provided*, *however*, that if, at the end of such 10-day period, Buyer is proceeding in good faith to cure such breach, Buyer shall have an additional 10 days from the end of such 10-day period to effect such cure (or any shorter period of time that remains between the date Seller provides written notice of such violation or breach and the Outside Date); or
- (e) By Buyer, if Buyer is not then in material breach of any provision of this Agreement, if satisfaction of any of the conditions in <u>Article VII</u> on or prior to the Outside Date is or becomes impossible (other than through failure of Buyer to comply with its obligations under this Agreement); *provided*, *however*, that if, at the end of such 10-day period, Seller is proceeding in good faith to cure such breach, Seller shall have an additional 10 days from the end of such 10- day period to effect such cure (or any shorter period of time that remains between the date Buyer provides written notice of such violation or breach and the Outside Date).

9.2 Effect of Termination.

(a) If any Party terminates this Agreement pursuant to Section 9.1, then such Party shall promptly give notice to the other Party, specifying the provision hereof pursuant to which such termination is made, and all obligations and liabilities of the Parties under this Agreement shall terminate and become void and neither Party shall have any further rights against the other Party by virtue of this Agreement for breach of any representation, warranty, covenant or agreement, save and except the Parties' rights and obligations under Section 9.2(b) and Section 9.2(c); provided, however, that the terms of Section 6.2(b), Section 6.2(c), Section 6.11 (which shall survive in accordance with its terms), this Section 9.2, Section 9.2(b), Section 10.10, Section 10.13, Section 10.14, Section 10.15, Section 10.16, Section 10.17, Section 10.18, Section 10.19, Section 10.20, Section 10.21, Section 10.22, Section 10.23, Section 10.24, Section 10.25 and Section 10.26 shall remain in full force and effect and survive any termination of this Agreement. Upon the termination of this Agreement, Seller shall be free immediately to enjoy all rights of ownership of the Companies, the Interests and the Assets of the Companies and to sell, transfer, encumber or otherwise dispose of the same to any Person without any restriction under this Agreement. The Confidentiality Agreement shall not be affected by a termination of this Agreement.

- (b) If at the time this Agreement is terminated by Seller pursuant to <u>Section 9.1(b)</u> or <u>Section 9.1(d)</u>, Buyer is in material breach and Seller is not in breach of this Agreement and is otherwise ready, willing and able to consummate the Closing, then Seller may pursue all available remedies at law or in equity against the Buyer, including seeking to obtain a money judgment against the Buyer for any damage to the Seller that may result from any breach by the Buyer of any of its representations, warranties or covenants contained in this Agreement.
- (c) If at the time this Agreement is terminated by Buyer pursuant to <u>Section 9.1(b)</u> or <u>Section 9.1(e)</u>, Seller is in material breach and Buyer is not in breach of this Agreement and is otherwise ready, willing and able to consummate the Closing, Buyer may pursue all available remedies at law or in equity against the Seller, including seeking to obtain a money judgment against the Seller for any damage to the Buyer that may result from any breach by the Seller of any of its representations, warranties or covenants contained in this Agreement.
- 9.3 **Return of Documentation**. Upon termination of this Agreement, Buyer shall return to Seller all original (and destroy all copies of) title, engineering, geological and geophysical data, environmental assessments and reports, maps and other information furnished by or on behalf of Seller to Buyer in connection with its due diligence investigation of the Companies, the Interests and the Assets of the Companies, and an officer of Buyer shall certify same to Seller in writing.

ARTICLE X MISCELLANEOUS

10.1 Survival.

- (a) The Fundamental Representations of the Parties shall survive until the termination of the statute of limitations for any underlying claim.
- (b) All other representations and warranties of the Seller contained in this Agreement, or in any instrument or certificate delivered pursuant to this Agreement, shall terminate as of the Closing and no claim shall be brought with respect to any such representation or warranty from and after the Closing; *provided*, *however*, this <u>Section 10.1(b)</u> shall not affect the time periods during which any claim by Buyer may be made solely under the R&W Policy and solely for the purposes of the R&W Policy, all such representations and warranties shall survive until the expiration of the R&W Policy pursuant to its terms.
- (c) The agreements, obligations and covenants of the Parties contained in this Agreement, to the extent such agreements, obligations or covenants do not by their terms require performance in whole or in part after the Closing and any other rights arising out of any breach of such agreements, obligations or covenants shall terminate at the Closing and no claim shall be brought with respect to any such agreements, obligations and covenants from and after the Closing. The agreements, obligations and covenants of the Parties contained in this Agreement, to the extent such agreements, obligations or covenants by their terms require performance in whole or in part after the Closing, shall survive until the earlier of the time that they are fully discharged or the expiration of the applicable statute of limitations and no claim shall be brought with respect to any such agreements, obligations and covenants from and after such time.

- (d) No indemnifying Person will be liable for any Losses that are subject to indemnification under $\underline{\text{Section 10.4}}$ unless a written demand for indemnification under this Agreement is delivered by the indemnified Person to the indemnifying Person with respect thereto prior to 5:00 P.M. Central Time on the date pursuant to this $\underline{\text{Section 10.1}}$ on which the survival period of the applicable representations and warranties or covenants expires, to assert a Claim for indemnification describing such Claim in reasonable detail, including the factual circumstances giving rise to and the provisions under this Agreement on which such Claim is based. In the event notice of any claim for indemnification under this $\underline{\text{Article X}}$ has been given within the applicable survival period, the representations, warranties or covenants that are the subject of such Indemnification Claim shall survive solely with respect to such claim until such time as such claim is finally resolved, it being understood that such representations, warranties and covenants shall not survive with respect to any other claim that has not been timely given within such survival period.
- 10.2 **Sole and Exclusive Remedy**. Subject to <u>Section 10.3</u>, after the Closing, except in the case of Actual Fraud, the sole and exclusive remedy of Buyer and its Affiliates with respect to the breach of any representations and warranties of Seller pursuant to <u>Section 10.1(b)</u> shall be limited to the R&W Policy (including, for the avoidance of doubt, whether Buyer obtains such R&W Policy at or after the Closing, if at all, and maintains, following the Closing, the R&W Policy, whether the R&W Policy is revoked, cancelled or modified in any manner after the Closing, or whether any Proceeding by Buyer or any of its Affiliates or Representatives is denied under the R&W Policy). Buyer acknowledges and agrees, on behalf of itself and its Affiliates, that following the Closing, neither Seller nor any of its Affiliates or Representatives shall have any liability for any breach or inaccuracy of any representation or warranty within <u>Section 10.1(b)</u>, except in the event of Actual Fraud. For the avoidance of doubt, the absence of coverage under the R&W Policy for any reason, including due to exclusions from coverage thereunder or the failure of the R&W Policy to be in full force and effect for any reason, shall not expand, alter, amend, change or otherwise affect Seller's or its Affiliates' or Representatives' or any of their respective successors' and assigns' liability under this Agreement.
- 10.3 *Indemnification by Seller*. Seller hereby agrees to indemnify and hold harmless the Buyer, its Affiliates and the respective directors, managers, officers, employees, successors and permitted assigns of any of the foregoing ("*Buyer Indemnified Parties*") from and against any and all Losses which any of them may suffer, sustain or become subject to, as the result of, arising out of, relating to or in connection with:
- (a) any misrepresentation or breach of any of the Seller Fundamental Representations made by the Seller in this Agreement;
 - (b) the breach by Seller of any covenant in this Agreement or any Transaction Document by Seller;
- (c) (i) any Taxes of Seller, (ii) any Taxes of the Companies for all Pre-Closing Periods, and (iii) with respect to any Straddle Period, any Taxes of the Companies attributable to the portion of such Straddle Period that ends on and includes the Closing Date;

- (d) any Indebtedness of the Company outstanding as of the Closing to the extent not deducted in the Purchase Price;
 - (e) any Proceeding pending as of the Closing; and
 - (f) any Excluded Liability.
- 10.4 *Indemnification by Buyer*. Buyer hereby agrees to indemnify and hold harmless Seller, its Affiliates and the respective directors, managers, officers, employees, successors and permitted assigns of any of the foregoing ("*Seller Indemnified Parties*") from and against any and all Losses which any of them may suffer, sustain or become subject to, as the result of, arising out of, relating to or in connection with:
- (a) any misrepresentation or breach of any of the Buyer Fundamental Representations made by the Buyer in this Agreement;
 - (b) the breach of any covenant in this Agreement or any Transaction Document by buyer.

10.5 *Indemnification Order of Recovery.*

- (a) In the event any Buyer Indemnified Parties seek indemnification from Seller for Losses arising based on matters set forth in Section 10.3(a) or Section 10.3(c), Buyer shall first use commercially reasonable efforts to promptly seek collection and recovery pursuant to the R&W Policy to the fullest extent of the limits under the R&W Policy; provided, however, that nothing shall require the Buyer to commence litigation to recover proceeds under such R&W Policy. In the event Buyer is unable to recover for such Losses in accordance with the foregoing, Buyer may seek recovery for such Losses directly from the Seller.
- (b) Subject to <u>Section 10.5(a)</u>, in the event Buyer Indemnified Parties seek indemnification from Seller for any Losses arising from Actual Fraud or any other matters set forth in <u>Section 10.3</u>, Buyer may seek recovery for such Losses directly from the Seller.
- (c) In the event Seller Indemnified Parties seek indemnification from Buyer for any Losses arising from Section 10.4, Seller may seek recovery for such Losses directly from the Buyer.

10.6 Limits on Indemnification.

- (a) Notwithstanding anything to the contrary herein, absent Actual Fraud, in no event shall the aggregate liability of the Seller in connection with this Agreement or the Transaction Documents, including the transactions contemplated herein and therein, exceed the Purchase Price.
- (b) The Buyer Indemnified Parties shall take, and shall cause the Companies to take, all commercially reasonable steps to mitigate a Loss upon becoming aware thereof and of the event which gives rise thereto. The Seller Indemnified Parties shall take and cause their respective Affiliates to take all commercially reasonable steps to mitigate a Loss upon becoming aware

thereof and of the event which gives rise thereto. Such parties will each also use commercially reasonable efforts to pursue claims and collect any amounts to which they may be entitled under insurance policies or from third parties (pursuant to indemnification agreements or otherwise). The Parties agree that no Indemnified Party shall have any recourse under this Article X for any Losses that such Indemnified Party would not have suffered had such Indemnified Party exercised commercially reasonable efforts to mitigate such Losses within a reasonable amount of time following the discovery by such Indemnified Party of the fact, event or circumstance giving rise to such Losses (and for an Indemnified Party that is not a natural Person, the bringing to the attention of a responsible officer thereof of such fact, event or circumstance).

- (c) Notwithstanding the foregoing or anything herein in the contrary, the Buyer Indemnified Parties shall not be entitled to indemnification pursuant to this <u>Article X f</u>or any Loss underlying such indemnification claim to the extent that such Loss (or any part thereof) arises, or is increased, as a result of a change after the Closing in any accounting principle, method or policy (including any such change in GAAP or application thereof), in any Tax reporting practice of the Companies and in no event shall Seller have any liability for indemnification under this <u>Article X</u> for any Losses to the extent such Losses are caused or initiated by any action or omission by any Buyer Indemnified Party or any Company at the request or direction of any Buyer Indemnified Party, including to the extent any Losses resulted from the bad faith, gross negligence or willful misconduct of such Buyer Indemnified Parties.
- (d) The amount of any and all Losses indemnifiable under this Agreement shall be determined net of any amounts actually recovered by a Buyer Indemnified Party (or any of its Affiliates) with respect to such Losses from any third party, including any indemnity, contribution or other similar payment, or under or pursuant to any insurance policy pursuant to which or under which such Buyer Indemnitee (or any of its Affiliates) has rights (including the R&W Policy), it being understood that any amounts recovered under or pursuant to any insurance policy shall be calculated net of any reasonable costs associated with pursuing such insurance proceeds and reasonable out-of-pocket costs and any costs or expenses attributable to increases in insurance premiums. If Seller has actually indemnified or held harmless Buyer or another Buyer Indemnified Party upon a claim for indemnification pursuant to this Agreement and, subsequently, there is an actual payment under the R&W Policy (whether to Buyer, another insured under the R&W Policy or a third party) for a claim involving the same or related facts that gave rise to the indemnification claim, then Buyer shall (or shall cause the other insured under the R&W Policy to) use such insurance proceeds (net of any costs of recovery or other reasonable costs and expenses related to the insurance claim) and promptly reimburse Seller for any indemnification amount previously paid by Seller.
- (e) For purposes of this <u>Article X</u>, (i) any inaccuracy in or breach of any representation or warranty and (ii) the amount of any Losses arising from such breach of any representation or warranty, shall be determined without regard to any materiality, Material Adverse Effect or other similar qualification contained in or otherwise applicable to such representation or warranty; provided, however, that this <u>Section 10.6</u> shall not apply to any reference to a Material Contract or a material Permit or <u>Section 3.5(a)</u>.
- (f) Notwithstanding anything to the contrary contained in this Agreement, under no circumstances will any Party or any of its Affiliates be entitled to recover more than one

time for any Loss under this Agreement, and to the extent a Party or any of its Affiliates is compensated for a matter through the adjustments provided for in Section 2.6 or otherwise, such Party and its Affiliates will not have a separate right to indemnification or adjustment in another portion of Section 2.6 for such matter. Without limiting the generality of the prior sentence, if a set of facts, conditions or events constitutes a breach of more than one representation, warranty, covenant or agreement that is subject to the indemnification obligations under this Article X as applicable, only one recovery of Losses shall be allowed, and in no event shall there be any indemnification or duplication of payments or recovery under different provisions of this Agreement arising out of the same facts, conditions or events.

10.7 Indemnification Procedures.

- In the event that any Proceedings shall be instituted, or that any Claim shall be asserted, by any third party in respect of which payment may be sought under Section 10.3 and Section 10.4 hereof (excluding any Proceeding of a claim in respect of which payment is sought solely from the R&W Policy) (an "Indemnification Claim"), the Indemnified Party shall promptly cause written notice of the assertion of any Indemnification Claim of which it has knowledge which is covered by this Agreement to be forwarded to the Indemnifying Party, including reasonable supporting details (including an estimate of the Losses associated therewith) and copies of all correspondence related thereto. The failure of the Indemnified Party to give reasonably prompt notice of any Indemnification Claim shall not release, waive or otherwise affect the Indemnifying Party's obligations with respect thereto except to the extent that the Indemnifying Party is materially prejudiced as a result of such failure. With respect to any Indemnification Claim, the Indemnifying Party shall be entitled, at its sole cost and expense, (i) to participate in the defense of such Indemnification Claim to the extent giving rise to the Indemnified Party's claim for indemnification or (ii) at its option (subject to the limitations set forth below), to assume control of such defense and appoint lead counsel reasonably acceptable to the Indemnified Party; provided that, as a condition precedent to the Indemnifying Party's right to assume control of such defense, it must first notify the Indemnified Party in writing within thirty days after the Indemnified Party has given notice of the Indemnification Claim that the Indemnifying Party will indemnify the Indemnified Party from and against the entirety of any Losses (without any limitations) the Indemnified Party may suffer resulting from, arising out of, relating to, or caused by the Indemnification Claim in accordance with the terms of this Agreement (including the limitations set forth in Section 10.6). Notwithstanding the foregoing, the Indemnifying Party shall not have the right to assume control of such defense if the Indemnification Claim for which the Indemnifying Party seeks to assume control (I) seeks non-monetary relief, (II) involves criminal or quasi-criminal allegations, (III) involves a claim by a Governmental Authority that would adversely affect the licensure, enrollment or certification of any of the Companies or their Affiliates, or (IV) involves a claim that, in the good faith judgment of the Indemnified Party, the Indemnifying Party failed or is failing to prosecute or defend in good faith.
- (b) In the event that the Indemnifying Party fails to elect to assume control of the defense of any Indemnification Claim in the manner set forth in Section 10.7(a), the Indemnified Party may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, the Indemnification Claim in any manner it may deem appropriate (and the Indemnified Party need not consult with, or obtain any consent from, any Indemnifying Party in connection therewith). If the Indemnifying Party has not yet admitted its liability to defend the

Indemnified Party against a third party Claim, the Indemnified Party shall send written notice to the Indemnifying Party of any proposed settlement and the Indemnifying Party shall have the option for ten (10) days following receipt of such notice to (i) admit in writing its liability to indemnify the Indemnified Party from and against the Loss and if liability is so admitted, either (a) consent to such settlement or (b) reject, in its reasonable judgment, the proposed settlement or

(ii) deny liability. Any failure by the Indemnifying Party to respond to such notice shall be deemed to be an election under subsection (ii) above.

- (c) If the Indemnifying Party is controlling the defense of any Indemnification Claim in accordance with Section 10.7(a), (i) the Indemnified Party may nonetheless participate, at his, her or its own expense, in the defense of such Indemnification Claim, and (ii) the Indemnifying Party will not settle or compromise any Indemnification Claim or permit a default or consent to entry of any judgment without the prior written consent of the Indemnified Party (such consent not to be unreasonably withheld, conditioned, or delayed); provided, that the Indemnified Party shall have no obligation of any kind to consent to the entrance of any judgment or into any settlement unless such judgment or settlement (A) is for only money damages, the full amount of which shall be paid by or on behalf of the Indemnifying Party and (B) includes, as a condition thereof, an express, unconditional release of the Indemnified Party from any liability or obligation with respect to such Indemnification Claim.
- (d) Irrespective of which party controls the defense of any Indemnification Claim, the other parties to this Agreement will, and will cause any non-party Affiliate to, cooperate with the controlling party in such defense and make available to the controlling party all witnesses, pertinent records, materials and information in such non-controlling party's possession or under its control relating thereto as is reasonably required by the controlling party. The parties agree that all communications between any party and counsel responsible for or participating in the defense of any Indemnification Claim shall, to the extent possible, be made so as to preserve any applicable attorney-client or work-product privilege.
- (e) If an Indemnified Party seeks indemnification under this <u>Article X</u> with respect to any matter that does not involve an Indemnification Claim, the Indemnified Party shall give written notice to the Indemnifying Party promptly after discovering the liability, obligation or facts giving rise to such claim for indemnification, describing the nature of the claim in reasonable detail, the amount thereof (if known and quantifiable), and the basis thereof; <u>provided</u>, that any failure to so notify or any delay in notifying the Indemnifying Party shall not relieve the Indemnifying Party of its or his obligations hereunder except to the extent that the Indemnifying Party is materially prejudiced by such failure or delay, and <u>provided further</u>, that the Buyer Indemnified Parties shall be responsible to submit claims in respect of the R&W Policy. If the Indemnifying Party has delivered an indemnity dispute notice to the Indemnified Party, the Indemnifying Party and the Indemnified Party shall proceed in good faith to negotiate a resolution to such dispute. Notwithstanding <u>Section 10.11</u>, if the Indemnifying Party and the Indemnified Party cannot resolve such dispute within thirty (30) days after delivery of the indemnity dispute notice, such dispute shall be resolved in accordance with <u>Section 10.23</u>.
- (f) Any indemnification payment due and owing pursuant to this <u>Article X</u> shall be effected by wire transfer of immediately available funds to an account designated by the Indemnified Party within five (5) Business Days after the determination of the amount thereof,

whether as a result of a final decision, judgment or award having been rendered by a Governmental Authority of competent jurisdiction and the expiration of the time in which to appeal therefrom, or a settlement shall have been consummated, or the Indemnified Party and the Indemnifying Party shall have arrived at a mutually binding agreement with respect to an Indemnification Claim hereunder; provided, that, that any payments required to be made in accordance with the R&W Policy shall be submitted by Buyer pursuant to the R&W Policy.

- (g) Notwithstanding anything to the contrary in this <u>Section 10.7</u>, to the extent that coverage may be available in respect of any Indemnification Claim under the R&W Policy, the insurer shall have the rights contemplated by the R&W Policy in connection with the control or participation in the defense of such Indemnification Claim, and this <u>Section 10.7</u> shall not apply to the extent the rights or obligations set forth in this <u>Section 10.7</u> conflict with the R&W Policy.
- (h) Notwithstanding anything in this Agreement to the contrary, the Excluded Liability, including any Claims in connection therewith, shall be handled in accordance with and subject to Exhibit D.

10.8 Waiver of Other Representations.

- (a) NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, IT IS THE EXPLICIT INTENT OF EACH PARTY, AND THE PARTIES HEREBY AGREE, THAT NEITHER SELLER NOR ANY OF ITS AFFILIATES OR THEIR RESPECTIVE REPRESENTATIVES HAS MADE OR IS MAKING ANY REPRESENTATION OR WARRANTY WHATSOEVER, EXPRESS OR IMPLIED, WRITTEN OR ORAL, INCLUDING ANY IMPLIED REPRESENTATION OR WARRANTY AS TO THE CONDITION, MERCHANTABILITY, USAGE, SUITABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE WITH RESPECT TO THE INTERESTS, THE COMPANIES, THE ASSETS OF THE COMPANIES OR THE BUSINESS OF THE COMPANIES, OR ANY PART THEREOF, EXCEPT THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY CONTAINED IN ARTICLE III AND ARTICLE IV.
- (b) EXCEPT THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY CONTAINED IN <u>ARTICLE III</u> AND <u>ARTICLE IV</u>, THE INTERESTS ARE BEING TRANSFERRED "AS IS, WHERE IS, WITH ALL FAULTS," AND SELLER, ITS AFFILIATES AND THEIR RESPECTIVE REPRESENTATIVES EXPRESSLY DISCLAIM ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, AS TO THE CONDITION, VALUE OR QUALITY OF THE COMPANIES OR THEIR ASSETS OR THE PROSPECTS (FINANCIAL OR OTHERWISE), RISKS AND OTHER INCIDENTS OF THE COMPANIES OR THEIR ASSETS.
- 10.9 *Exclusive Remedy*. Notwithstanding anything contained herein to the contrary but subject to Section 10.24, except in the case of Actual Fraud, recovery against the R&W Policy and indemnification pursuant to the provisions of this Article X shall be the exclusive remedy for the Parties for any misrepresentation or breach of any representation, warranty, covenant, agreement or other provision contained in this Agreement, the other Transaction Documents or in any certificate or other instrument or document delivered pursuant hereto or thereto. Without limiting the generality of the foregoing and notwithstanding anything to the contrary herein, except in the

case of Actual Fraud, Buyer acknowledges and agrees that, from and after the Closing, its sole and exclusive remedy with respect to any and all Claims arising out of, concerning or relating, directly or indirectly, to the subject matter of this Agreement, the Interests, the Companies, the Companies' Assets, the Business or the transactions contemplated hereby, regardless of the legal theory under which such liability or obligation may be sought to be imposed, whether sounding in contract or tort, or whether at law or in equity, or otherwise, shall be recovery against the R&W Policy or indemnification pursuant to the provisions set forth in this Article X (which shall be limited and restricted as set forth herein) and the Buyer Indemnified Parties shall have no other remedy or recourse with respect to any of the foregoing other than specific performance pursuant to, and subject to the terms and conditions of, Section 10.24. Buyer acknowledges and agrees, on its own behalf and on behalf of the other Buyer Indemnified Parties, that they may not avoid the limitation on liability set forth in this Section 10.9 by (x) seeking damages for breach of contract, tort or pursuant to any other theory of liability, all of which are hereby expressly waived or (y) asserting or threatening any claim against any Person that is not a party (or a successor to a Party) for breaches of the representations, warranties or covenants contained in this Agreement.

- 10.10 *Access to Information*. After the Closing Date, Seller and Buyer shall grant each other (or their respective designees), and Buyer shall cause the Companies to grant to Seller (or its designee), access at all reasonable times to all of the books, records, documents, instruments, accounts, correspondence, writings, evidences of title and other papers and electronic files relating to the Business in the possession of such granting party or its Affiliates (the "*Records*"), and shall afford the other party the right (at such party's expense) to take extracts therefrom and to make copies thereof, to the extent reasonably necessary to implement the provisions of, or to investigate or defend any Claims among the Parties and/or their Affiliates arising under, this Agreement, and/or as required or requested by Seller or its Affiliates in connection with the Excluded Liabilities, including the defense thereof. Buyer shall maintain, and shall cause the Companies to maintain, such Records until the seventh anniversary of the Closing Date (or for such longer period of time as Seller shall advise Buyer is necessary in order to have the Records available with respect to Tax matters), or if any of the Records pertain to any Claim or Dispute pending on the seventh anniversary of the Closing Date, Buyer shall maintain any of the Records designated by Seller or its Representatives until such Claim or Dispute is finally resolved and the time for all appeals has been exhausted.
- 10.11 *Dispute Resolution.* In the event of any dispute, controversy or Claim between the Parties arising out of or relating to this Agreement, or the breach or invalidity thereof (collectively, a "Dispute"). The Parties agree to attempt to resolve all Disputes arising hereunder promptly, equitably and in a good faith manner provided that either Party may seek to enforce his rights and remedies under this Agreement at any time. The Parties further agree to provide each other with reasonable access during normal business hours to any and all non-privileged records, information and data pertaining to such Dispute, upon reasonable advance notice.
- 10.12 Damages Waiver. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THIS AGREEMENT, NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY OR TO SUCH OTHER PARTY'S INDEMNIFIED PARTIES FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES SUFFERED BY SUCH PARTY RESULTING FROM OR ARISING OUT OF THIS AGREEMENT, THE OTHER TRANSACTION DOCUMENTS OR THE

TRANSACTIONS CONTEMPLATED HEREIN OR THEREIN, INCLUDING LOST PROFITS, LOSS OF REVENUE OR DIMINUTION OF VALUE; PROVIDED, HOWEVER, THE FOREGOING SHALL NOT BE CONSTRUED AS LIMITING CLAIMS RELATED TO ACTUAL FRAUD OR AN OBLIGATION OF A PARTY TO INDEMNIFY, DEFEND AND HOLD HARMLESS THE OTHER PARTY OR THE OTHER PARTY'S INDEMNIFIED PARTIES AGAINST CLAIMS ASSERTED BY THIRD PARTIES, INCLUDING, BUT NOT LIMITED TO, THIRD PARTY CLAIMS FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES.

10.13 *Notices*. Unless this Agreement specifically requires otherwise, any notice, demand or request provided for in this Agreement, or served, given or made in connection with it, shall be in writing and shall be deemed properly served, given or made if delivered in person or sent by electronic delivery (including facsimile or delivery of a document in Portable Document Format), by registered or certified mail, postage prepaid or by a nationally recognized overnight courier service that provides a receipt of delivery, in each case, to the Parties at the addresses specified below:

If to Seller, to:

Summit Midstream Holdings, LLC 910 Louisiana Street, Suite 4200

Houston, TX 77002 Attn: James Johnston

Email: James.Johnston@summitmidstream.com With copies (which shall not

constitute notice) to:

Locke Lord LLP 600 Travis Street, Suite 2800 Houston, Texas 77002

Attention: H. William Swanstrom and Jennie Simmons

Email: <u>BSwanstrom@lockelord.com</u> and <u>Jennie.Simmons@lockelord.com</u>

If to Buyer, to:

Matador Resources Company 5400 LBJ Freeway, Suite 1500 Dallas, TX 75240 Attn: Bryan Erman

Email: berman@matadorresources.com With a copy (which shall not

constitute notice) to:

O'Melveny & Myers LLP 2501 N. Harwood, Suite 1700 Dallas, TX 75201

Attn: Jason Schumacher

Email: jschumacher@omm.com

- (b) Notice given by personal delivery, mail or overnight courier pursuant to this <u>Section 10.13</u> shall be effective upon physical receipt. Notice given by facsimile or other electronic transmission pursuant to this <u>Section 10.13</u> shall be effective as of the date of confirmed delivery if delivered before 5:00 P.M. Central Time on any Business Day at the place of receipt (*provided* that such intended recipient shall have an affirmative duty to reply promptly upon receipt if received during business hours; and *provided further*, that an automated response from the email account or server of the intended recipient shall not constitute an affirmative reply) or the next succeeding Business Day if confirmed delivery is after 5:00 P.M. Central Time on any Business Day or during any non-Business Day at the place of receipt.
- 10.14 *Entire Agreement*. Except for the Confidentiality Agreement, this Agreement (including the related Exhibits and Schedules) and the other Transaction Documents supersede all prior discussions, understanding, negotiations and agreements, whether oral or written, between the Parties and/or their respective Affiliates with respect to the subject matter hereof and contain the sole and entire agreement between the Parties and their respective Affiliates hereto with respect to the subject matter hereof.
- 10.15 *Expenses*. Except as otherwise expressly provided in this Agreement, whether or not the transactions contemplated hereby are consummated, each Party shall pay all costs and expenses it has incurred or will incur in anticipation of, relating to and in connection with the negotiation and execution of this Agreement and consummation of the transactions contemplated hereby.
- 10.16 **Disclosure**. Seller may, at its option, include in the Schedules items that are not material or required by this Agreement, and any such inclusion, or any references to dollar amounts, shall not be deemed to be an acknowledgment or representation that such items are material, to establish any standard of materiality or to define further the meaning of such terms for purposes of this Agreement. In no event shall disclosure of any matter, fact, occurrence, information or circumstance in the Schedules be deemed or interpreted to broaden the scope of the representations and warranties, obligations, covenants, conditions, indemnities or agreements contained in this Agreement, or to create any representation, warranty, obligation, covenant, condition, indemnity or agreement that is not contained in this Agreement. In particular, if the Schedules contain supplementary information not specifically required under this Agreement to be included in the Schedules, such additional matters are set forth solely for informational purposes, are not represented or warranted in this Agreement and do not necessarily include other matters of a similar nature. In addition, the disclosure of any matter in the Schedules is not to be deemed an admission against any Party that such matter actually constitutes noncompliance with or a violation of contract or Law or other topic to which such disclosure is applicable. In disclosing any matter, fact, occurrence, information or circumstance in the Schedules, no Party is waiving any attorney-client privilege associated with any such matter, fact, occurrence, information or circumstance, or any protection afforded by the "work product doctrine" with respect to any of the same. Any exception, qualification or other disclosure set forth on the Schedules attached to this Agreement with respect to a particular representation or warranty in this Agreement shall be deemed to be an exception, qualification or other disclosure with respect to all other representations or warranties contained in this Agreement to the extent any description of facts regarding the event, item or matter disclosed is adequate so as to make reasonably apparent on the face of such disclosure that such exception, qualification or disclosure is applicable to such other

representations or warranties, whether or not such exception, qualification or disclosure is so numbered.

- 10.17 *Waiver*. Any term or condition of this Agreement may be waived at any time by the Party that is entitled to the benefit thereof, but no such waiver shall be effective unless set forth in a written instrument duly executed by or on behalf of the Party waiving such term or condition. No waiver by either Party of any term or condition of this Agreement, in any one or more instances, shall be deemed to be or construed as a waiver of the same or any other term or condition of this Agreement on any future occasion.
- 10.18 *Amendment*. This Agreement may be amended, supplemented or modified only by a written instrument duly executed by or on behalf of Buyer and Seller.
- 10.19 *No Third Party Beneficiary*. Except for the express provisions of this Agreement are intended solely for the benefit of the Parties and their respective successors or permitted assigns, and it is not the intention of the Parties to confer third-party beneficiary rights upon any other Person. Only a Party and its successors and permitted assigns will have the right to enforce the provisions of this Agreement on its own behalf or on behalf of any of its related indemnified parties under this Agreement (but shall not be obligated to do so).
- 10.20 **Assignment; Binding Effect.** Neither this Agreement nor any right, interest or obligation hereunder may be assigned by any Party without the prior written consent of the other Party, and any attempt to do so will be void, except for assignments and transfers by operation of Law; *provided*, *however*, Buyer shall be permitted to assign this Agreement to any Affiliate without the prior written consent of the Seller; *provided further*, that such assignment shall not relieve Buyer of any liabilities or obligations hereunder. Subject to this <u>Section 10.20</u>, this Agreement is binding upon, inures to the benefit of and is enforceable by the Parties and their respective successors and permitted assigns.
- 10.21 *Invalid Provisions*. If any provision of this Agreement is held to be illegal, invalid or unenforceable under any present or future Law, and if the rights or obligations of any Party under this Agreement will not be materially and adversely affected thereby, such provision shall be fully severable, this Agreement shall be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, the remaining provisions of this Agreement shall remain in full force and effect and shall not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom and in lieu of such illegal, invalid or unenforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.
- 10.22 *Counterparts; Facsimile*. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Any facsimile or PDF copies hereof or signature hereon shall, for all purposes, be deemed originals.
 - 10.23 Governing Law; Enforcement.

- (a) THIS AGREEMENT SHALL BE GOVERNED, CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, WITHOUT GIVING EFFECT TO ANY CONFLICT OR CHOICE OF LAW PROVISION THAT WOULD RESULT IN THE IMPOSITION OF ANOTHER JURISDICTION'S LAW. THE LAWS OF THE STATE OF TEXAS SHALL GOVERN ANY DISPUTE, CONTROVERSY, REMEDY OR CLAIM BETWEEN THE PARTIES ARISING OUT OF, RELATING TO, OR IN ANY WAY CONNECTED WITH THIS AGREEMENT, INCLUDING THE EXISTENCE, VALIDITY, PERFORMANCE, OR BREACH THEREOF. WITH RESPECT TO ANY PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.
- The Parties hereby irrevocably submit to the exclusive jurisdiction of any court of competent jurisdiction located in Midland, Texas, and appropriate appellate courts therefrom, over any dispute arising out of or relating to this Agreement or any Transaction Document or any of the transactions contemplated hereby or thereby, and each Party irrevocably agrees that all claims in respect of such dispute shall be heard and determined exclusively in such courts and each Party irrevocably agrees to not institute suit over any dispute arising out of relating to this Agreement in any other forum. The Parties irrevocably waive, to the fullest extent permitted by Law, any objection which they may now or hereafter have to the laying of venue of any dispute arising out of or relating to this Agreement or any Transaction Document or any of the transactions contemplated hereby or thereby brought in such courts or any defense of inconvenient forum for the maintenance of such dispute. Each of the Parties agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. This consent to jurisdiction is being given solely for purposes of this Agreement and is not intended to, and shall not, confer consent to jurisdiction with respect to any other dispute in which a Party may become involved. Each of the Parties hereby consents to process being served by any Party in any Proceeding of the nature specified in this Agreement by the mailing of a copy thereof in the manner specified by the provisions of Section 10.13. To the extent that any Party or any of its Affiliates has acquired, or hereafter may acquire, any immunity from jurisdiction of any court or from any legal process (whether through service or notice, attachment prior to judgment, attachment in aid of execution, execution or otherwise) with respect to itself or its property, such Party (on its own behalf and on behalf of its Affiliates) hereby irrevocably (i) waives such immunity in respect of its obligations with respect to this Agreement and (ii) submits to the personal jurisdiction of any court described in this Section 10.23.
- 10.24 *Specific Performance*. Subject to the limitations set forth in this Agreement, the Parties expressly agree in the event of breach of this Agreement by either Party, the other Party will, in addition to any other rights and remedies hereunder, have the right to enforce this Agreement by seeking an order for specific performance of the Parties' obligations hereunder.
- 10.25 *No Recourse.* This Agreement may only be enforced against, and any claims or causes of action that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement may only be made against the entities that are expressly identified as Parties hereto in their capacities as such and, except to the extent otherwise provided herein, no former, current or future equity holders, controlling Persons, directors, officers, employees, agents or Affiliates of any Party hereto or any former, current or future, direct or indirect, equity holder, controlling Person, director, officer, employee, general or limited

partner, member, manager, agent or Affiliate of any of the foregoing (each, a "*Non-Recourse Party*") shall have any liability for any obligations or liabilities of the Parties to this Agreement or for any claim (whether in tort, contract or otherwise) based on, in respect of, or by reason of, the transactions contemplated hereby or in respect of any representations made or alleged to be made in connection herewith. Without limiting the rights of any Party against the other Party hereto, in no event shall any Party or any of its Affiliates seek to enforce this Agreement against, make any claims for breach of this Agreement against, or seek to recover monetary damages from, any Non-Recourse Party.

10.26 **Legal Representation**. Following the Closing, Locke Lord, LLP may serve as counsel to Seller and its Non-Recourse Parties in connection with any litigation, claim or obligation arising out of or relating to this Agreement, notwithstanding such representation or any continued representation of any other Person, and each of the Parties (on behalf of itself and each of its Non-Recourse Parties) consents thereto and waives any conflict of interest arising therefrom. The decision to represent Seller and its Affiliates shall be solely that of Locke Lord, LLP. Any privilege attaching as a result of Locke Lord, LLP representing Seller or any of its Non-Recourse Parties (including the Companies) in connection with this Agreement shall survive the Closing and shall remain in effect; provided, that such privilege from and after the Closing shall be assigned to and controlled by Seller; provided, further, that in the event that any dispute arises after the Closing between Buyer or any Company, on the one hand, and any party other than the Parties or any of their respective Affiliates, on the other hand, then Buyer and the Companies may assert such privilege to prevent the disclosure of any Privileged Communications by Locke Lord, LLP to such third party. In furtherance of the foregoing, each of the Parties agrees to take the steps necessary to ensure that any privilege attaching as a result of Locke Lord, LLP representing the Companies in connection with this Agreement shall survive the Closing, remain in effect and be assigned to and controlled by Seller. As to any privileged attorney client communications between Locke Lord, LLP and the Companies prior to the Closing Date (collectively, the "Privileged **Communications**"), Buyer, together with any of its Affiliates, successors or assigns, agree that no such party may use or rely on any of the Privileged Communications in any action or claim against or involving any of the Parties or any of their respective Non-Recourse Parties after the Closing.

10.27 **Summit Parent**. Summit Parent agrees to be jointly and severally liable with Seller under Section 10.3(f) of this Agreement solely related to all Losses arising from, based upon, related to or associated with the Excluded Claims; provided however, that Summit Parent is a party to this Agreement for such sole purpose but further agrees to be bound to the provisions of this Agreement necessary for Buyer to have the rights and remedies against Summit Parent as Buyer would have against Seller under Section 10.3(f) of this Agreement related to all Losses arising from, based upon, related to or associated with the Excluded Claims.

[Signature Page Follows]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officer of each Party as of the date first above written.

SELLER:

SUMMIT MIDSTREAM HOLDINGS, LLC

By:

Name: J. Heath Deneke Title: Chief Executive Officer

FOR THE LIMITED PURPOSES AS SET FORTH IN SECTION 10.27

SUMMIT PARENT:

SUMMIT MIDSTREAM PARTNERS, LP By: Summit Midstream GP, LLC, its General Partner

By:

Name: J. Heath Deneke Title: Chief Executive Officer

BUYER:

LONGWOOD GATHERING AND DISPOSAL SYSTEMS, LP

By: Longwood Gathering and Disposal Systems GP, Inc., its General Partner

By:

Name: G. Gregg Krug

Title: Executive Vice President - Marketing and

Midstream Strategy

CERTIFICATIONS

- I, Heath Deneke, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Summit Midstream Partners, LP;
 - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
 - 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information: and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	August 5, 2022	/s/ Heath Deneke
		Heath Deneke President, Chief Executive Officer and Director of Summit Midstream GP, LLC (the general partner of Summit Midstream Partners, LP)

CERTIFICATIONS

- I. William J. Mault, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Summit Midstream Partners, LP;
 - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
 - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
 - 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared:
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles:
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation: and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	August 5, 2022	/s/ William J. Mault
		William J. Mault Executive Vice President and Chief Financial Officer of Summit Midstream GP, LLC (the general partner of Summit Midstream Partners, LP)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of Summit Midstream Partners, LP (the "Registrant") for the quarterly period ended June 30, 2022, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, Heath Deneke, as President, Chief Executive Officer and Director of Summit Midstream GP, LLC, the general partner of the Registrant, and William J. Mault, as Executive Vice President and Chief Financial Officer of Summit Midstream GP, LLC, the general partner of the Registrant, each hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to his knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Registrant.

/s/ Heath Deneke

Name: Heath Deneke

Title: President, Chief Executive Officer and Director of Summit Midstream GP, LLC

(the general partner of Summit Midstream Partners, LP)

Date: August 5, 2022

/s/ William J. Mault

Name: William J. Mault

Title: Executive Vice President and Chief Financial Officer of Summit Midstream GP, LLC

(the general partner of Summit Midstream Partners, LP)

Date: August 5, 2022